School of Physiotherapy and Exercise Science

Telehealth in Cystic Fibrosis: Does the Integration of Modern Technology with Traditional Care Improve Health Related Outcomes?

Jamie Wood

0000-0003-3172-0863

This thesis is presented for the Degree of Doctor of Philosophy (Physiotherapy) of Curtin University

APRIL 2020

AUTHOR'S DECLARATION

I declare that this thesis is my own account of my research and contains as its main content

work which has not previously been submitted for a degree at any tertiary education institution.

Jamie Wood

30th April 2020

STATEMENT OF ORIGINALITY

This thesis is presented for the degree of Doctor of Philosophy (Physiotherapy) at Curtin University, Western Australia. This degree was completed on a part-time basis. Studies were undertaken between May 2013 and March 2018, through the School of Physiotherapy and Exercise Science at Curtin University, in association with the Physiotherapy Department and Department of Respiratory Medicine at Sir Charles Gairdner Hospital, and the Institute for Respiratory Health, Western Australia.

This research project was developed in association with my supervisors, who have been involved in editing both the thesis and all associated publications.

All material presented in this thesis is original.

ABSTRACT

This chapter presents the four studies that form this research programme in extended abstract form. This research programme was designed to:

- evaluate and compare the magnitude of decline in spirometry between adults with cystic fibrosis (CF) living in the Perth metropolitan area and in rural and remote
 Western Australia (WA) (Study 1, presented in Chapter 3);
- examine the uptake and participant satisfaction with telehealth clinics provided via videoconferencing for adults with CF living in rural and remote WA (Study 2; Part A, presented in Chapter 4);
- iii) explore the working profile of adults with CF living in rural and remote WA (Study 2; Part B, presented in Chapter 4);
- investigate the usability of a smartphone application (app) developed for adults with
 CF to report their symptoms suggestive of a respiratory exacerbation to the CF team
 (Study 3; Part A; presented in Chapter 5); and observer agreement between
 clinicians interpreting the app data (Study 3, Part B; presented in Chapter 5); and,
- v) examine the impact of the app on health outcomes in adults with CF in a randomised controlled trial (RCT) (Study 4, presented in Chapter 6).

A comparison of the magnitude of decline in spirometry between adults with cystic fibrosis living in the Perth metropolitan area and in rural and remote Western Australia (Study 1)

Background and aims

The Australian CF Standards of Care recommend that all adults with CF attend a minimum of four outpatient clinics with a specialist CF team per year. In the year preceding this study, this recommendation was not met for those living in rural and remote WA. This may negatively impact on long-term health outcomes, as it is known that those with CF who receive care in a specialist centre have improved forced expiratory volume in 1 second (FEV₁) and weight. This study evaluated and compared the magnitude of decline in spirometry between adults with CF living in metropolitan Perth and in rural and remote WA.

Methods

Adults with CF were included if they i) attended the Sir Charles Gairdner (SCGH) CF Centre for the entire period from January 2010 to December 2012; and ii) performed spirometry on at least two occasions during this time period. An analysis was conducted to determine the magnitude of decline in spirometry (FEV₁ and percentage predicted FEV₁ [ppFEV₁], forced vital capacity [FVC] and percentage predicted FVC [ppFVC]) of all adults with CF attending the adult CF clinic at SCGH during this time period. The decline in spirometry over this 3-year period was compared between adults with CF living in metropolitan Perth vs. those living in rural and remote WA (defined as living greater than 100km from the geographical centre of Perth). Analysis of the primary outcome was conducted using mixed effects models.

Results

A total of 156 participants met the inclusion criteria and were included in the audit and analyses. In total, 2,748 ((mean [range] 219 (8 [2 to 24]) and (21 [3 to 58])) data points (episodes of spirometry) were available for participants living in metropolitan (n = 130) and in rural and remote (n = 26) areas respectively.

ii

There was no clear effect on the magnitude of decline in FEV₁ (mean difference [95% confidence interval, CI] 0.00 [-0.01 to 0.01], p = 0.16), $ppFEV_1$ (0.07 [-0.07 to 0.21], p = 0.31), FVC (0.00 [-0.01 to 0.01], p = 0.31) or ppFVC (0.05 [-0.10 to 0.20], p = 0.53) between participants living in metropolitan and in rural and remote areas. Results remained unchanged when stratified for age (18 to 29 years or ≥ 30 years).

Discussion and conclusion

No clear effect on the magnitude of decline of spirometry was demonstrated when comparing adults with CF living in the metropolitan area and in rural and remote WA. This was unexpected, however may have been due to several limitations. The sample size was limited to the available participants and therefore the analyses may have been underpowered to detect small differences. Further, the 26 participants living in rural and remote WA averaged a total of only eight spirometry recordings over the 3-year study period, rendering it difficult to formulate any precision around the estimated magnitude of decline. This would draw attention to the lack of appropriate disease surveillance in this group. The results of this audit highlight the need for improved access to care and disease monitoring in adults with CF living in rural and remote WA.

Telehealth clinics for adults with cystic fibrosis living in rural and remote Western Australia (Study 2, Part A)

Background and aims

At the commencement of this study, 15% (n = 28) of adults with CF in WA lived in rural and remote areas. Those living in these areas have difficulty accessing specialist care at the SCGH CF Centre due to geographical, time and financial barriers. This study sought to improve access to care by offering telehealth clinics via videoconferencing, and evaluate its uptake and impact on satisfaction and health-related outcomes such as healthcare utilisation (HCU), spirometry, weight and health-related quality of life (HRQoL).

Methods

Adults with CF aged 18 years or more and living a minimum of 100km from the SCGH CF Centre were eligible to participate in this study. Telehealth clinics were offered via videoconferencing over a 12-month period (approximately every 3 months or sooner if clinically indicated). Participants were still able attend clinics 'in-person' at the CF centre if they preferred. On the day of the telehealth clinic, participants attended their nearest regional hospital. The participant's height, weight and spirometry were measured (according to American Thoracic Society spirometry guidelines) by a designated health professional. Uptake, satisfaction and measures of HCU were collected at the end of the study. The HCU data obtained during the study were compared with the 12-month period preceding the study. Participants completed other measurements at baseline, at each clinic (telehealth or in-person) and at the completion of the study period. These measures comprised spirometry, weight, HRQoL and absenteeism and presenteeism. Analysis of the primary outcome was conducted using descriptive statistics.

Results

For the 21 participants, total CF clinics attended increased from 46 (median [range] per participant 2 [0 to 6]) in the 12-month period preceding the study to 100 (5 [2 to 8]) during the intervention (incidence rate ratio [IRR] 3.7, p < 0.01). Of the 100 clinics attended in total, 66

were delivered via telehealth. Satisfaction with telehealth was high and most (94%) participants agreed that telehealth was a good way to deliver CF care. There was an improvement in the vitality domain of the Cystic Fibrosis Questionnaire – Revised (mean difference [95% CI] 11 [1 to 22], p = 0.04). No clear differences were observed in other outcomes.

Discussion and conclusion

Telehealth had good uptake and increased clinic attendance in adults with CF living in rural and remote WA, and had high satisfaction amongst participants. The increase in HCU, resulting from increased detection and treatment of exacerbations, may improve long-term outcomes in this population. These findings highlight the inadequate clinical surveillance amongst participants in the 12-months preceding the study.

The impact of cystic fibrosis on work attendance and performance in adults living in rural and remote Western Australia: a sub-analysis (Study 2, Part B)

Background and aims

As the expected age of survival for people with CF increases to above 40 years, there is a need for adults with CF to stay in the work force for longer. However, there are few data available regarding the impact that CF has on work attendance and performance. Telehealth clinics for adults with CF were implemented and evaluated, and, as part of this study work absenteeism and presenteeism was quantified.

Methods

During CF clinics (telehealth and in-person) throughout a 12-month period, participants completed the absenteeism and presenteeism questions of the World Health Organisation's Health and Work Performance Questionnaire (HPQ). Responses were used to calculate relative absenteeism and relative presenteeism. Relative absenteeism is the time absent from work. A maximum score of 1 indicates 100% absenteeism; a score of 0 indicates no absenteeism. Negative scores indicate the hours worked exceeded the employer's expectations. Relative presenteeism is self-rated work performance compared to others in similar employment. A score of 1 represents equal performance, 0 represents worst performance, and a maximum score of 2 represents performance at least twice as good as colleagues. Analyses were conducted using descriptive statistics.

Results

Data from 23 adults with CF (14 female, mean [standard deviation, SD] age 31 [10] years) were available. Data were collected during a total of 91 CF clinic attendances. Participants were engaged in paid work on 61 (67%) of these occasions. Twenty-two (96%) participants were employed on at least one occasion and 11 (48%) were employed throughout the entire 12 months. For those employed, the median (range) number of hours worked per week was 38 (10

vi

to 80). Relative absenteeism (mean [range]) in those employed was 0.1 (-1.2 to 0.9). Relative presenteeism in those employed was 1.0 (0.0 to 1.8).

Discussion and conclusion

Despite having a chronic disease that is associated with a substantial treatment burden, most adults with CF living in rural and remote WA were engaged in paid work during the 12-month study period. Absenteeism scores were variable, and demonstrated that CF affected the work attendance of many, but not all participants. Presenteeism scores, while also variable, demonstrated that on average participants felt they performed as well as their colleagues at work. The use of the HPQ is novel in this population, and warrants further consideration as a tool for quantifying absenteeism and presenteeism in adults with CF.

The usability of a smartphone application for reporting respiratory symptoms in adults with cystic fibrosis, and observer agreement between clinicians (Study 3, Part A/Part B)

Background and aims

In CF, respiratory exacerbations increase HCU and costs, impair lung function and HRQoL, and reduce survival. Delayed identification and treatment can lead to more severe respiratory exacerbations and worse clinical outcomes. Delayed reporting of worsening respiratory symptoms by people with CF is one of the key factors in the late identification of respiratory exacerbations by the CF team. Therefore, there is a need for a novel approach to facilitate the early identification and treatment of respiratory exacerbations in this population. This study investigated the usability of a smartphone application (app), designed by the research team, for adults with CF to report respiratory symptoms to the CF team (Part A). Further, it examined the observer agreement in clinical decision making between experienced clinicians interpreting the app responses (Part B).

Methods

Part A: Adults with CF used the app weekly for 4 weeks. The app comprised 10 yes/no questions regarding respiratory symptoms and two questions regarding emotional well-being. Usability was measured with the System Usability Scale (SUS). Participants were also asked to provide feedback on the wording of the app questions. Part B: Observer agreement was examined by providing a physician and a nurse practitioner who were experienced in providing CF care with 45 symptom responses from the app embedded across three separate clinical scenarios. For each scenario the clinicians, who were blinded to each other's responses, were asked to indicate whether or not they would: (i) initiate telephone contact, and/or (ii) request the person with CF attend a CF clinic. Analyses were conducted using descriptive statistics and Cohen's kappa statistic

Results

Part A: Ten participants (5 females, mean [SD] age 33 [11] years, ppFEV₁ 49 [27]) completed the study. The mean (SD) SUS score was 94 (6). Participants provided feedback regarding the wording of certain app questions. Part B: For the clinical scenarios, there was perfect agreement between clinicians deciding it was appropriate to initiate contact ($\kappa = 1.0$, p < 0.001), and nearperfect agreement for requesting the person to attend a CF clinic ($\kappa = 0.86$, p < 0.001).

Discussion and conclusion

The results of this study demonstrated that amongst adults with CF, an app used to report respiratory symptoms suggestive of a respiratory exacerbation has excellent usability. Further, the symptom responses provided can be interpreted consistently between clinicians who are experienced in the management of this clinical population. The app was revised based on participant feedback, and the difference in responses from clinicians relating to one specific question, before investigating its impact on health outcomes in an RCT.

A smartphone application for reporting respiratory symptoms in adults with cystic fibrosis improves the detection of respiratory exacerbations: a randomised controlled trial (Study 4)

Background and aims

In CF, respiratory exacerbations have a profound and long-lasting detrimental impact. An app has previously been developed for adults with CF to report respiratory symptoms to the CF team. In a previous study, it was demonstrated that the app had excellent usability, and the data could be interpreted consistently between clinicians. In this RCT, the impact of the weekly use of this app was examined on the number of courses of IV antibiotics and other health-related outcomes.

Methods

A 12-month RCT was undertaken. Adults were eligible for inclusion if they had a diagnosis of CF and had required IV antibiotics in the preceding 12 months for a respiratory exacerbation. Participants were randomised to an intervention group or a control group. Participants in both groups received usual care, which comprised routine outpatient clinic appointments and the ability to contact the nurse practitioner via phone or email. Those in the intervention group were provided with the app and asked to use it weekly or sooner if they felt their respiratory symptoms had worsened. The app comprised 12 yes/no questions relating to symptoms suggestive of a respiratory exacerbation, and two questions relating to emotional wellbeing. Participants were contacted by the nurse practitioner if they answered yes to any questions, and their care was triaged accordingly (e.g. advised to attend the CF clinic, commence oral and/or inhaled antibiotics, or no intervention). The primary outcome measure was the number of courses of IV antibiotics. Other outcomes included the number of courses of oral and inhaled antibiotics, number and days of hospital admissions, time to detection of exacerbation requiring oral and/or IV antibiotics by the CF team, antibiotic and hospital admission costs, spirometry, body mass index (BMI), HRQoL, feelings of anxiety and depression, treatment adherence and adherence to the weekly use of the app. The physicians prescribing the antibiotics were blinded

to group allocation. Analysis of the primary outcome was conducted using negative binomial regression.

Results

Sixty participants (31 male, mean [SD] age 31 [9] years, ppFEV₁ 60 [18] %) were recruited, with 29 (48%) allocated to the intervention group. Over the 12-month follow-up, there was no clear effect of the app on the number of courses of IV antibiotics (IRR [95% CI] 1 [0.6 to 1.7]), however the number of courses of oral antibiotics increased in the intervention group (1.5 [1.0 to 2.2]). The median (interquartile range [IQR]) time to detection of a respiratory exacerbation requiring oral or IV antibiotics was shorter in the intervention group compared with the control group (70 [123] vs.141 [140] days; p = 0.02). There was a good level of adherence to the weekly use of the app amongst participants in the intervention group (mean [range] 77 [25 to 100]%). Subgroup analyses demonstrated a meaningful increase in oral antibiotic prescription, and reduction in IV antibiotic prescription, compared with the 12 months preceding the study in participants in the intervention group with adherence to the weekly use of the app $\ge 80\%$ (n = 15). No between-group differences were observed in other outcomes.

Discussion and conclusion

This was the first app used by adults with CF to report changes in their respiratory symptoms directly to the CF team. The high level of adherence to the weekly use of the app was attributed to the low reporting burden and use of technology already owned by participants, and was somewhat better than earlier work that has investigated home monitoring interventions in a CF population. Provision of the app facilitated the earlier detection of respiratory exacerbations and treatment in the form of oral antibiotics. While these data do not demonstrate a clear effect on the number of courses of IV antibiotics for the participants over a 12-month period, this study supports the continued development and investigation of similar apps and technologies used to identify respiratory exacerbations, with the aim of improving long-term health outcomes.

ACKNOWLEDGEMENTS

Curtin University, for providing me with the opportunity to undertake a Doctor of Philosophy degree, and for the ongoing support throughout this time.

My PhD supervisors, A/Prof Kylie Hill and Dr Sue Jenkins, who have been amazing teachers and a tremendous wealth of knowledge, support and encouragement.

Mrs Nola Cecins for your mentorship and guidance, and great friendship, not only during my PhD journey but throughout my whole career.

Mrs Sue Morey, who personifies 'making things happen'. This research would not have been possible without your hard work and support.

Clin Prof Siobhain Mulrennan, for your support throughout my time as a clinician and researcher in cystic fibrosis and respiratory medicine.

A/Prof David Putrino, who has dedicated so much of his time to a cause that he believes in, and has the ultimate 'solutions-focussed' attitude to health care and technology.

Dr Gerard Ryan for fostering my passion in cystic fibrosis and developing a strong multidisciplinary cystic fibrosis team at Sir Charles Gairdner Hospital.

The participants in the studies undertaken as part of this PhD. This would not be possible without you all and your attitude to research is second to none amongst patient groups.

The Department of Physiotherapy at Sir Charles Gairdner Hospital, and especially Ian Cooper and Tracy Hebden-Todd for supporting my work and research journey for the past 12 years.

The Respiratory Medicine, Pulmonary Physiology, Pharmacy and Finance departments at Sir Charles Gairdner Hospital for in-kind contributions and support to various parts of this research.

Ms Angela Jacques and Mrs Natasha Bear for statistical support and advice.

FUNDING

This research was supported in-kind by Curtin University and Sir Charles Gairdner Hospital. Study 2 was partially supported by the 2013 Institute for Respiratory Health Glenn Brown Memorial Grant, and Study 3 and Study 4 were partially supported by a 2015 Sir Charles Gairdner Osborne Park Healthcare Group Hospital Research Advisory Committee Grant.

TABLE OF CONTENTS

Abstract	i
A comparison of the magnitude of decline in spirometry between adults with cystic fibrosis living in the Perth metropolitan area and in rural and remote Western Australia (Study 1)	ii
Background and aims	ii
Methods	ii
Results	ii
Discussion and conclusionii	ii
Telehealth clinics for adults with cystic fibrosis living in rural and remote Western Australia (Study 2, Part A)i	v
Background and aimsi	v
Methodsi	v
Resultsi	v
Discussion and conclusion	v
The impact of cystic fibrosis on work attendance and performance in adults living in rural and remote Western Australia: a sub-analysis (Study 2, Part B)	<i>i</i>
Background and aims	'n
Methods	'n
Results	'n
Discussion and conclusion	ii
The usability of a smartphone application for reporting respiratory symptoms in adults with cystic fibrosis, and observer agreement between clinicians (Study 3, Part A/Part B)vi	ii
Background and aimsvi	ii
Methodsvii	ii
Resultsi	x
Discussion and conclusioni	х
A smartphone application for reporting respiratory symptoms in adults with cystic fibrosis improves the detection of respiratory exacerbations: a randomised controlled trial (Study 4)	x
Background and aims	х
Methods	х
Resultsx	i
Discussion and conclusionx	i

Acknowled	lgements	xii
Funding		xiii
Table of co	ontents	xiv
List of figu	ires	xix
List of table	es	xx
List of app	endices	xxi
Publication	ns arising as part of this thesis	xxii
Publication	arising using data from this thesis	xxiii
Abstracts r	elating to this thesis	xxiv
Invited pro	contactions	AAI V
		XXV
Awards and	d grants	XXV1
List of abb	reviations	xxvii
Chapter 1 I	Introduction	1
1.1	Telehealth for adults with cystic fibrosis living in rural and remote W Australia	Vestern
1.1.1	Background	2
1.1.2	Research questions	
1.1.3	Significance	
1.2	The working profile of adults with cystic fibrosis living in rural and a Western Australia	emote 4
1.2.1	Background	4
1.2.2	Research question	4
1.2.3	Significance	4
1.3	A smartphone application for reporting respiratory symptoms in adul with cystic fibrosis	.ts 5
1.3.1	Background	5
1.3.2	Research questions	6
1.3.3	Significance	7
1.4	Summary	
Chapter 2 I	Literature review	9
2.1	Cystic fibrosis: incidence, pathophysiology, morbidity and mortality	10
2.1.1	Incidence, prevalence and cost	
2.1.2	Pathophysiology, morbidity, mortality and monitoring disease traject	ory 12
2.2	Cystic fibrosis care	
2.2.1	Centre-based care	
2.2.2	Barriers to care	17
2.2.3	Treatment burden and adherence	
2.2.4	Adult cystic fibrosis care in Western Australia	

2.3	Respiratory exacerbations in cystic fibrosis	
2.3.1	Definition of a respiratory exacerbation in cystic fibrosis	21
2.3.2	Impact of exacerbations	
2.3.2.1	Healthcare utilisation and cost	
2.3.2.2	Lung function, risk of further exacerbations and survival	
2.3.2.3	Health-related quality of life and mood	
2.3.2.4	Other impacts (on physical activity, body mass index and complications of intravenous antibiotic use)	
2.3.3	Identification and treatment of respiratory exacerbations	
2.4	Telehealth	
2.4.1	Definition	
2.4.2	Telehealth in CF	
2.4.2.1	Improving access to outpatient care	
2.4.2.2	Facilitating home monitoring	39
2.4.2.3	Promotion of exercise, exercise testing, and physical activity	45
2.4.2.4	Promotion of adherence to recommended treatment	47
2.5	Conclusion	49
Chapter 3 Stu	ıdy 1	
3.1	Study design	
3.1.1	Participants	
3.1.2	Procedure and study measurements	
3.1.3	Analyses	
3.2	Results	
321	Magnitude of decline of spirometry	54
3 3	Discussion	57
Chapter 4 Stu	Idy 2 (Part A, Part B)	
4.1	Part A Design	
4.1.1	Participants	60
4.1.2	Telehealth clinics	60
4.1.3	Study measurements	61
4.1.4	Primary outcome	61
4.1.4.1	Uptake of telehealth	61
4.1.5	Secondary outcomes	62
4.1.5.1	Participant satisfaction	
4.1.5.2	Healthcare utilisation	
4.1.5.3	Spirometry	
4.1.5.4	Nutritional status	63
4.1.5.5	Health-related quality of life	63
4.1.6	Analyses	63
4.2	Part A Results	64

4.2.2 Participant satisfaction 4.2.3 Healthcare utilisation 4.2.4 Spirometry 4.2.5 Nutritional status 4.2.6 Health-related quality of life 4.3 Part A Discussion 4.4 Part B Design 4.4.1 Study measurements 4.4.1 Absenteeism and presenteeism 4.4.1 Absenteeism and presenteeism 4.5 Part B Results 4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Participant satisfaction 5.3.1 Study measurements 5.3.1.1 Observer agreement 5.3.2 Analyses 5.3.1.1 Observer agreement 5.3.2 Analyses 5.3.4 Part B Results	lehealth clinics
4.2.3 Healthcare utilisation 4.2.4 Spirometry 4.2.5 Nutritional status 4.2.6 Health-related quality of life 4.3 Part A Discussion 4.4 Part B Design 4.4.1 Study measurements 4.4.1 Absenteeism and presenteeism 4.4.1 Absenteeism and presenteeism 4.5 Part B Results 4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Participant satisfaction 5.3 Part B Design 5.3.1 Study measurements 5.3.1.1 Observer agreement 5.3.2 Analyses 5.3 Part B Results 5.3.1 Observer agreement 5.3.2 Part B Results	faction64
4.2.4 Spirometry	sation
4.2.5 Nutritional status 4.2.6 Health-related quality of life 4.3 Part A Discussion 4.4 Part B Design 4.4.1 Study measurements 4.4.1 Absenteeism and presenteeism 4.4.5 Part B Results 5.1 Part A Part B) 5.1 Part A Design 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.1.4 Study measurements 5.2 Part A Results 5.2.1 System usability 5.2.2 Participant satisfaction 5.3 Part	
4.2.6 Health-related quality of life 4.3 Part A Discussion 4.4 Part B Design 4.4.1 Study measurements 4.4.1 Absenteeism and presenteeism 4.5 Part B Results 4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.1 Participant satisfaction 5.1.4 Study measurements 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Part cipant satisfaction 5.3 Part B Design 5.3.1 Study measurements 5.3.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.5 Part A and Part B Discussion Chapter 6 Study 4 6.1 6.1.1 Inclusion criteria	ıs
4.3 Part A Discussion 4.4 Part B Design 4.4.1 Study measurements 4.4.1 Absenteeism and presenteeism 4.5 Part B Results 4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.1 Part A Results 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Part cipant satisfaction 5.3.1 Study measurements 5.3.1 Study measurements 5.3.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.5 Part A and Part B Discussion Chapter 6 Study 4 6.1 6.1.1 Inclusion criteria 6.1.1.2 Exclusion criteria 6.1.2 Randomisation and allocation concealment	uality of life
4.4 Part B Design 4.4.1 Study measurements 4.4.1.1 Absenteeism and presenteeism 4.5 Part B Results 4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.1 Part A Results 5.1.4 Study measurements 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Participant satisfaction 5.3 Part B Design 5.3.1 Study measurements 5.3.1.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.5 Part A and Part B Discussion Chapter 6 Study 4 Study design 6.1.1 Participant screening and selection 6.1.1.1 Inclusion criteria 6.1.1.2 <t< td=""><td>on72</td></t<>	on72
4.4.1.1 Absenteeism and presenteeism. 4.5 Part B Results. 4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.5 Analyses 5.1.6 Part A Results 5.1.7 Part A cosign 5.1.8 Study measurements 5.1.4 Study measurements 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Part A Results 5.2.1 System usability 5.2.2 Part B Results 5.3.1 Observer agreement 5.3.2 Analyses 5.3.1.1 Observer agreement 5.3.2 Part B Results 5.4 Part B Results 5.5 Part A and Part B Discussion Chapter 6 Study 4	
4.4.1.1Absenteeism and presenteeism.4.5Part B Results.4.6Part B DiscussionChapter 5 Study 3 (Part A, Part B).5.1Part A Design5.1.1Recruitment.5.1.2Smartphone application5.1.3Procedure5.1.4Study measurements5.1.5Analyses5.2Part A Results.5.3Part B Results.5.3.1Study measurements5.3.2Part A Results.5.3Part B Design5.3.1Study measurements5.3.2Analyses5.3Part B Design5.3.1Observer agreement5.3.2Analyses5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	nents
4.5 Part B Results	m and presenteeism76
4.6 Part B Discussion Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.4.1 System usability 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Part a Results 5.2.1 System usability 5.2.2 Participant satisfaction 5.3 Part B Design 5.3.1 Study measurements 5.3.1.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.3.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.5 Part A and Part B Discussion Chapter 6 Study 4	
Chapter 5 Study 3 (Part A, Part B) 5.1 Part A Design 5.1.1 Recruitment 5.1.2 Smartphone application 5.1.3 Procedure 5.1.4 Study measurements 5.1.4 Study measurements 5.1.4 System usability 5.1.5 Analyses 5.2 Part A Results 5.2.1 System usability 5.2.2 Part A Results 5.2.1 System usability 5.2.2 Part B Design 5.3 Part B Design 5.3.1 Study measurements 5.3.1.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.3.1 Observer agreement 5.3.2 Analyses 5.4 Part B Results 5.5 Part A and Part B Discussion Chapter 6 Study 4 6.1 6.1.1 Inclusion criteria 6.1.2 Randomisation and allocation concealment 6.1.3 Study measurements	on78
5.1Part A Design5.1.1Recruitment5.1.2Smartphone application5.1.3Procedure5.1.4Study measurements5.1.4Study measurements5.1.4.1System usability5.1.4.2Participant satisfaction5.1.5Analyses5.2Part A Results5.2.1System usability5.2.2Part a Results5.3Part B Design5.3.1Study measurements5.3.1Study measurements5.3.2Analyses5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	rt B)
5.1.1Recruitment	
5.1.2Smartphone application5.1.3Procedure5.1.4Study measurements5.1.4.1System usability5.1.4.2Participant satisfaction5.1.5Analyses5.2Part A Results5.2.1System usability5.2.2Participant satisfaction5.3Part B Design5.3.1Study measurements5.3.1.1Observer agreement5.3.2Analyses5.4Part B Results5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	
5.1.3Procedure5.1.4Study measurements5.1.4.1System usability5.1.4.2Participant satisfaction5.1.5Analyses5.2Part A Results5.2.1System usability5.2.2Participant satisfaction5.3Part B Design5.3.1Study measurements5.3.1Observer agreement5.3.2Analyses5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Exclusion criteria6.1.3Study measurements	lication
5.1.4Study measurements5.1.4.1System usability5.1.4.2Participant satisfaction5.1.5Analyses5.2Part A Results5.2.1System usability5.2.2Participant satisfaction5.3Part B Design5.3.1Study measurements5.3.2Analyses5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Exclusion criteria6.1.3Study measurements	
5.1.4.1System usability	nents
5.1.4.2Participant satisfaction5.1.5Analyses5.2Part A Results5.2.1System usability5.2.2Participant satisfaction5.3Part B Design5.3.1Study measurements5.3.2Analyses5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Exclusion criteria6.1.3Study measurements	bility
5.1.5Analyses5.2Part A Results5.2.1System usability5.2.2Participant satisfaction5.3Part B Design5.3.1Study measurements5.3.2Analyses5.4Part B Results5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	satisfaction
 5.2 Part A Results	
 5.2.1 System usability	
5.2.2Participant satisfaction5.3Part B Design5.3.1Study measurements5.3.1.1Observer agreement5.3.2Analyses5.4Part B Results5.4.1Observer agreement5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Inclusion criteria6.1.2Exclusion criteria6.1.3Study measurements	y
 5.3 Part B Design	faction
5.3.1Study measurements5.3.1.1Observer agreement5.3.2Analyses5.4Part B Results5.4.1Observer agreement5.5Part A and Part B DiscussionChapter 6 Study 46.1Study design6.1.1Participant screening and selection6.1.1.2Exclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	
5.3.1.1Observer agreement	nents
 5.3.2 Analyses	greement
 5.4 Part B Results	
 5.4.1 Observer agreement	
 5.5 Part A and Part B Discussion Chapter 6 Study 4 6.1 Study design 6.1.1 Participant screening and selection 6.1.1.1 Inclusion criteria 6.1.2 Exclusion criteria. 6.1.2 Randomisation and allocation concealment 6.1.3 Study measurements 	nent
Chapter 6 Study 46.1Study design6.1.1Participant screening and selection6.1.1.1Inclusion criteria6.1.1.2Exclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	B Discussion
6.1Study design6.1.1Participant screening and selection6.1.1.1Inclusion criteria6.1.1.2Exclusion criteria6.1.2Randomisation and allocation concealment6.1.3Study measurements	
 6.1.1 Participant screening and selection	
 6.1.1.1 Inclusion criteria	ening and selection
 6.1.1.2 Exclusion criteria	riteria
6.1.2 Randomisation and allocation concealment6.1.3 Study measurements	criteria
6.1.3 Study measurements	and allocation concealment95
	nents
6.1.4 Primary outcome	ıe96

6.1.4.1	Number of courses of intravenous antibiotics and intravenous antibiotic days	96
6.1.5	Secondary outcomes	96
6.1.5.1	Other measures of healthcare utilisation	96
6.1.5.2	Smartphone application adherence	97
6.1.5.3	Smartphone application symptom responses	97
6.1.5.4	System usability – System Usability Scale	97
6.1.5.5	Spirometry	97
6.1.5.6	Nutritional status - body mass index	98
6.1.5.7	Health-related quality of life – Cystic Fibrosis Questionnaire – Revised	98
6.1.5.8	Feelings of anxiety and depression – Hospital Anxiety and Depression Scale	98
6.1.5.9	Treatment adherence – Treatment Adherence Questionnaire – Cystic Fibrosis	98
6.1.5.10	Absenteeism/presenteeism – World Health Organisation's Heath and Work Performance Questionnaire	98
6.1.6	Intervention group	99
6.1.7	Control group	100
6.1.8	Analyses	100
6.1.8.1	Sample size calculation	100
6.1.9	Data privacy	100
6.2	Results	102
6.2.1	Participant characteristics	102
6.2.2	Number of courses of intravenous antibiotics and intravenous antibiotic days	105
6.2.3	Secondary outcomes	105
6.2.3.1	Other measures of healthcare utilisation	105
6.2.3.2	Smartphone application adherence	109
6.2.3.3	Smartphone application symptom responses	109
6.2.3.4	Other secondary outcomes	112
6.3	Discussion	120
Chapter 7 Sur	nmary and conclusions	123
7.1	Telehealth for adults with cystic fibrosis living in rural and remote Western Australia	. 124
7.2	The working profile of adults with cystic fibrosis living in rural and remote Western Australia.	127
7.3	A smartphone application for reporting respiratory symptoms in adults with cystic fibrosis	128
7.4	Future directions	131
References		132
Appendices		144

LIST OF FIGURES

Figure 3.1 Locally Weighted Scatterplot Smoothing for percentage predicted forced expiratory volume in 1 second (a) and percentage predicted forced vital capacity (b) for participants living in metropolitan and rural and remote areas of Western Australia	56
Figure 6.1 Consolidated Standards of Reporting Trials (CONSORT) diagram	104
Figure 6.2 Time to detection of respiratory exacerbation requiring a) intravenous antibiotics; and b) oral or intravenous antibiotics plotted against time (days)	107
Figure 6.3 Intravenous and oral antibiotic use in the 12 months prior to randomisation and during the study for a) participants in the intervention group who were ≥ 80%; and b) < 80% adherent to the weekly use of the smartphone	
application	110

LIST OF TABLES

Table 3.1 Participant characteristics at the time of their first occasion of spirometry for those living in the Perth metropolitan area and rural and remote Western Australia	55
Table 4.1 Participant characteristics at commencement of the study	65
Table 4.2 Healthcare utilisation data for participants who completed the intervention, comparing the 12-month periods prior to and during telehealth	66
Table 4.3 Participant responses to questions from the Telehealth Satisfaction Scale	67
Table 4.4 Participant responses to questions from the purpose developed satisfaction survey, with data presented as number (percentage) of participants	69
Table 4.5 Spirometry measured prior to telehealth and at the end of the 12-month telehealth study period for participants who completed the study	70
Table 4.6 Participant responses to the Cystic Fibrosis Questionnaire-Revised prior to and following telehealth intervention	71
Table 5.1 Smartphone application questions	81
Table 5.2 Participant characteristics	85
Table 5.3 Participant responses to items of the System Usability Scale	86
Table 5.4 Participant responses to the purpose developed satisfaction survey	87
Table 6.1 Participants' characteristics at baseline	103
Table 6.2 Participants' healthcare utilisation data	106
Table 6.3 Healthcare utilisation costs	108
Table 6.4 Smartphone application questions and response data for participants in the intervention group	111
Table 6.5 Spirometry and body mass index measured prior to randomisation and 12 months following randomisation	114
Table 6.6 Participant responses to the Cystic Fibrosis Questionnaire-Revised measured prior to randomisation and 12 months following randomisation	115
Table 6.7 Participant responses to the Hospital Anxiety and Depression Scale prior to randomisation and 12 months following randomisation	116
Table 6.8 Treatment Adherence Questionnaire – Cystic Fibrosis scores for participants with complete data prior to and 12 months following randomisation	118
Table 6.9 Absenteeism and presenteeism scores for participants with complete data prior to and 12 months following randomisation	119

LIST OF APPENDICES

Appendix 1 Author attribution statement
Appendix 2 Telehealth clinics for adults with cystic fibrosis improve access to care
Appendix 3 The impact of cystic fibrosis on work attendance and performance in adults living in rural and remote Western Australia
Appendix 4 Telehealth Satisfaction Survey and purpose developed satisfaction survey 176
Appendix 5 High usability of a smartphone application for reporting symptoms in adults with cystic fibrosis
Appendix 6 Observer agreement example clinical case scenario 197
Appendix 7 CONSORT diagram 198
Appendix 8 A smartphone application for reporting symptoms in adults with cystic fibrosis: protocol of a randomised controlled trial
Appendix 9 A smartphone application for reporting symptoms in adults with cystic fibrosis improves the detection of exacerbations: Results of a randomised controlled trial
Appendix 10 Modified Treatment Adherence Questionnaire – Cystic Fibrosis

PUBLICATIONS ARISING AS PART OF THIS THESIS

Wood, J, Mulrennan, S, Hill, K, Cecins, N, Morey, S, Jenkins, S. Telehealth clinics increase access to care for adults with cystic fibrosis living in rural and remote Western Australia. J Telemed Telecare. 2017;23(7):673–679.

Wood J, Jenkins S, Mulrennan S, Hill K. The impact of cystic fibrosis on work attendance and performance in adults living in rural and remote Western Australia. J Cyst Fibros. 2017;16(2):e1-e2.

Wood J, Jenkins S, Mulrennan S, Cecins N, Hill K. High usability of a smartphone application for reporting symptoms in adults with cystic fibrosis. J Telemed Telecare. 2017;24(8):547-552.

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey S, Cecins N, et al. A smartphone application for reporting symptoms in adults with cystic fibrosis: protocol of a randomised controlled trial. BMJ Open. 2018;8(4):e021136.

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey S, Cecins N, et al. A smartphone application for reporting symptoms in adults with cystic fibrosis improves the detection of exacerbations: Results of a randomised controlled trial. J Cyst Fibros. 2019;S1569-1993(19)30887-2.

PUBLICATIONS ARISING USING DATA FROM THIS THESIS

Lian R, Cavalheri V, **Wood J**, Jenkins S, Straker LM, Hill K. Higher levels of education are associated with full-time work in adults with cystic fibrosis. Respir Care. 2019;64(9):1116-1122.

ABSTRACTS RELATING TO THIS THESIS

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey D, Cecins N, et al. A smartphone application for reporting symptoms in adults with cystic fibrosis: a randomised controlled trial. J Cyst Fibros. 2018;7(3):WS12.5, pS22–S23.

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey D, Cecins N, et al. Adherence to a smartphone application used for reporting symptoms in adults with cystic fibrosis. Respirology. 2018;23(S1):TP065, p140.

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey D, Cecins N, et al. Adherence to a smartphone application used for reporting symptoms in adults with cystic fibrosis. Proceedings of the 12th Australasian Cystic Fibrosis Conference. 2017;P38.

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey D, Cecins N, et al. A smartphone application for reporting symptoms in adults with CF has excellent usability and its data is interpreted consistently between clinicians. Am J Respir Crit Care Med. 2016;193:A1090.

Wood J, Jenkins S, Hill K. The working profile and performance of adults with cystic fibrosis living in rural and remote Western Australia. Respirology. 2016;21(S2);TP180, p168.

Wood J, Mulrennan S, Morey S, Cecins N, Hill K, Jenkins S. Provision of telehealth improves clinic attendance to meet recommended standards of care in adults with CF living in rural and remote Western Australia. Respirology. 2016;21(S2):TO087, p65.

Wood J, Mulrennan S, Morey S, Cecins N, Hill K, Jenkins S. Telehealth clinics for adults with cystic fibrosis living in rural and remote Western Australia. Proceedings of the 11th Australasian Cystic Fibrosis Conference. 2015:P22.

INVITED PRESENTATIONS

Thoracic Society of Australia and New Zealand Annual Scientific Meeting 2018: "Using modern technology to optimise clinical care in cystic fibrosis" (March 23rd to 27th, Adelaide, Australia).

Australasian Cystic Fibrosis Conference 2017: "Opportunities for telemedicine and technology in cystic fibrosis" (August 5th to 8th, Melbourne, Australia).

AWARDS AND GRANTS

Thoracic Society of Australia & New Zealand and the Australian Annual Scientific Meeting 2019, Gold Coast, Queensland – Physiotherapy Special Interest Group Prize: best oral presentation.

European Cystic Fibrosis Conference 2018, Belgrade, Serbia – Best Physiotherapy Abstract Award.

Western Australian Health Excellence Awards 2016 Winner – Overcoming inequities: telehealth improves access to care for adults with cystic fibrosis living in rural and remote Western Australia.

Curtin University Faculty of Health Sciences Mark Liveris Seminar 2016 – best oral presentation.

Thoracic Society of Australia & New Zealand and the Australian Annual Scientific Meeting 2016, Perth, Western Australia – Physiotherapy Special Interest Group Prize: best oral presentation.

Principal Investigator: Sir Charles Gairdner Hospital Research Advisory Committee Grant 2015: "A pilot study and randomised controlled trial investigating the usability of a smartphone application, and its impact on the number of exacerbations requiring intravenous antibiotics in adults with cystic fibrosis – \$45,000.

Associate Investigator: Institute for Respiratory Health Glenn Brown Memorial Grant 2013: "Telehealth for adults with cystic fibrosis living in rural and remote Western Australia" – \$50,000.

LIST OF ABBREVIATIONS

3MST	3-minute step test
AAD	Adaptive aerosol delivery
ACT	Airway clearance technique
App	Application
ATS	American Thoracic Society
AUD	Australian dollars
BMI	Body mass index
CF	Cystic fibrosis
CFQ-R	Cystic Fibrosis Questionnaire-Revised
CFRD	Cystic fibrosis-related diabetes
CFRSD	Cystic Fibrosis Respiratory Symptom Diary
CFTR	Cystic fibrosis transmembrane regulator
CHQ	Child Health Questionnaire
CI	Confidence interval
COPD	Chronic obstructive pulmonary disease
CPET	Cardiopulmonary exercise test
CRDQ	Chronic Respiratory Disease Questionnaire
СТ	Computed tomography
eICE	Early Intervention in Cystic Fibrosis Exacerbation
EQ-5D	EuroQol-5D
FEV_1	Forced expiratory volume in 1 second
FFM	Fat free mass
FVC	Forced vital capacity
HADS	Hospital Anxiety and Depression Scale
HCU	Healthcare utilisation
HPQ	Health and Work Performance Questionnaire
HR	Hazard ratio
HRQOL	Health-related quality of life
IQR	Interquartile range
IRR	Incidence rate ratio
IV	Intravenous
LCI	Lung clearance index
MET	Metabolic equivalent of task
MVPA	Moderate-vigorous physical activity
OR	Odds ratio
$ppFEV_1$	Percent predicted forced expiratory volume in 1 second
ppFVC	Percent predicted forced vital capacity
RCT	Randomised controlled trial
rhDNase	Recombinant human deoxyribonuclease

SCGH	Sir Charles Gairdner Hospital
SD	Standard deviation
SF-36	Short Form 36
SpO2	Oxygen saturation
SUS	System Usability Scale
TAQ-CF	Treatment Adherence Questionnaire – Cystic Fibrosis
TeSS	Telehealth Satisfaction Scale
TIDES	International Depression/Anxiety Epidemiological Study
UK	United Kingdom
US	United States
USD	United States dollars
WA	Western Australia
WHO	World Health Organisation

CHAPTER 1

INTRODUCTION

This chapter outlines the background, aims and significance of this programme of research. The research undertaken is presented in this chapter in three sections. The first section relates to access to care for adults with cystic fibrosis (CF) living in rural and remote Western Australia (WA). Specifically, the role of telehealth, via videoconferencing, was explored as a way to improve access to care and other health outcomes. The research aims contained within this section correspond with the methods, results and discussion presented in Study 1 (Chapter 3) and Part A of Study 2 (Chapter 4). The second section relates to the working profile, in particular absenteeism and presenteeism, of adults with CF living in rural and remote WA. The research aims contained within this section correspond with the methods, results and discussion presented in Part B of Study 2 (Chapter 4). The third section relates to the identification of respiratory exacerbations in adults with CF. Specifically, the first two studies aimed to examine the usability of a smartphone application (app) for reporting symptoms and the observer agreement between clinicians interpreting the app data. Finally, the impact of the app on health outcomes was investigated in a randomised controlled trial (RCT). The research aims contained within this section correspond with the methods, results and discussion presented in Part A and Part B of Study 3 (Chapter 5), and Study 4 (Chapter 6).

1.1 Telehealth for adults with cystic fibrosis living in rural and remote Western Australia

This section provides the rationale for the research undertaken in Study 1 (Chapter 3) and Part A of Study 2 (Chapter 4).

1.1.1 Background

In the general population, people living in areas of low population density, such as rural and remote areas of Australia, have poorer health outcomes when compared with those living in metropolitan areas (1). A key contributor to this is reduced access to care, that is, the inability of a person to utilise health services that are necessary to improve or maintain their health. This is due in part to the lower numbers of health professionals with recently updated skills (in specialist clinical areas) working in rural and remote areas, and the large distances needed to travel to specialist healthcare facilities (2). Of note, earlier work has demonstrated that people with CF, who have complex healthcare needs (3, 4), have poorer health outcomes when not managed in specialist CF centres (5, 6).

In WA, the state adult specialist CF centre is at Sir Charles Gairdner Hospital (SCGH), in Perth. Of the 190 adults with CF who attended this centre at the time of commencing this study, 28 (15%) lived outside Perth, with distances to the centre ranging between 100 and 2,567 km. Due to limited specialist CF resources in these rural and remote areas, these adults with CF travel long distances to attend their scheduled routine clinics at SCGH. The geographical, time and financial barriers associated with travelling long distances is a major contributor to low rates of attendance at CF clinics.

The Australian CF Standards of Care recommend that people with CF attend CF clinics with a specialist multidisciplinary CF team a minimum of four times per year (4). In the years preceding this study, an internal audit revealed that this recommendation was not met for adults with CF living in rural and remote WA. This may negatively impact on long-term health outcomes, as it is known that those with CF receiving care in a CF centre have better forced expiratory volume in 1 second (FEV₁) and weight compared to those that do not (5, 6).

2

Therefore, Study 1explored the differences in magnitude of decline in spirometry between adults with CF living in the Perth metropolitan area and in rural and remote areas, and Study 2 (Part A) examined the use of telehealth (via videoconferencing) in the provision of CF clinics for adults with CF living in rural and remote WA.

1.1.2 Research questions

Do adults with CF living in rural and remote areas of WA have a different magnitude of decline in spirometry over a 3-year period compared with those living in the Perth metropolitan area?

In adults with CF living in rural and remote WA:

- i. What is the uptake of and satisfaction with telehealth clinics?
- ii. What is the impact of telehealth clinics on healthcare utilisation (HCU) outcomes including CF clinic attendance, hospital admissions and antibiotic use?
- iii. What is the impact of telehealth clinics on other health outcomes including spirometry, nutritional status and health-related quality of life (HRQoL)?

1.1.3 Significance

Adults with CF living in rural and remote WA are at a geographical, time and financial disadvantage in regard to attending the SCGH CF Centre. Further, people living in areas of low population density often have limited access to healthcare professionals with recently updated skills in specialist areas, such as CF. Describing differences in the magnitude of decline in spirometry over a 3-year period may provide a catalyst to improve services to people with CF living in these areas. The provision of CF clinics via telehealth, in the form of videoconferencing, is a novel intervention that has not been investigated in an Australian CF cohort. Telehealth may improve access to care, and have a subsequent positive impact on longer-term health outcomes, by slowing the magnitude of decline in spirometry over time, and optimising HRQoL.

1.2 The working profile of adults with cystic fibrosis living in rural and remote Western Australia

This section provides the rationale for the research undertaken in Part B of Study 2 (Chapter 4).

1.2.1 Background

As the expected age of survival increases, adults with CF will likely stay in the work force for longer. Previous studies have reported that, in adults with CF, employment status is related to age and time spent in hospital (7), lung function and HRQoL (8). While it is known that many (80%) have engaged in paid employment at some point in their adult life (9), there are minimal data available regarding the impact that CF has on work attendance (absenteeism) (9) and work performance (presenteeism) (10).

1.2.2 Research question

i. In adults with CF living in rural and remote WA, what are the levels of absenteeism and presenteeism?

1.2.3 Significance

Many adults with CF will need to maintain employment throughout their life. There are few data available regarding the impact of CF on absenteeism and presenteeism in the workplace. For the first time, this study will quantify absenteeism and presenteeism using the World Health Organisation's Health and Work Performance Questionnaire in a CF population. The results of this study will describe the work attendance and performance of adults with CF, providing an insight that may be used by other people with CF to inform future employment decisions.

1.3 A smartphone application for reporting respiratory symptoms in adults with cystic fibrosis

This section provides the rationale for the research undertaken in Part A and B of Study 3 (Chapter 5) and Study 4 (Chapter 6).

1.3.1 Background

Respiratory exacerbations can have a severe and long-lasting impact on people with CF, by accelerating the magnitude of decline in FEV_1 (11), increasing the time spent in hospital (12), reducing HRQoL (13, 14) and physical activity (15), increasing feelings of depression (16), and ultimately increasing the risk of mortality (17). A respiratory exacerbation has been defined as having a change in four or more disease-specific criteria comprising cough, sputum, haemoptysis, dyspnoea, fatigue, sinus pain and discharge, fever, reduced FEV₁, anorexia or weight loss, and changes in physical examination of the chest or chest radiography (18).

The early detection of respiratory exacerbations is an integral component of CF care. Up to 25% of people with CF who have a respiratory exacerbation fail to have their lung function completely recover following treatment (19). Delayed initiation of treatment (i.e. antibiotics) has been associated with this failure of lung function to recover to pre-exacerbation levels (19). Therefore, strategies that are targeted at the early identification of respiratory exacerbations may be a vital step in optimising long-term outcomes in people with CF.

In people with CF who develop a respiratory exacerbation, delayed identification of the exacerbation is often due to delayed presentation to a CF clinic. The reasons for this delay are multifactorial, and include commitments such as work, study and family, as well as the barriers to attending a CF clinic such as travel and financial burden, a lack of education and understanding around the need to report respiratory symptoms sooner, and feelings of anxiety and depression associated with attending a CF clinic (20). The use of telehealth is becoming an increasingly popular way of reducing the barriers associated with accessing CF care.
The impact of telehealth technology in CF has been evaluated in interventions aimed at identifying respiratory exacerbations (21-25), providing increased support for people awaiting lung transplantation (26), providing exercise assessment and training (27, 28) and promoting physical activity (29). Although adults with CF are likely to be comfortable with the use of mobile technologies, the effectiveness of telehealth at detecting respiratory exacerbations has previously been compromised by suboptimal adherence (i.e. 10% to 67%) (21-25). This is likely due to the high reporting burden experienced by participants, such as daily reporting of respiratory symptoms and/or regular spirometry. Minimising this burden in order to optimise adherence is an important consideration when designing telehealth technology.

There is a need for a novel approach to facilitate the early identification and treatment of respiratory exacerbations in this population. For this reason, a smartphone application (app) was developed that can be used by adults with CF to report symptoms suggestive of a respiratory exacerbation to the CF team. To optimise adherence, the app was designed to exclusively utilise smartphone technology that was mobile and already owned by the majority of young adults in today's society. It was hypothesised that more frequent surveillance of symptoms would shorten the time to detect respiratory exacerbations, reducing the risk of more severe exacerbations requiring treatment with intravenous (IV) antibiotics.

1.3.2 Research questions

- i. In adults with CF, what is the usability and satisfaction of an app used to report symptoms suggestive of a respiratory exacerbation?
- ii. Amongst clinicians experienced in providing CF care, what is the observer agreement in clinical decision making based on the symptom responses obtained using this app?
- iii. In adults with CF, what is the impact of an app used to report symptoms suggestive of a respiratory exacerbation over a 12-month period on:
 - a) The number of respiratory exacerbations requiring IV antibiotics and IV antibiotic days.

- b) Measures of HCU (time to detection of first respiratory exacerbation requiring IV, oral and inhaled antibiotics, antibiotics costs, number and cost of hospital admissions, and number and cost of CF clinics attended).
- c) Spirometry, nutritional status, HRQoL, feelings of anxiety and depression, absenteeism and presenteeism, and treatment adherence.
- iv. In adults with CF, what is the adherence to and usability of this app over a 12-month period?

1.3.3 Significance

This app was a novel technology that aimed to overcome the barriers leading to delayed treatment, by enabling the participant to identify and report symptoms suggestive of a respiratory exacerbation without attending a CF clinic. The app was downloaded on the participant's own smartphone, allowing for integration into their lifestyle more seamlessly. The majority of home monitoring interventions investigated in past studies have noted somewhat poor adherence amongst users, likely due to the high reporting burden using monitoring equipment that was large and not portable outside of the home.

Parts A and B of Study 3 were designed to ensure the usability of the app was high, and that there was agreement in clinical decision making by those interpreting the app data (i.e. demonstrating feasibility). Study 4 was designed to investigate the impact of the app on health outcomes, as well as to further examine its usability and the level of adherence to its use. It was contended that the use of the app would result in faster detection of respiratory exacerbations and initiation of treatment in adults with CF. This in turn would lead to less severe exacerbations and decrease the need for treatment with IV antibiotics. Further, improved surveillance and detection of respiratory exacerbations would possibly lead to improved longerterm outcomes such as spirometry and HRQoL.

1.4 Summary

This programme of research investigated novel solutions utilising modern technology. The global aim was to use telehealth to improve access to care and facilitate the detection of respiratory exacerbations amongst adults with CF, with the intention of meeting recommended standards of care and ultimately improving health outcomes The telehealth interventions investigated, and subsequent results, provide a substantial, original and significant contribution to the knowledge and understanding of this field, being the integration of modern technology with traditional care for adults with CF.

CHAPTER 2

LITERATURE REVIEW

This literature review comprises five parts, which each part containing several sections.

Part 1 describes the incidence, pathophysiology, morbidity and mortality of cystic fibrosis (CF); Part 2 discusses CF care in detail, including the evidence for centre-based care, the treatment burden experienced by people with CF and the levels of adherence to treatment regimens, and adult CF care in Western Australia (WA); Part 3 explores the definition, impact, identification and treatment of respiratory exacerbations in CF; and, Part 4 describes the role of telehealth in CF care to date; and Part 5 provides a conclusion relating to the literature covered in this review.

2.1 Cystic fibrosis: incidence, pathophysiology, morbidity and mortality

This part contains two sections. Specifically, i) the incidence, prevalence and cost of CF; and, ii) the pathophysiology, morbidity and mortality associated with this condition, and the monitoring of disease trajectory.

2.1.1 Incidence, prevalence and cost

Cystic fibrosis is one of the most common genetic diseases affecting the Caucasian population (30). In Australia, the incidence of CF is 1 in 3,000 live births (31). Currently, there are approximately 3,150 people with CF living in Australia (31) (out of a national population of 25,647,003) (32) and approximately 90,000 people with CF worldwide (33). Previously considered to be largely a paediatric disease, advancements in treatment options and models of care have seen the median life expectancy of those with CF in Australia increase in recent decades to 44 years currently (31). The median age of survival shows geographical disparities, with differences seen between the United States (US) (43 years), United Kingdom (UK) and Germany (47 years), and Canada (53 years) (33). While it is difficult to determine, it has been suggested that differences in survival age between countries may be due differences in healthcare systems and access to care (34). This increase in survival has resulted in a greater proportion of people with CF living into adulthood, with adults now comprising 54% of the total CF population in Australia (31).

With the recent development and availability of cystic fibrosis transmembrane regulator (CFTR) modulator therapies (medications that improve the production, processing and/or function of the defective CFTR) (35), the life expectancy of people with CF is expected to continue to increase. Trials investigating these new therapies have demonstrated reductions in mucus plugging and peribronchial thickening as seen on computed tomography (36), as well as clinically important improvements in lung function and quality of life (37-39). This poses an interesting future dilemma for healthcare; people with CF will have, on average, milder disease severity but the number of people with CF will

grow rapidly as a result of fewer premature deaths, potentially placing increased strain on CF centres and the healthcare system.

An analysis of Australian CF Registry data from 2003 to 2005 (of 2,255 cases, published in 2013) estimated that the mean annual cost per person for providing CF care ranged between \$14,700 (Australian dollars; AUD) and \$48,800 AUD, with the cost increasing with more severe disease (40). The lifetime cost per person was estimated to be \$443,400 AUD in this same sample (40). These estimates reflect the cost of medications and hospital resources (staffing, equipment, clinic rooms, and bed days) utilised in both outpatient and inpatient settings. The high cost of care for adults with CF in WA reflects these data; calculations extrapolated from unpublished data obtained from the Department of Activity Based Costing at Sir Charles Gairdner Hospital (SCGH) for the financial year 2016/17, demonstrated that the approximate total cost to the hospital for inpatient and outpatient care was \$4,771,465 AUD and \$635,807 AUD, respectively. With 185 adults attending the CF centre at SCGH during that financial year, the mean cost per person equated to \$29,228 AUD.

More recently in 2018, a retrospective analysis of 1,196 people with CF (mean [standard deviation, SD] age 16 [9] years) in the US found that 90% had at least one respiratory exacerbation in the 12month study period, at an estimated cost of \$44,589 (United States dollars; USD) to \$116,169 USD per exacerbation (41). Of note, US costing estimates are likely to be greater than those in Australia due to the differences in healthcare funding (i.e. insurance vs. government).

Similar economic evaluations undertaken in 2020 would look vastly different given the cost of CFTR modular therapies. Each of the widely used modulator therapies costs the Australian government approximately \$270,000 AUD per person annually (42), which will increase the estimated lifetime cost of care for a person with CF by 30 to 40 times. However, these are simplistic estimates that do not factor in the expected reduction in healthcare utilisation resulting from the expected improved outcomes relating to CFTR modulator therapies. Further, the true cost of CF to society is likely to be more accurate when additional factors are considered such as absenteeism and presenteeism relating to school, university and work, as well as the time and cost associated with disease-related travel to

and from healthcare facilities. These factors contribute to a secondary financial impact on families and carers.

2.1.2 Pathophysiology, morbidity, mortality and monitoring disease trajectory

Cystic fibrosis is a multi-system disease, predominantly affecting the lungs, pancreas, intestinal tract and liver (43). It results from a defective gene on the seventh chromosome which is responsible for encoding human CFTR protein (43). The CFTR protein contained in epithelial cells functions as a channel for the movement of chloride ions across the cell membrane, regulating the movement of water and the production of protective layers of mucus within organs (43). Defective CFTR inhibits the ability of the cells to move chloride ions and thus water to the cell surface, leading to a dehydrated and thick cell mucus layer, negatively impacting the function of the cilia. In the lungs, this leads to a cycle of inflammation and infection, and permanent airway damage known as bronchiectasis (43). Secondary to the progression of bronchiectasis, respiratory complications experienced can include haemoptysis and pneumothorax (44). Throughout the lifespan, the progression of bronchiectasis eventually leads to respiratory failure, with this being the most common cause of death (45).

The extrapulmonary manifestations of CF include pancreatic destruction and an inability to produce digestive enzymes (known as pancreatic insufficiency) and insulin, meconium ileus in infancy, distal intestinal obstructive syndrome and liver disease (46). Many people with CF will also develop osteoporosis, CF-related diabetes (CFRD) and renal insufficiency (46). The psychological impact of CF also needs to be considered, with levels of anxiety and depression reported as being at least similar (47, 48) or higher (49, 50) than that observed in the general population.

Disease trajectory is most commonly assessed using spirometry, in particular forced expiratory volume in 1 second (FEV₁) (51, 52). Regular use of spirometry is well established in CF clinics and healthcare centres across both metropolitan and rural and remote areas. Based on measures of FEV₁ expressed as a percentage of that estimated in a healthy population (percentage predicted FEV₁ [ppFEV₁]), people with CF can be classified as having mild (FEV₁ > 80% predicted), moderate (40 to

80% predicted) or severe (< 40% predicted) lung disease. Analysis of 14,372 people with CF in the European Cystic Fibrosis Society Patient Registry demonstrated that factors which increased the odds of an FEV₁ decline, after adjustment for age, were low body mass index (BMI) (odds ratio [OR], 95% confidence interval [CI] 6.0, 5.0 to 7.3), chronic infection with *Pseudomonas aeruginosa* (2.4, 2.0 to 2.7), pancreatic insufficiency (2.0, 1.6 to 2.5) and presence of CFRD (1.8, 1.6 to 2.2) (53). Other epidemiological studies suggest that frequent exacerbations (11, 17, 54), female sex (55), genotype, poor childhood growth, infection with mucoid *Pseudomonas aeruginosa* and meconium ileus (56) also influence the progression of lung disease. Two smaller but often cited studies also draw attention to the role of centre-based CF care in optimising disease outcomes including FEV₁ (5, 6), which will be explored more thoroughly in Section 1.2.1.

Notwithstanding the widespread use of spirometry to monitor disease trajectory and quantify disease severity, measures of FEV₁ have several limitations. These relate predominantly to the effort required during the repeated forced expiratory manoeuvres to perform spirometry (52), which makes collection of these measures challenging during respiratory exacerbations and impossible in young children (e.g. under the age of 5 or 6 years). Further, FEV₁ can be highly variable and lack the sensitivity to detect small changes in lung function, especially in those with mild disease (51, 57, 58).

In light of these limitations, other outcomes are increasingly being utilised to provide guidance to clinicians regarding disease trajectory. Computed tomography (CT) is the gold standard for detecting structural lung disease in CF and CT scores have been correlated with several important measures in this population such as FEV₁ (59), lung clearance index (LCI) (60), respiratory exacerbation rates (61), health-related quality of life (HRQoL) (62) and, survival in those awaiting lung transplantation (63). The LCI requires less effort to measure, and appears to be more sensitive to changes in lung mechanics than spirometry, with data from 67 adults with CF demonstrating that the sensitivity to changes in lung function of LCI was 81%, compared with 63% for ppFEV₁ (64). Peak rate of oxygen uptake measured during cardiopulmonary exercise testing (CPET) has been shown to be an independent predictor of lung transplantation or death at 10-year follow-up (65).

Despite these findings, access to CT, CPET and LCI is more limited compared to spirometry, especially in rural and remote areas. Further, CT and CPET are far more expensive to undertake, and CT results in substantial radiation exposure (66). Thus, while being increasingly utilised for research purposes, the uptake of CT, CPET and LCI still remains limited in clinical practice, and therefore FEV₁ is likely to remain part of the routine assessment of people with CF for the foreseeable future.

2.2 Cystic fibrosis care

This part contains four sections. Specifically, i) centre-based care in the management of CF; ii) barriers to care for people with CF; and, iii) treatment burden and adherence; and, 4) adult CF care in WA.

2.2.1 Centre-based care

International standards of care recommend that people with CF attend specialist CF centres (4, 67), which are most often located within a tertiary healthcare setting. Each centre is recommended to have a specialist multidisciplinary team, consisting of medical, physiotherapy, pharmacy, nursing, dietetic, social work and psychology professionals (4, 67). Centralised care in tertiary centres also provides the opportunity for the CF team to collaborate with on-site specialist secondary services (e.g. intravascular access, endocrinology, immunology and hepatology). While the recommendations regarding centre-based care are generally accepted and agreed with by clinicians, it is often not practical to offer this care to all people with CF for geographical reasons, especially in countries with areas of low population density such as Australia.

Even in the absence of evidence from randomised controlled trials (RCTs), centre-based care is generally viewed as superior to decentralised care (i.e. managed in local healthcare facilities by general practitioners). There are however, two well-known retrospective studies that provide evidence to support a specialist model of care. Data from 97 adults with CF, published in 1998, demonstrated that those who were managed in a paediatric and then adult CF centre throughout their lifespan in the UK had a higher BMI which correlated with greater FEV₁ (r =0.52, p < 0.001) (6). Data published in 2009 in Belgium on 217 children with CF showed that those who were referred late to a specialist CF centre following diagnosis had a lower FEV₁ at age 13 years (mean [SD] 77 [22]% vs. 87 [19]%, p = 0.01) (5). While these were retrospective observational studies with relatively small samples in differing geographical locations, the results lend support to the need to provide a centralised and collaborative approach to the management of this complex disease.

Outreach services, whereby the clinical team from the CF centre visits the person with CF in their local community and/or collaborates with local healthcare professionals, have been implemented in Australia for some time. Children and adolescents (n =116) with CF living in Queensland and northern New South Wales receiving outreach services from the CF team at the Royal Children's Hospital in Brisbane reported their HRQoL (using the PedsQLTM and Cystic Fibrosis Questionnaire) to be similar as those managed at the CF centre (68). Children (n = 79) in rural and remote Victoria, and Tasmania, who received outreach services from the Royal Children's Hospital in Melbourne had similar FEV₁ and BMI to those attending the CF centre (69). While outreach services may not always be the most practical solution given the travel and time burden on the clinicians, these two studies support the continued role of the CF centre in providing care to people with CF living in rural and remote areas. The use of technology such as videoconferencing, which can be used to provide outpatient clinics, may help to reduce this burden on both the people with CF and the CF centre clinicians.

The Australian CF Standards of Care recommend that people with CF attend a minimum of four CF clinic per year for disease monitoring and evaluation of their treatment regimen (4). Further, guidelines are provided regarding the specific provision of outpatient and inpatient care at tertiary centres, as well as outreach clinics for people with CF living in rural and remote areas. As the most recent edition of the Australian CF Standards of Care was published in 2008, telehealth was not included as a potential way of providing care to people with CF. Also, several of the recommended standards are based on clinician consensus rather than robust published data. The delivery of adult CF care in WA prior to the commencement of the research undertaken as part of this thesis will be described in more detail in Section 2.2.4.

More recently, it has been acknowledged that future models of care need to consider different management approaches that incorporate newer, innovative technologies for improving disease monitoring, reducing the disease burden and improving access to care (3, 33). The emerging role of telehealth in CF will be further described in Section 2.4.2.

2.2.2 Barriers to care

As life expectancy and the proportion of people with CF living into adulthood increases (31, 33), so does the burden facing people with CF in regard to accessing health care and completing treatment regimens. The main barriers to accessing care are complex and include distance, time, financial and psychological factors (20, 70, 71).

The strong clinical recommendation to offer care through specialist CF centres creates a barrier to care for many people with CF, due to the distance required to travel to the centre. This challenge is further amplified in Australia and is reflected in the general population; that is, it is known that people living in areas of low population density, such as rural and remote areas of Australia, have poorer health outcomes when compared with those living in metropolitan areas (1). A key contributor to poorer health outcomes is the inability of people to utilise health services that are necessary to improve or maintain their health. This is due in part to the lower numbers of health professionals with recent clinical experience in these specialist areas, and as mentioned, the large distances needed to travel to specialist healthcare facilities (2). This is discussed in regard to WA in Section 2.2.4.

Several factors lead to time being perceived as a barrier, encompassing competing adult commitments such as family, study and work, combined with the expectation of attendance at regular outpatient clinics (4) and disease-related treatment burden (71) (described further in Section 2.2.3). It is also now a realistic expectation for most people with CF that enrolment in higher levels of education and ongoing employment are attainable (72). In reality, this expectation is somewhat more of a necessity (i.e. to support oneself financially) as more people with CF live into adulthood. While many people with CF report that CF adversely impacts on their ability to work (7, 8, 73), UK data from 2014 in 254 adults with CF demonstrated that 65% were engaged in higher education or employment (including being self-employed), with the median (interquartile range [IQR]) hours worked per week being 37 (22 to 40) (9). Similarly, French data from 2012 in 207 adults with CF reported that 76% were engaged in study or employment (74). While the ability for adults with CF to study and work is

undoubtedly positive, it does compete with the time available for completing treatments vital for maintaining clinical stability.

As mentioned, the necessity of attending four or more CF clinics per year at a CF centre is in itself a major lifestyle disruption. As each clinic is multidisciplinary in nature, attendance can take several hours, not including the time waiting for new prescriptions to be filled at the pharmacy. Further, the costs of attending include private or public transport, parking at the centre, paying for prescriptions for several medications, and taking unpaid time off work or using annual or sick leave. Often CF centres will not provide expensive equipment such as nebulisers and airway clearance devices free of charge, further adding to the financial strain on people with CF. Finally, psychological barriers to care can include denial of one's own disease status, not wanting to have poor health status confirmed by clinicians or objective measures such as spirometry, or an inability to engage with the CF centre due to overwhelming feelings of anxiety and depression (20).

While the barriers to receiving care for people with CF are prolific, technology-based solutions tailored to the demographic of this population are very appealing as a tool to attempt to overcome the barriers described. In particular, solutions that can minimise the geographical and time barriers are highly desirable in order to improve access to care and potentially improve health outcomes.

2.2.3 Treatment burden and adherence

As described in Section 2.2.2, people with CF have many competing commitments in daily life (work, study, family), constantly threatening their ability to adhere to a complex prescribed treatment regimen. For adults with CF, the daily treatment undertaken to maintain clinical stability involves airway clearance, inhaled therapy, supplemental nutritional intake, exercise, and oral medications including antibiotics and digestive enzymes (33, 75, 76). This is in addition to the regular (at least 3-monthly) CF clinics attended at the CF centre as described in Section 2.2.1.

Sawicki and colleagues investigated the treatment burden amongst 204 adults with CF (71), and observed that participants were completing a median (range) of 7 (0 to 20) treatments per day, for a

mean (SD) total of 108 (58) minutes. The majority of treatment was related to physiotherapy, specifically inhaled therapy (41 [31] minutes), airway clearance (29 [27] minutes) and exercise (29 [23]) minutes.

Given the amount of time consumed by the daily treatment regimen of people with CF, it is not surprising to see at least some level of non-adherence with treatment regimens. Sawicki and colleagues reported that 139 (68%) of participants completed an airway clearance technique (ACT) as part of their prescribed treatment regimen, however only 49% reported completing their ACT on the day of data collection. This is consistent with a literature review of treatment adherence in CF, finding adherence to ACT and exercise being 40 to 55%, with varying levels reported for other treatments (inhaled medications and pancreatic enzymes = 65 to 80%, antibiotics = 80 to 95%) (77).

Common reasons reported by adults with CF for not completing treatment include feeling healthier than usual (i.e. don't need to do the treatment), too busy/no time, forgetting, being too tired, the medications are too expensive, not wanting to complete treatment in public, and having other commitments (70, 78-80). While it is understandable that adherence to such a time-consuming treatment regimen will not be perfect, the consequences of not completing regular treatment are severe for people with CF and include increased rates of respiratory exacerbations, longer hospitalisation and poorer lung function (81).

This large treatment burden, combined with a busy lifestyle and the need to attend the CF centre regularly, can negatively impact adherence to essential treatment amongst people with CF. Therefore, strategies must be considered that aim to reduce some of this burden in order to facilitate improved adherence to treatment. This may come in the form of newer technologies that allow for care to be delivered with less impact on factors such as time, travel and finances.

2.2.4 Adult cystic fibrosis care in Western Australia

In WA, the adult CF centre is at SCGH in Perth. Approximately 190 adults with CF attend SCGH, with approximately 25 to 30 (13% to 16%) living outside of the Perth metropolitan area at any time.

The distance this cohort lives from the SCGH CF centre ranges between 100 and 2,567km, with as few as one adult with CF living in many of the rural and remote areas, in towns that can have populations as low as 1,200 people. With limited specialist CF resources in these areas, and the difficulty in providing outreach clinics across such a vast geographical area (approximately 2,500,000 km²) due to limited CF centre staffing and travel costs, adults with CF have travelled long distances (via car or plane) to attend their scheduled CF clinics at SCGH. The travel and financial burden for people with CF is suspected to be a major contributor to their previously observed low rates of attendance at clinics.

In 2012 (prior to the commencement of this PhD research), an internal audit revealed that only 22% of the adults living in rural and remote WA met the recommendation of attending four or more CF clinics per year, in contrast to 79% of adults living in the metropolitan area. This lack of routine clinical assessment is likely to have impacted on the CF team's ability to identify respiratory exacerbations and provide necessary treatment, potentially worsening health outcomes such as lung function, nutritional status and HRQoL in this group over time. Telehealth, in the form of providing outpatient clinics via videoconferencing, is a potential solution to this problem.

2.3 **Respiratory exacerbations in cystic fibrosis**

This part contains three sections. Specifically, i) the definition of a respiratory exacerbation in CF;

ii) the impact of respiratory exacerbations on the healthcare system and people with CF; and, iii) the identification and treatment of respiratory exacerbations.

2.3.1 Definition of a respiratory exacerbation in cystic fibrosis

Defining a respiratory exacerbation in CF is a point of contention (82-87), and thus a universally agreed definition is still lacking. Despite this lack of consensus, the rate of respiratory exacerbations is one of the key outcome measures used in clinical trials (88). This is because these events have a negative impact on health outcomes (13, 17, 54, 89, 90) and healthcare utilisation (HCU) (40, 41, 91) (discussed in Section 2.3.2), and an apparent ease of being measured (as a rate or frequency) when comparing the efficacy of a new intervention (92). In clinical practice, physicians often adopt a pragmatic approach to define respiratory exacerbations (rather than using a specific set of criteria), resulting in differing approaches being used between physicians, CF centres and in clinical trials (84-86). The lack of an agreed definition, and use of a wide-ranging list of potential outcomes to define a respiratory exacerbation. The provision of treatment is often relied upon as surrogate marker for a respiratory exacerbation, with exacerbation severity being judged on the level of treatment required (i.e. oral and inhaled vs. intravenous [IV] antibiotics). This provides further challenges to accurately defining respiratory exacerbations in people with CF.

One of the first clinical trials to define a respiratory exacerbation for research purposes investigated the effect of inhaled recombinant human deoxyribonuclease (rhDNase), also known as Pulmozyme [®]. In this double blind, placebo RCT involving 968 adults and children with CF, a respiratory exacerbation was defined as a change in four or more disease-specific criteria comprising cough, sputum, haemoptysis, dyspnoea, fatigue, sinus pain and discharge, fever, worsening FEV₁, anorexia or weight loss, and changes in physical examination of the chest or

chest radiography (18). The intervention demonstrated a reduction in the number of respiratory exacerbations in this group by 22% (relative risk [95% CI] 0.78 [0.57 to 1.06]) with once daily administration of rhDNase, and by 34% (0.66 [0.48 to 0.91]) with twice daily administration (18). Following publication, this particular definition of a respiratory exacerbation, known as Fuchs criteria, was more widely adopted.

In 2001, Dakin and colleagues sought to further define respiratory exacerbations by surveying physicians experienced in providing CF care throughout Australia (83). They also raised concern with the use of 'therapeutic intervention as a surrogate measure of exacerbations' due to the requirement that all physicians would need to be utilising the same therapeutic approach in clinical practice; and therefore aimed to evaluate the level of consensus via their survey. Participants were provided with a list of signs, symptoms and investigations (based on previous literature) and asked to rate their helpfulness (not helpful; slightly helpful; helpful; or very helpful) in detecting respiratory exacerbations according to clinical scenarios specific to paediatric and adult care of people with CF. Analysis of responses (n = 62) highlighted the variability between physicians, with consensus (defined as > 74% or < 26% of respondents rating the criteria as helpful or very helpful) achieved on only eight (50%) of the 16 criteria. While this study did little to further define a respiratory exacerbation in CF, it did draw attention to the need for wider consensus in both clinical practice and research.

Other studies have attempted to provide clarity on defining a respiratory exacerbation of CF over subsequent years. Analysis of data from 11,692 children and adults with CF from the Epidemiologic Study of CF (US and Canadian observational study) determined that the signs or symptoms with the strongest association with the initiation of treatment for respiratory exacerbations in people \geq 18 years of age (complete data available on n = 999) were a relative reduction in ppFEV₁ (OR [95% CI], 3.8 [2.7 to 5.3]), newly detected *Pseudomonas aeruginosa* (3.2 [1.0 to 10.1]), increased cough frequency (2.8 [1.4 to 5.8]), sinusitis (2.7 [1.6 to 4.9]), new crackles on auscultation (2.5 [1.7 to 3.7)] and haemoptysis (2.4 [1.8 to 3.4]) (87). In those \geq 6 years of age, there was a large increase in being prescribed treatment with antibiotics (oral or IV) when comparing those with any two criteria (48% increase in prescribed treatment) and

those with any three criteria (75% increase in prescribed treatment) of the 10 disease-specific criteria investigated. This indicated that the presence of three or more of the defined exacerbation criteria was a strong indicator for the need for treatment. The authors of the Epidemiologic Study of CF made recommendations regarding the definition of a respiratory exacerbation that were not too dissimilar to Fuchs criteria, being four out of a possible 12 signs and symptoms.

A report from the EuroCareCF Working Group, published in 2011, drew attention to the ability of the Fuchs criteria in discriminating differences in rates of respiratory exacerbations in a large multicentre RCT investigating the efficacy of hypertonic saline (93). This working group proceeded to recommended that the Fuchs criteria be used in future European clinical trials to identify respiratory exacerbations (82). They also provided a modified version of the Fuchs criteria that they intended to validate in the future, defining a respiratory exacerbation as the need for antibiotic treatment based on the presence of two out of six of the original signs and respiratory symptoms (change in sputum, increased cough, increased fatigue or lethargy, anorexia or weight loss, decreased relative $ppEV_1 \ge 10\%$, increased dyspnoea).

The earlier work that attempted to define a respiratory exacerbation in people with CF is overtly clinician-centric. In 2009, two groups of researchers sought to incorporate feedback from people with CF into the definition of a respiratory exacerbation. Abbot and colleagues interviewed 47 adults with CF (age ≥ 16 years) regarding the symptoms they experienced at the commencement of a respiratory exacerbation, as well as perceived changes in respiratory symptoms during recovery from the exacerbation (94). The participants reported a range of symptoms that they associated with the onset of a respiratory exacerbation, which appeared to be influenced by their underlying disease severity. Common themes across all disease severities (reported in $\geq 80\%$ of the sample) were the presence of tiredness and fatigue, disturbed sleep and increased cough and sputum. The main differences observed were that participants with mild underlying lung disease reported that they often felt unwell with general aches, cold symptoms, runny nose and sore throat; while those with more severe disease described a decrease in their ability to undertake activities and increased dyspnoea at rest or with light activities. There was greater consensus

across disease severities regarding the respiratory symptoms that participants associated with recovery; these were increased energy, less fatigue, better sleep, decreased cough and sputum, and improved mood.

Goss and colleagues also explored this theme, by undertaking 25 qualitative interviews with parents of young children with CF (n = 4), adolescents (age 12 to 17, n = 9) and adults (n = 12) with CF who (or their children) were currently being treated with oral or IV antibiotics for a respiratory exacerbation (95). The authors aimed to obtain commonalities in reported respiratory symptoms and then use these to create a new symptom diary for use in clinical practice and research. The respiratory symptoms reported as most 'bothersome' during a respiratory exacerbation were (percentage [%] of times cited) cough (70%), fatigue (43%), chest congestion (39%), difficulty breathing/shortness of breath (39%) and wheezing (39%). The three most commonly cited triggers for seeking treatment for a respiratory exacerbation were constant cough (36%), thick dark sputum (32%) and shortness of breath (28%). Eight respiratory symptoms (cough, chest tightness, difficulty breathing, wheeze, coughing up mucus, fevers and chills, fatigue) were selected for the new symptom diary known as the CF Respiratory Symptom Diary (CFRSD), which also assesses symptom severity. This diary has been used as an outcome in CF studies (95, 96).

Bringing together the themes explored in this section, a study published in 2014 explored the views of both CF multidisciplinary team members (n = 38) and adults with CF (n = 31) in regard to signs and symptoms suggestive of a respiratory exacerbation (97). McCourt and colleagues conducted parallel Delphi surveys across 13 centres in the UK and Ireland. The five highest ranked responses from the clinicians (increased sputum, decrease in relative ppFEV₁ \geq 10%, increased shortness of breath, increased inflammatory markers, fever or increased temperature) and adults with CF (decrease in relative ppFEV₁ \geq 10%, increased shortness of breath, increase in relative ppFEV₁ \geq 10%, increased shortness of breath, increase in relative ppFEV₁ \geq 10%, increased shortness of breath, increase in relative ppFEV₁ \geq 10%, increased shortness of breath, increase in relative ppFEV₁ \geq 10%, increased shortness of breath, increase in relative ppFEV₁ \geq 10%, increased shortness of breath, increase in relative ppFEV₁ \geq 10%, increased shortness of breath, trouble breathing, feeling the need to do more airway clearance, increased night symptoms) varied somewhat, highlighting the need for consideration of the views of both clinicians and people with CF when defining a respiratory exacerbation.

It is clear that the definition of a respiratory exacerbation in CF is a point of contention. This is in contrast to other respiratory diseases such as chronic obstructive pulmonary disease (COPD) (98) and asthma (99), where more delineated criteria are accepted. For future studies seeking to define respiratory exacerbations in CF cohorts, it would be reasonable to use signs and symptoms acknowledged by both clinicians and people with CF, combined with a level of pragmatism to ensure it is relevant in a real-world setting.

2.3.2 Impact of exacerbations

In CF, respiratory exacerbations have profound and long-lasting negative consequences which will be detailed in this section. Specifically, themes will be explored relating to i) HCU; ii) lung function, further respiratory exacerbations and survival; and, iii) HRQoL and mood; and iv) other impacts.

2.3.2.1 Healthcare utilisation and cost

According to the Australian CF Data Registry 2017 Annual Report (31), 734 (23%) of the 3,151 people with CF required treatment with IV antibiotics at least once for a respiratory exacerbation during that 12-month period. The data reported pertaining to oral and inhaled antibiotic use specifically for respiratory exacerbations were less clear, as they were captured as either 'continuous' or 'as needed', whereas IV antibiotics would likely have only been prescribed for an acute respiratory exacerbation. Taking this into account, over a 12-month period 'as needed' oral and inhaled antibiotics in isolation were prescribed to 1,090 (35%) and 406 (13%) people at least once.

Also using data from the Australian CF Data Registry, approximately one quarter (24%) of people with CF in Australia in 2017 had at least one hospital admission for respiratory causes (n = 753) (31). However, further interpretation of the registry data needs to be approached with caution, as there is no way to determine whether those listed as having no antibiotics or having no respiratory exacerbations, reflects an absence of data recorded and submitted to the registry. In regard to CF clinic attendances, the mean number of clinics was 5.9 for children and adolescents, and 5.4 for adults, yet there is no way to establish whether each CF clinic

attendance was routine only, or an additional attendance for monitoring of an acute respiratory exacerbation.

In comparison, the US CF Foundation Patient Registry reports that of 30,775 people with CF in the US in 2018, 10,439 (34%) experienced a respiratory exacerbation requiring treatment with IV antibiotics, and received a mean of 27.5 days of IV antibiotics per year per person, with an average of 17.9 days in hospital per person (100). The mean number of CF clinics attended per person was 4.3. In the UK, there were data from 10,509 people with CF in the UK Cystic Fibrosis Registry in 2017 (101). During this year, 4,400 (45%) people required IV antibiotics for a respiratory exacerbation, and 3,715 (38%) had at least one hospital admission, with the median (IQR) admission days per person throughout the year being 27 (14 to 45). No UK data were available in 2017 regarding the average number of CF clinic attendances per person.

While it is hard to estimate the exact cost of respiratory exacerbations in CF due to the complexity of the economics and burden involved, the use of 2018/19 Australian activity based funding (ABF) data provided by the Independent Hospital Pricing Authority (102) can provide a lower bound estimate of the costs associated with inpatient hospital care. Sir Charles Gairdner Hospital receives funding of between \$12,846 and \$19,294 AUD for each inpatient admission for a respiratory exacerbation (based on a 14-day admission). The use of ABF data as a surrogate marker of true hospital costs does have limitations. Specifically, ABF (which is a federal source of funding in Australia) potentially underestimates the true cost associated with hospital admissions as it does not include the other sources of hospital funding (e.g. state government, philanthropic or grant funding for clinical trials) that contribute to its ability to provide ongoing care. A different method of estimating the cost of an inpatient admission utilises figures provided by the SCGH Department of Activity Based Costing. Each inpatient day is calculated to cost the hospital \$1,309 AUD providing an estimate of \$18,326 AUD for a 14-day admission. This estimate also likely underestimates true costs, at it excludes the cost of medications and allied health services.

As discussed in Part 2.2, US data from 2018 estimated that the cost of care per respiratory exacerbation (receiving oral or IV antibiotics, with or without hospitalisation) was \$44,589 USD to \$116,169 USD (41). This study used Medicaid insurance information to provide estimates, which potentially inflated the costs when compared to healthcare systems that are not insurance driven. In contrast, another US group published data in 2017 and estimated the mean cost of providing treatment for respiratory exacerbations for people with CF to be \$12,784 USD for an exacerbation of any type (receiving oral or IV antibiotics, with or without hospitalisation), or \$36,319 USD for those treated with IV antibiotics (with or without hospitalisation) (91). This study, which was led by a pharmaceutical company, again estimated costs based on the data obtained from medical records and the database of a national healthcare insurer.

Overall, it is clear that respiratory exacerbations lead to variable, but generally high costs to the healthcare system through hospital admissions, cost of antibiotics and CF clinic attendances. With the advancement of pharmacological options for treatment, specifically CFTR modulators, the future projected costs of providing care for respiratory exacerbations may change, for example, if they are less frequent and/or less severe, and require fewer hospital admissions. However, this reduction in cost for the treatment of respiratory exacerbations will likely be overshadowed by the costs of the CFTR modulator treatments (approximately \$270,000 per person per year), and therefore efforts still need to be directed toward detecting respiratory exacerbations sooner to allow for the faster provision of antibiotic treatment.

2.3.2.2 Lung function, risk of further exacerbations and survival

As previously described (Section 2.1.2), FEV_1 has historically been the most common marker of disease trajectory in CF, with the immediate impact on FEV_1 widely accepted as an indicator of a respiratory exacerbation (18, 82, 83, 87). However, it is the medium to long-term impact of respiratory exacerbations on FEV_1 and survival that is more concerning.

In 2010, Sanders and colleagues reported on data that had been collected by the US Cystic Fibrosis Foundation Registry between 2003 and 2006 on 8,479 respiratory exacerbations in people with CF treated with IV antibiotics (19). Specifically, they analysed the proportion of participants (age \geq 6 years) in whom FEV₁ returned to baseline levels (defined as the best FEV₁ in the 6 months prior to initiation of treatment) within 3 months following treatment. In 25% of respiratory exacerbations, the participants' FEV₁ did not recover to baseline levels, which would likely lead to an accelerated loss of lung function over time. In 2017, the same authors used US Cystic Fibrosis Foundation Registry data from 2004 to 2011 (13,954 respiratory exacerbations) to report that the participants in whom FEV₁ had failed to return to baseline levels, described as non-responders (n = 2,762 [19.8%]), had a shorter median (95% CI) time to the next respiratory exacerbation, being 235 (218 to 252) days, vs. > 365 days in responders (hazard ratio [HR] [95% CI] 1.1 [1.1 to 1.2]) (103). Further, non-responders had a greater number of respiratory exacerbations over the following 3 years than responders, being mean (95% CI) 5.0 (4.8 to 5.1) vs. 3.5 (3.4 to 3.5), respectively.

Several other studies have explored the relationship between respiratory exacerbations and FEV₁. In a small Italian sample (n = 51), the number of respiratory exacerbations per year, quantified retrospectively, was strongly related to FEV₁ decline per year, with each exacerbation resulting in an additional mean (95% CI) loss of 30 (9 to 52) ml (89). In Canada, two studies reported similar findings; the first, a 3-year prospective cohort study (n = 446), demonstrated an increased risk of a 5% decline in FEV₁ from baseline in those with more than two respiratory exacerbations per year (HR [95% CI] 1.5 [1.1 to 2.0]) (17); the second, a retrospective study (n = 851), found that the mean (95% CI) rate of annual FEV₁ decline was double in those who had a respiratory exacerbation (2.5 [2.1 to 2.8]%) vs. those who had not had an exacerbation (1.2 [1.0 to 1.5]%) (54). The former Canadian study also demonstrated that participants who had experienced > 2 respiratory exacerbations requiring IV antibiotics per year were at increased risk of lung transplantation or death during the 3-year study period (HR [95% CI] 4.05 [1.15 to 14.28]) (17).

People with CF who experience multiple respiratory exacerbations requiring treatment with IV antibiotics in a year are at increased risk of having a respiratory exacerbation in the future, as follows. A prospective study which followed 249 people (97% adults) with CF for 4.5 years

found that having a several (≥ 4) respiratory exacerbations requiring treatment with IV antibiotics in the previous two years increased the odds of future respiratory exacerbations (OR [95% CI] 3.16 [1.93 to 5.17) (104). Similarly, data from 13,579 children and adults in the US demonstrated that the risk of having a respiratory exacerbation requiring treatment with IV antibiotics increased with the number of respiratory exacerbations experienced in the previous year (i.e. in those with 1, 2, 3 or \geq 4 exacerbations) (105).

Finally, one of the earliest epidemiological studies exploring survivorship models based on longitudinal data in 11,630 people with CF from the US demonstrated that a higher number of annual respiratory exacerbations predicted decreased 5-year survivorship (106). The results of this work, along with those presented earlier, highlight the impact that respiratory exacerbations have on FEV₁, the risk of future respiratory exacerbations and, ultimately, survival.

2.3.2.3 Health-related quality of life and mood

Measurements of HRQoL and feelings of anxiety and depression are often collected in CF research and are increasingly used in the clinical setting. Several different instruments have been used to measure HRQoL and mood in CF previously, with the Cystic Fibrosis Questionnaire-Revised (CFQ-R) (107) and Hospital Anxiety and Depression Scale (HADS) (108) seen most commonly in the published literature to date.

In 2013, Bradley and colleagues investigated the impact of changes in respiratory exacerbation status (categorised as no, mild [no hospitalisation] or severe [hospitalisation] respiratory exacerbation) on HRQoL using the CFQ-R and EuroQol (EQ-5D) in people with CF age ≥ 16 years (109). Participants from five CF centres in the UK were asked to complete measures during two CF clinic attendances, between 8 and 12 weeks apart. Of the 94 participants, 60 (64%) had no respiratory exacerbations, 15 (16%) had a mild respiratory exacerbation, and 19 (20%) had a severe respiratory exacerbation. Worse HRQoL scores on the CFQ-R and EQ-5D were observed for participants with a mild exacerbation compared to those with no exacerbation, and worse again scores were seen for participants with a severe exacerbation when compared with those with a mild or no exacerbation (HRQoL data not available).

In a cross-sectional observational study of 162 children and adults with CF from the US published in 2002, those with a recent respiratory exacerbation (defined as within the previous 6 months) had reduced physical and psychosocial summary scores on the Child Health Questionnaire (CHQ) and Short Form 36 (SF-36) (scores were combined from both measures) (13). Specifically, physical summary scores were worse when compared to a baseline score of mean (SD) 50 (7) out of 100 for participants with no exacerbations, if one (42 [12]), two (40 [10]) or three or more (32 [11]) respiratory exacerbations were experienced (all p < 0.01). Psychosocial summary scores fell from a baseline of 53 (7) for those with no exacerbations if one (50 [12]), two (47 [11]) or three or more (42 [11]) exacerbations were experienced (all p < 0.01). A difference in score of ≥ 5 is considered clinically important for the combined CHQ and SF-36 scores (13).

Bradley and colleagues explored the relationship between respiratory exacerbations requiring treatment with IV antibiotics in hospital and HRQoL (as measured by the Chronic Respiratory Disease Questionnaire [CRDQ]) in Northern Ireland as early as 2001 (110). In an adult sample (n = 18), mean (95% CI) CRDQ scores (also measured at the commencement of treatment) had improved by 20 (12 to 28) out of 100 at the cessation of treatment. Of note, this study also found that the relationship between FEV₁ and HRQoL was weak (although the correlation coefficient was not reported). The improvement in HRQoL result following IV antibiotic treatment is also supported by US data from 2010, in which mean (95% CI) scores on the Respiratory Scale of the CFQ-R (also measured at the commencement of treatment) improved by 12 (6 to 17) out of 100 in 52 children and adolescents with CF (111). A change in score of four is considered clinically important on this scale of the CFQ-R (112).

The International Depression/Anxiety Epidemiological Study (TIDES) included 154 centres across nine countries. Data from the UK centres extracted from this study sought to quantify feelings of anxiety and depression in people with CF using the HADS (47). In total, 2,065 adolescents (age \geq 12 years) and adults completed the HADS during CF clinics, including those who were currently experiencing a respiratory exacerbation, with data compared to a normative sample (1,788 healthy adults from the UK). In general, feelings of anxiety and depression in the

Chapter 2: Literature Review

participants with CF were similar to those in the general population, however adults with CF had slightly greater feelings of anxiety and depression than adolescents with CF. This was observed for both males (mean [SD] anxiety 5.7 [3.9] vs. 4.9 [3.2], p < 0.01, depression 3.4 [3.3] vs. 1.9 [2.0], p < 0.01) and females (anxiety 6.6 [4.3] vs. 5.7 [3.8], p = 0.01, depression 3.4 [3.4] vs. 2.4 [2.0], p < 0.01). Clinically, the combined mean anxiety and depression scores for adult males (9.1) and females (10.0) fell within the 'mild' range (8 to10) on the HADS. The combined scores for adolescent females (8.1) were just within the 'mild' range, while adolescent males (6.8) were in the normal range. In this study, for all adults, older age, unemployment for health reasons and poor lung function were associated with greater feelings of anxiety and depression.

While 130 (6%) of the TIDES sample were on IV antibiotics at the time the survey was administered, a regression analysis demonstrated that current IV antibiotic use was not associated with greater feelings of anxiety or depression (47). It was however, beyond the scope of that study to determine whether feelings of anxiety and depression had worsened from baseline in these participants specifically. Of note, the presence of recent haemoptysis, which is one of the widely accepted criteria used to define a respiratory exacerbation, was associated with greater feelings of anxiety and depression in women.

The findings of these studies, demonstrating the impact of respiratory exacerbations on HRQoL, confirms the importance of using the number or rate of respiratory exacerbations as an outcome in trials investigating interventions that aim to detect and/or prevent the occurrence of respiratory exacerbations. Telehealth studies with this aim will be discussed in detail in Section 2.4.2.2.

2.3.2.4 Other impacts (on physical activity, body mass index and complications of intravenous antibiotic use)

Other notable impacts of respiratory exacerbations are reduced physical activity (15), loss of weight and bone density (113, 114), and increased resistance to antibiotics through their repeated use (33).

In Australian adults with CF admitted to hospital with a respiratory exacerbation (n = 24), Ward and colleagues demonstrated that less time was spent in daily physical activity over a 24 hour measurement period in hospital compared to one month post discharge (mean [SD] time \geq 3 metabolic equivalent of task [METs] 95 [58] min/day vs. 209 [111] min/day) (15). As the exacerbation component of this study took place during an inpatient stay, the necessity of spending large amounts of time within the hospital room (to meet clinician review/treatment requirements) may have contributed to the decreased participation in physical activity.

Two studies have further explored the relationship between physical activity and respiratory exacerbations. In 2015, Savi and colleagues measured physical activity in 60 adults with CF from Italy and the UK with a SenseWear Pro3 Armband (115). Data were averaged over 5 days and reported with the participants grouped according to how many respiratory exacerbations they had experienced in the year preceding the study. The group of adults who had > 2exacerbations in the previous year spent less time in daily physical activity (> 3 METs) than their counterparts with fewer exacerbations; however, this group had a lower FEV_1 at baseline and the association between respiratory exacerbations and physical activity was no longer present when other baseline variables were controlled for in the analysis. Cox and colleagues examined whether levels of daily moderate-vigorous physical activity (MVPA) was a predictor of respiratory exacerbations requiring hospitalisation, and magnitude of change in lung function, over a prospective 3-year period in 53 adults with CF (116). Following a regression analysis, they found no association between levels of physical activity and respiratory exacerbations requiring hospitalisation. This was despite the study demonstrating that 30 mins of MVPA in bouts of at least 10 mins throughout the day was a predictor of slower magnitude of decline in FEV_1 over the 3-year period.

Data collected on 85 Italian adults with CF who underwent bone densitometry demonstrated that males with > two respiratory exacerbations in the 12 months prior to the scan had lower whole-body fat free mass (FFM) (divided by height squared and reported as FFM indices) when compared to those with fewer exacerbations (mean [SD] 18.0 [1.9] kg/m² vs. 19.3 [1.4] kg/m², p = 0.02) (113). Further, the same male participants also had lower total bone mineral density

measures as a Z-score (-1.4 [1.2] vs. -0.67 [0.9], p = 0.03). These results were not replicated in the female participants. Possible reasons for this include that the total sample of females was relatively small (n = 85) and that, on average, females (n = 41) had a poorer nutritional status compared to males at baseline when measured as BMI (mean [SD] 20 [3] kg/m² vs. 22 [2] kg/m²) and a greater proportion were underweight as defined by a BMI < 18.5kg/m² (number [%] 10 [24%] vs. 2 [5%]).

As previously discussed (Section 2.1.2), low BMI is a predictor of future FEV_1 decline (53). This relationship is likely to be due, at least in part, to the effect acute respiratory exacerbations have on both outcomes. During respiratory exacerbations, the higher metabolic demand (117) is often accompanied by a temporary loss of appetite (118). The temporal relationship between respiratory exacerbations and BMI is however somewhat unclear; that is, is lower BMI influenced by frequent respiratory exacerbations, or are frequent respiratory exacerbations influenced by lower BMI? The association is likely complex and synergistic.

While there are several potential risks associated with the regular use of oral, inhaled and IV antibiotics (including but not limited to allergy, bronchospasm, kidney and liver failure, IV access related complications), concerns regarding antibiotic resistance are prominent (33). The main reasons for increased antibiotic resistance are thought to be prolonged use as well as inappropriate choices of antibiotics by clinicians (33).

In CF, respiratory exacerbations clearly have a substantial negative impact on the person with CF and the healthcare system, adversely impacting on clinical trajectory, HRQoL and other outcomes. It is therefore essential to focus strategies on the early and accurate identification, and effective treatment of respiratory exacerbations.

2.3.3 Identification and treatment of respiratory exacerbations

The controversy surrounding the agreed definition of a respiratory exacerbation (discussed in Section 2.3.1) only increases the difficulty faced by clinicians regarding identifying and treating such exacerbations. Traditional CF care involves the identification of changes in respiratory symptoms during routine outpatient clinics (recommended at least 4 times per year), where

spirometry and assessment by the multidisciplinary CF team is completed (4). It is also not uncommon for people with CF, or their families or carers, to contact the CF team sooner than their next clinic appointment if they feel that they have worsening respiratory symptoms. It is considered important to identify and treat respiratory exacerbations promptly once worsening symptoms appear, with data supporting that delayed identification of respiratory exacerbations and initiation of treatment has been associated with failure of FEV_1 to return to pre-exacerbation levels (19). Although, these data were not collected prospectively, and further investigation of the relationship between symptom onset, presentation to a CF clinic and health outcomes is warranted.

The delayed identification of a respiratory exacerbation in people with CF is often due to delayed presentation to a CF clinic. The reasons for this delay are multifactorial, and include competing life commitments such as work, study and family. There are also barriers to attending a CF clinic, such as travel and financial burden, a lack of education and understanding around the need to report respiratory symptoms sooner, and feelings of anxiety and depression associated with attendance (20).

As discussed in Section 2.2.3, people with CF are expected to complete a daily treatment regimen designed to reduce airway obstruction via ACT and inhaled mucolytic therapy, maintain/increase physical activity and exercise capacity, optimise nutritional status, and prevent respiratory exacerbations with maintenance oral and inhaled antibiotics and vaccinations (33, 75, 76). The treatment of a respiratory exacerbation often involves increasing the frequency or time of some of these daily treatments, and commencing additional oral, inhaled or IV antibiotics, with or without hospitalisation. Complications associated with respiratory exacerbations (haemoptysis, pneumothorax, sinusitis, fungal disease, and distal intestinal obstructive syndrome) may also need concurrent treatment and/or result in modifications to the treatment plan (33, 75, 76, 119).

Similar to their definition, the optimal provision of treatment for respiratory exacerbations is not universally agreed (87, 93, 120-122). This is likely to be due to several factors. First, the

decision regarding whether to treat a respiratory exacerbation and what treatment to provide can vary between clinicians (84, 86). Second, people with CF may choose to delay or not accept particular treatments based on their current lifestyle, treatment burden and beliefs about treatments. Third, it is widely felt that in vitro antibiotic susceptibility testing of bacteria does not accurately represent the observed clinical response (123), and therefore best practice guidelines for the use of antibiotic therapy, based on observed clinical responses, are instead followed (75). Finally, the optimal duration of antibiotic treatment, in particular for IV administration, is not evidence based and therefore remains clinician and CF centre dependent (generally given for 10 to 14 days) (121, 122, 124, 125). The use of two IV antibiotics that work synergistically has been demonstrated to provide better outcomes than one antibiotic alone (75), however the best combination of IV antibiotics is still unclear (126).

Mild respiratory exacerbations are generally managed in the outpatient setting with oral and inhaled antibiotics (92, 127, 128). The addition or change of mucolytic therapy and ACTs may also be appropriate (92). If the exacerbation is deemed more severe, if the person is considered high risk (e.g. low baseline FEV₁, severe complication such as haemoptysis or pneumothorax), or if response to oral and inhaled antibiotic therapy has been suboptimal, treatment with IV antibiotics with or without hospitalisation will be considered. The cessation of treatment for a respiratory exacerbation is guided by an improvement in FEV₁ to or near baseline levels, and improvement in symptoms (75, 121, 122, 125). These goals may differ depending on the clinical scenario, and may also be guided by the preference or lifestyle factors of the person with CF.

Identifying respiratory exacerbations in people with CF is challenging due a number of factors which have been described in this chapter. This includes the controversy around defining an exacerbation, and the reliance on traditional methods of care for detecting changes in symptoms. Given the relatively younger demographic of adults with CF, it makes sense to consider the use of newer technologies such as smartphone applications to assist in the identification of respiratory exacerbations. This may facilitate the provision of necessary treatment sooner, potentially leading to improved longer-term health outcomes.

2.4 Telehealth

This part contains two sections. Specifically, i) the definition of telehealth; and, ii) the use of telehealth in CF, including for outpatient care, for home monitoring, to improve adherence to treatment, and to promote exercise, exercise testing, and physical activity.

2.4.1 Definition

There are well over 100 peer reviewed definitions of the term that is defined in this thesis as 'telehealth' (129), with telemedicine and eHealth also being commonly used. The World Health Organisation (WHO) supports the use of the terms telehealth and telemedicine interchangeably, and has adopted the broad description of telehealth as "The delivery of health care services, where distance is a critical factor, by all health care professionals using information and communication technologies for the exchange of valid information for diagnosis, treatment and prevention of disease and injuries, research and evaluation, and for the continuing education of health care providers, all in the interests of advancing the health of individuals and their communities" (130).

The WHO also goes on to define the key elements of telehealth as those which: i) provide clinical support; ii) overcome geographical barriers to connect users in different locations; iii) utilise various types of information and communication technologies; and aim to improve health outcomes (130). Specific types of modern technology that are already available for use in the delivery of telehealth include videoconferencing, text messaging and other forms of digital messaging and apps.

In countries with similar areas of low population density to Australia, such as the US and Canada, telehealth has been utilised for some time in an attempt to improve access to healthcare and patient outcomes by overcoming geographical barriers, and reducing the burden on the healthcare system in people with chronic respiratory diseases such as asthma (131, 132) and COPD (133-137). The use of telehealth is also increasing in chronic conditions such as heart failure and diabetes, with the aim of improving patient self-management and to deliver forms of home monitoring and rehabilitation (138-141).

With the increasing utilisation of newer information and communication technologies throughout society, it seems inevitable that many of these innovations will be integrated within the healthcare setting in the future. It is important however, that this is approached with some caution to ensure that a sufficient evidence base is established supporting the role of telehealth in improving access to care and health outcomes. Consideration also needs to be given to data protection and the privacy of participants when using telehealth technologies, and the need for access and connectivity to the relevant systems. These potential barriers need to be overcome to ensure the integration of modern technology with traditional care is a success.

2.4.2 Telehealth in CF

The demographic profile of adults with CF makes them an obvious case for the use of newer technologies in the provision of healthcare. While earlier work has corroborated that telehealth technology is, in general, acceptable to people with CF, consistent improvements in health outcomes have yet to be demonstrated, likely due to the variability in study designs and outcomes used (142). To date, health outcomes most commonly used to evaluate telehealth in CF pertain to respiratory exacerbations and the number of hospitalisations, magnitude of decline of FEV₁, HRQoL, feelings of anxiety and depression, and disease-related burden.

The use of telehealth in different aspects of CF care will be discussed in themes of outpatient care, home monitoring, adherence to recommended treatment, and the promotion of exercise, exercise testing and physical activity.

2.4.2.1 Improving access to outpatient care

The first work in CF to explore the use of videoconferencing for supportive care was conducted by Wilkinson and colleagues in the UK (26). The participants in this study were age > 16 years and listed for lung transplantation. A 6-month prospective pilot RCT was undertaken examining the use of videoconferencing to provide weekly physiotherapy support in between monthly routine face to face outpatient clinics. The impact on HRQoL, feelings of anxiety and depression, HCU, carer burden and satisfaction was measured. Sixteen participants (median [range] age 27 [21 to 41] years) were enrolled, of whom only seven completed the study with

the high attrition explained by death (n = 4) and lung transplantation (n = 3). The intervention group was provided with internet access and videoconferencing equipment, as well as a spirometer, oximeter and thermometer. These participants undertook weekly sessions with a physiotherapist (discussing aspects of care and reviewing the available clinical outcomes), with other members of the CF team available if necessary.

While no between-group differences were seen in feelings of anxiety and depression and HCU, both groups had increased feelings of anxiety and depression when compared to baseline (26). This finding may due to the fact that the participants were 6 months further into the time they had been listed for lung transplantation and potentially closer to death at the time of the follow-up assessments, rather than an effect from the interventions provided as part of the study. Satisfaction with the videoconferencing was high; with 100% of participants in the intervention group feeling that it was a 'good service' and expressing a desire for it to be continued. Participants in the intervention group also preferred videoconferencing to telephone calls or extra face to face follow-up. Despite the study being small with high levels of attrition (56%), it showed that videoconferencing could be used with high levels of satisfaction amongst participants, and was preferred over other modes of contact. This provides encouragement for the multidisciplinary CF team to explore the use of videoconferencing in the provision of care to people with CF, particularly those with high travel and time burden and living in rural and remote areas.

One other small, short-term study conducted in Israel explored the use of message and videoconferencing communication to provide additional multidisciplinary input in the outpatient setting over a 3-month period, on this occasion using WhatsApp and Skype (143). Eighteen children and adults with CF were allocated to intervention (n = 9, mean [SD] age 21 [6] years) or control (n = 9, age 25 [11] years) groups, with the intervention group receiving motivational and treatment related messages and videoconferencing calls from the multidisciplinary CF team. Despite the intervention group receiving a median (range) of 5 (4 to 6) Skype videoconferencing calls and between 22 and 45 WhatsApp messages per participant throughout the 3-month

period, no clear effect was demonstrated on HRQoL, adherence and satisfaction. This study was however unlikely to be powered to detect changes in these outcomes.

Of note, three (17%) of the participants in this study who had been assigned to the intervention group during randomisation changed to the control group as they did not want to, or were unable to, undertake the intervention due to busy schedules (n = 2) and lack of internet connection (n = 1) (143). The authors also described difficulties experienced with the scheduling of videoconferencing calls as the participants were required to be in their own home, and two or three attempts were often required to successfully undertake each session. Poor internet connection was also experienced at times for those participants living in remote areas. These issues highlight some of the concerns with the integration of new technology and subsequent services into clinical practice; in this instance incompatibility with the person's lifestyle/schedule as they were required to be home, and the need for users to have sufficient connectivity to participate. Future telehealth technologies should consider utilising mobile technologies that provide the user with the option of being in a location that preferentially suits their lifestyle.

2.4.2.2 Facilitating home monitoring

Home monitoring consists of the person with CF performing clinical assessments in their own home, potentially facilitating the earlier detection of respiratory exacerbations by the CF team. As early as 2005, the feasibility of home monitoring solutions for use in people with CF were being investigated, such as the performance of home spirometry by the participant and the recording of respiratory symptoms using an online portal (144).

In 2009, Bella and colleagues published data relating to the implementation of a home monitoring programme for the detection of respiratory exacerbations involving the use of the OXYTEL M32 system (Vivisol, Monza, Italy), which has the ability to record spirometry, percutaneous oxygen saturation (SpO₂) and pulse rate (21). This was a non-randomised, matched control study conducted over a 4-year period. The control participants were recruited each time a participant was allocated to the intervention group and received usual care, and were

matched for age, sex, FEV₁, bacterial colonisation, oxygen dependency and other comorbidities. Participants in the intervention group were asked to provide twice weekly data on continuous overnight SpO₂ and pulse rate, and morning spirometry. If changes in clinical status were observed by the CF medical team, the participant was contacted for assessment of the need for antibiotic treatment. Only 17 (57%) (mean [SD] age 16 [6] years, ppFEV₁ 68 [22] %) of the 30 participants recruited to the intervention group had > 6 months of transmitted data and included in the analysis. The length of follow-up for the intervention and matched controls were 29 [13] and 30 [20] months, respectively.

Adherence to the intervention was suboptimal, being measured as 'average compliance' to overnight oximetry, which was 52.4% (21). This, alongside the 43% attrition from the intervention group (compared to 7% in control), is likely explained by the requirement for twice weekly reporting over a relatively long study duration (mean 29 months). The intervention group had higher rates (per 100 person months) of CF clinic attendance (incidence rate ratio [95% CI] 1.25 [1.03 to 1.52]) and treatment cycles (treatment completed at home) (1.46 [1.09 to 1.96]) when compared to the control group, however there was no clear effect on hospitalisation rates and FEV₁. This draws attention to the cost-benefit of telehealth interventions; in this case overall HCU increased in the intervention group (via the telehealth intervention and CF clinic attendance) yet no improvements in number of hospitalisations or FEV₁ were observed. A cost analysis relating to HCU was explored by this group in a subsequent publication, discussed later in this section.

Grzincich and colleagues went on to conduct a multicentre pilot RCT, examining the feasibility of, and participant satisfaction with, home monitoring over a 4-week period (145). Sixty adults with CF (mean [range] age 29 [19 to 44] years) from four Italian CF centres were enrolled in the study. A total of 30 were allocated to the intervention group which consisted of using a digital recording device to record measurements of spirometry, SpO₂ and respiratory symptoms. The frequency of measurement was adjusted according to disease severity; participants with mild disease (ppFEV₁ > 77%, n = 22) were asked to complete only spirometry every day for the first week and then twice per week for the remaining weeks. Participants with moderate disease (n =

32) were asked to complete the same spirometry regimen with the addition of measuring daytime SpO₂ bi-daily every day, and overnight once per week during the first week, reducing to bi-daily twice per week and overnight once per week for the remaining weeks. Those with severe disease (n = 6), completed the same regimen as the moderate group except with overnight SpO₂ twice per week instead of once. The control group received usual care, consisting of attending regular CF clinics including assessment of spirometry and oximetry.

Fifteen (50%) of the participants allocated to the intervention group did not complete the study due to protocol 'failures' (145). A failure was defined as either refusing to continue monitoring, completing < four spirometry measurements in the first week, not sending acceptable SpO₂ data in the first week, or sending no acceptable data in any of the second to fourth weeks. The most common participant feedback regarding how to improve the system included reducing the frequency of data transfer (n = 7, 23%) and improving certain aspects of the equipment (n = 8, 27%). Despite the high rate of attrition, 80% of participants in the intervention group expressed a desire to continue using the system in some form in future. Given the high rate of attrition, future studies would need to consider minimising the monitoring burden to improve adherence, by reducing the frequency of reporting and/or utilising more mobile technologies to gather data.

Bella and colleagues continued to explore the role of telehealth in CF, publishing data in the subsequent years relating to their ongoing telehealth program, including FEV₁ outcomes and basic cost analysis relating to HCU. They reported data on the change in FEV₁ over time between participants utilising telehealth (twice weekly spirometry and overnight oximetry) over a 4.5 year period (n = 16) and a group of controls (n = 16) (matched for age, sex, FEV₁, bacterial colonisation, oxygen dependency and other co-morbidities) (24). Over the study period, they found a mean (SD) improvement in relative FEV₁ in the telehealth group of 4.0 (13.0)% compared with a reduction in the control group of -14.3 (14.8)% (p < 0.01). On face value these results are impressive, however the sample size was small, and the use of retrospectively matched controls potentially introduced selection bias. Further, a t-test was used rather than a regression analysis, which may be less accurate as baseline covariates could not be
included in the model. In this study, the yearly average adherence to twice weekly reporting ranged from 23% to 42% throughout the 4.5 year study period.

The authors also examined the cost benefit of telehealth over a retrospective 2-year period in 19 participants using the same telehealth system (146). While they reported a total cost saving of \in 132,144,91, or \in 3,303.62 per participant, these figures were based on the presumed prevented HCU as a result of the early detection of respiratory exacerbations (i.e. estimating the resultant HCU should the exacerbation not have been detected). These results need to be interpreted with caution, as this method is far less rigorous than obtaining cost benefit data using a prospective, RCT research design.

Similar to the work undertaken in Italy, Sarfaraz and colleagues examined the feasibility of once daily spirometry and symptom reporting for the detection of respiratory exacerbations in 51 children and adults with CF age ≥ 12 years, at two centres in the UK (25). All participants were asked to perform spirometry and grade their respiratory symptoms (i.e. cough, sputum, breathlessness, fatigue) from 1 to 4 depending on severity, every evening on a personal digital assistant for the 6-month study period. Using a somewhat pragmatic approach for defining a respiratory exacerbation (i.e. three consecutive days of a 1-point increase in 3 symptoms and/or a decline in FEV₁ > 10%), the CF team would contact the participant by phone if their reported data met these criteria, and invite them to attend a CF clinic for further assessment. If a participant failed to report data for two consecutive days, they were contacted by the CF team to remind them to continue. Respiratory exacerbation data were compared with the same participants' data from the same 6-month period (in regard to months of the year) in the year preceding the study, as well as the 6 months directly following the study.

This study experienced high attrition (n = 32 [63%]); 14 participants (27%) withdrew (lack of data in run-in period = 8 [16%], respiratory exacerbation during run-in period on two attempts = 3 [6%], lung transplantation = 1 [2%], liver transplantation = 1, relocated = 1) and 18 (35%) did not record sufficient data to be deemed eligible (defined as > 50% completion) for inclusion in the analyses (25). Nineteen participants (mean [standard error] age 26 [1] years) completed the

study, with 53 respiratory exacerbations (mean [range] 3 [0 to 7] per participant) detected, of which 27 (51%) were treated with oral antibiotics and 26 (49%) were treated with IV antibiotics. The only clear difference in prescription of antibiotics in either of the comparison periods was for oral antibiotics, where fewer oral antibiotics were administered in the 6 months following the study (14 total courses vs. 27 total courses in the study period, p = 0.02). Further examination of the raw data reveals that the prescription of 27 courses of oral antibiotics in the study period was also higher than that observed in the 6-month period in the previous year (total courses = 19). Therefore, this result likely reflects an increased rate of detection of respiratory exacerbations and subsequent provision of oral antibiotics as a result of the intervention.

While the authors stated that the poor adherence to the intervention was contrary to their expectations, this does not seem completely surprising given the daily reporting burden experienced by the participants (25). Similar to other work discussed earlier in this section, participants were also required to be at home to use the telehealth system, and therefore it did not appear compatible with their lifestyle/schedule. While not reported by the authors, it may have been beneficial to compare the adherence with reporting for the 19 participants who completed the study in the first and last 3 months of the study period, to ascertain potential reporting fatigue as the intervention progressed.

The most recent and methodologically robust study aiming to improve the detection of respiratory exacerbations through home monitoring was the Early Intervention in Cystic Fibrosis Exacerbation (eICE) study conducted by Lechtzin and colleagues in the US (96). This multicentre RCT in 267 adolescents (age \geq 14) and adults with CF (mean [SD] age 27 [12] years) examined the impact of home spirometry and symptom reporting via the CFRSD using a Viasys AM2 device (CareFusion, Yorba Linda, California, US) on outcomes measured over 12 months. Outcomes included change in FEV₁, time to and rates of respiratory exacerbation, other measures of HCU, HRQoL, change in respiratory symptoms, and adherence to the study protocol. The intervention group were asked to complete the home monitoring twice weekly, with the system alerting the CF team if a respiratory exacerbation was detected, defined as either a relative decrease in FEV₁ by > 10% or worsened CFRSD symptoms from baseline in

two or more categories. The control group were able to contact the CF clinic triage telephone line at any time to report acute changes in clinical status, and both groups were asked to attend 3-monthly CF clinics (total of 5 clinics, including at baseline, in the 12-month study period).

In this study, there was no clear effect of the intervention on the magnitude of decline of FEV₁ over the 12-month study period (mean difference [95% CI] 0.0 [– 0.1 to 0.1]L) (96). The intervention facilitated the detection of respiratory exacerbations sooner when compared to control (HR [95 % CI] 1.45 [1.1 to 1.9]). Participants in the intervention group also had a higher number of acute CF clinic attendances (153 vs. 64, defined as resulting from home monitoring or calls to the triage line) and a higher proportion of oral antibiotics were prescribed at acute CF clinic attendances for protocol defined respiratory exacerbations (72/108 [67%] vs. 19/44 [43%], p = 0.01.). To reduce the confounding effect of the increased contact with the CF team via the home monitoring system, the authors chose to measure the proportion of events for each HCU outcome during acute CF clinics, and only when the protocol defined exacerbation criteria were met. The increased prescription of oral antibiotics in the intervention group may have reduced the need for IV antibiotics and hospitalisation, as the proportions of participants in the intervention group requiring these were fewer than that seen in the control group (IV antibiotics, 35 [32%] vs. 23 [52%], p = 0.03; hospitalisation, 31 [29%] vs. 22 [50%], p = 0.01).

Although this study demonstrated that home monitoring resulted in the earlier detection of respiratory exacerbations and increased provision of oral antibiotic treatment amongst participants in the intervention group, this did not translate to a reduced magnitude of decline of FEV₁ (96). Participants in the intervention group did report less symptoms (measured by the CFRSD) over the study period (mean difference [95% CI] -4.1 [-7.8 to -0.5], p = 0.03) compared to those in the control group, which in this instance may have been due to receiving an increased number of courses of oral antibiotics. Consistent with earlier work, the level of attrition in this study was greater in the intervention group when compared to the control group (n = 33 [24%] vs. n = 21 [16%]). Of these, 20 (15%) participants in the intervention group withdrew due to 'participant decision', as opposed to 11 (8%) in the control group. This is not surprising given the high reporting burden associated with twice weekly spirometry and

symptom reporting which needed to be completed at home. Participants in the intervention group corroborated this with higher scores on self-reported protocol burden measured on a scale of 0 to 10 compared to the control group (mean 2.9 vs. 0.6, p < 0.001). Strengthening the argument that the reporting burden was too great, adherence to the home monitoring protocol was low, with 19% of participants reporting data twice weekly \geq 80% of the time, and 50% reporting data once weekly \geq 80% of the time (96).

Thus, previous research investigating home monitoring interventions aimed at detecting respiratory exacerbations more rapidly and facilitating the earlier provision of treatment share a common theme of high reporting burden when using devices that are not mobile (i.e. desktop computers or large, bulky medical devices). This subsequently required the participants to be present at home in order to complete the reporting, which is not conducive to the active and busy lifestyles lived by many adults with CF. On occasion, monitoring was also required continuously overnight (e.g. with oximetry) or required a large amount of physical effort (e.g. spirometry). Therefore, the low level of adherence to the interventions described reflects the inability of the reporting system to be integrated into the user's lifestyle/schedule, and this should be the focus of future interventions in this area. It would be advisable for home monitoring studies to incorporate small, mobile technology already owned by the participants. Smartphones are an example of a technology that meets this description with the capabilities of reporting medical information to the CF team.

2.4.2.3 Promotion of exercise, exercise testing, and physical activity

The role of telehealth in improving engagement with physical activity and exercise amongst people with CF has been explored in a small number of studies. In 2013, Cox and colleagues published data examining the feasibility of assessing exercise capacity (using the 3-minute step test [3MST]) via telehealth in the form of videoconferencing (27). Ten adults with CF (mean [SD] age 32 [7] years) completed the 3MST twice within a 24-hour period, once via telehealth and once in-person with a clinician. Both sessions were conducted in a controlled environment at the hospital (i.e. telehealth was conducted with the participant and clinician in separate rooms). There were no differences in physiological responses (pulse rate and SpO₂) between the

two tests, and participant satisfaction was high with similar ratings on a 5-point Likert scale for 'ease of clinician interaction' (mean [SD] 4.8 [0.4] vs. 5.0 [0.0]), p = 0.34) and 'participant comfort' (4.7 [0.9] vs. 4.8 [0.4], p = 0.68) observed with telehealth and in-person, respectively. Only one (10%) participant preferred in-person supervision of the assessment (as opposed to 'no preference') and the mean (95% CI) usability of the telehealth system, using the System Usability Scale (SUS), was rated high at 86 (80 to 91)%. The only negative observation relating to the use of the videoconferencing was the 'ease of hearing the metronome', which was reduced during the telehealth assessment compared to the in-person session (3.3 [0.4] vs. 4.8 [0.1], p < 0.01). This study, the first to use telehealth for the assessment of exercise capacity in this population, provided encouragement to researchers and clinicians in regard to repeating this model of assessment with participants' in their own homes.

Two studies have since evaluated the feasibility of using telehealth to promote exercise in people with CF within their own home. Chen and colleagues in the US implemented a 6-week program for 10 children with CF consisting of three 30-min interactive group sessions per week of aerobic, resistance, and flexibility exercises (148). A telehealth platform was used to provide videoconferencing and monitoring of heart rate via a Polar monitor, and sessions were supervised by a clinician. The participants completed 85% of the sessions and usability measured via the SUS was high at 91%. In nine adults (mean [SD] age 31 [9] years), Tomlinson and colleagues investigated the feasibility of offering an 8-week supervised, personalised exercise training intervention conducted via videoconferencing (Skype) for adults with CF (28). Participants completed 68% of the sessions, with two participants (22%) withdrawing due to time pressures. Satisfaction with telehealth was high (90%), despite technical issues (video or audio delay) being reported during 25% of the videoconferencing calls. These two studies, alongside the work undertaken by Cox and colleagues, demonstrate high usability and satisfaction with the use of telehealth in the assessment and performance of exercise, as well as highlighting the need for technology that provides a certain level of video and audio quality within the system.

Finally, Cox and colleagues designed an internet-based program for monitoring and promoting physical activity participation for adults with CF and examined its feasibility and acceptability amongst users (29). Ten adults (mean [SD] age 30 [8]) years used the program (ActivOnline) for an 8-week period, with fortnightly phone support from a physiotherapist designed to facilitate goal setting and improve motivation. There was a good level of feasibility reported, with participants logging in mean [SD] 13 (11) times during the study period and recording a mean (range) of 35 (15 to 57) sessions of physical activity (approximately 4 per week per participant). Acceptability was high, with a median (IQR) SUS score of 89 (84 to 95). When measured on a 5-point Likert scale, 'perceived benefit' (median [IQR] 4.0 [3.0 to 4.8]) and 'desire to use the program' (3.5 [3.0 to 4.8]) were also good. This intervention was well designed to target the demographic features of the participants, and the authors acknowledged that a more mobile-friendly version of the program (i.e. an app) may improve acceptability further.

2.4.2.4 Promotion of adherence to recommended treatment

A 6-week, multicentre study conducted in the UK investigated the acceptability of an adaptive aerosol delivery (AAD) nebuliser device used to monitor adherence and facilitate the reordering of nebulised medications from the pharmacy (148). The AAD device contained a 'disc' which stored information regarding the required medication dose, details of the completed treatment sessions (medication, date, time, duration, delivered dose), medication expiry date and the number of doses remaining, allowing for the timely dispensing of more medication. Nineteen participants (6 children, 13 adults) with CF completed the study, with 1,180 (80%) of 1,473 expected treatment sessions completed in full. Regarding the telehealth aspect of the study (i.e. remote ordering of more medication from the pharmacy and delivery to home), participants were divided on its benefit, with 44% viewing it as 'great', and 44% viewing it as 'inconvenient', most likely explained by the requirement to be at home at specific times in order to receive the medication at the time of scheduled delivery.

One further study investigated the use of text messaging to improve adherence to nebulised medication in 17 children and adolescents with CF (mean [range] age 12 [5 to 16] years) (149).

The authors developed an automated system that sent text message reminders up to twice daily to the participant and/or their parent(s)/carer for a period of 6 months. Adherence to the nebulised medications, measured with an AAD nebuliser device as in the previous study, was compared between the 6-month study period and the 6-month period preceding the study. Mean (SD) adherence during the study period was 80 (29) %, which was similar to the preceding 6-month period (81 [25] %). The lack of improvement in adherence was possibly due to the already high level of adherence amongst the participants in the preceding 6-month period (i.e. a ceiling effect), and this intervention may have more impact in a group with lower levels of baseline nebuliser medication adherence.

2.5 Conclusion

Cystic fibrosis is a common genetic, multi-system disease affecting the Caucasian population, with an incidence of 1 in 3,000 live births in Australia (31). Advancements in treatment options and models of care have seen the median life expectancy increase in recent decades to 44 years in Australia (31), with respiratory failure secondary to bronchiectasis remaining the most common cause of death (45). The estimated cost of providing care for a person with CF is variable, and it is now expected to increase dramatically with new CFTR modular therapies that cost approximately \$270,000 AUD per person annually (42).

It is recommended that people with CF have their care managed in centralised CF centres, by an experienced multidisciplinary team (3, 4), due to known improvements in health outcomes such as FEV₁ and nutritional status (5, 6). This creates barriers to care in itself, given that many people with CF live large distances away from a CF centre, especially in areas of low population density such as in WA. Further, people with CF have competing lifestyle commitments and experience high treatment burden (71), limiting the time available for engagement with the CF team. Strategies aimed at reducing the barriers to accessing care are required.

With the challenges faced regarding access to care for people CF, the diagnosis of respiratory exacerbations becomes even more challenging. What constitutes a respiratory exacerbation in CF is a contentious issue, and there is no universally accepted definition (82, 83, 86, 87, 94, 95). A pragmatic approach is often taken within clinical practice and research. Similarly, the treatment provided for respiratory exacerbations (with oral, inhaled and IV antibiotics) varies between physicians and CF centres (84, 86), due to the lack of evidence supporting a specific treatment approach. Regardless, respiratory exacerbations increase HCU (41, 91) and have a profound and detrimental long-lasting impact on FEV_1 (17, 19, 54, 89), HRQoL (13, 109), physical activity (15), nutritional status (113) and ultimately survival (17).

The delayed identification of respiratory exacerbations and initiation of treatment has been associated with failure of FEV_1 to return to pre-exacerbation levels (19). Delayed identification

is often due to delayed presentation to a CF clinic by a person when their respiratory symptoms have worsened; with the reasons for this delay being multifactorial (20). Traditionally, people with CF are required to attend routine (at least 3-monthly) CF clinics to be assessed by the CF team (4), or they can contact the centre sooner if required, usually by phone. Given the busy lifestyle of people with CF, and the high treatment burden experienced, solutions are required to facilitate the faster identification of respiratory exacerbations and provision of antibiotic treatment.

The impact of telehealth technology in CF has been evaluated in interventions aimed at providing CF clinics (26, 143), identifying respiratory exacerbations via home monitoring (21, 24, 25, 96, 144, 150), improving adherence (148, 149), and promoting exercise, exercise testing and physical activity (27-29). Although adults with CF are likely to be comfortable with the use of mobile technologies, the effectiveness of telehealth at detecting respiratory exacerbations has previously been compromised by suboptimal adherence (i.e. 10 to 67%) (21, 22, 24, 25, 96). This is likely due to the high reporting burden experienced by participants, such as daily reporting of symptoms and/or regular spirometry. Minimising this burden in order to optimise adherence is an important consideration when designing telehealth technology.

This programme of research aimed to overcome some of these shortcomings by investigating the impact of the integration of modern technology with traditional CF care on health outcomes. First, videoconferencing for the provision of CF clinics was implemented for adults with CF living in rural and remote WA. Second, an app was developed for use by adults with CF to facilitate the early identification of symptoms suggestive of a respiratory exacerbation and allow the CF team to initiate treatment sooner.

CHAPTER 3

STUDY 1

This chapter details the methodology, results and discussion for Study 1 (described in Section 1.1) of this programme of research. Specifically, Study 1 investigated whether there was a difference in the magnitude of decline of forced expiratory volume in 1 second (FEV₁) between adults with cystic fibrosis (CF) living in the Perth metropolitan area and those living in rural and remote Western Australia (WA), over a 3-year period.

Chapter 3: Study 1

3.1 Study design

This study was an audit conducted retrospectively that investigated whether adults with CF living in metropolitan Perth, and those in rural and remote WA, have a different magnitude of decline in spirometry. While unique ethical approval was not sought for this study, all participants had previously consented to these data being used for research purposes via the Australia CF Data Registry. Approval was obtained from the Australia CF Data Registry prior to the extraction of data.

3.1.1 Participants

Adults with CF aged 18 years or more were included if they had i) attended the Sir Charles Gairdner Hospital (SCGH) CF Centre (both metropolitan and rural and remote) for the entire period from January 2010 to December 2012; and ii) had performed spirometry on at least two occasions during this time period (as this was the minimum required to conduct a regression analysis). A 3-year period was chosen to ensure the majority of participants attending the CF centre had two or more data points. There were no exclusion criteria.

3.1.2 Procedure and study measurements

All measures of spirometry (FEV₁, percentage predicted FEV₁ [ppFEV₁], forced vital capacity [FVC] and percentage predicted FVC [ppFVC]) for adults with CF attending the SCGH CF Centre were extracted retrospectively if they were originally collected between January 2010 and December 2012 inclusive. Spirometry was obtained at outpatient CF clinics, attended by each participant according to clinical need. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted ppFEV₁ and ppFVC. Data were extracted from the Australian CF Data Registry and verified with the participant's medical record and Medgraphics Breeze software program (MGC Diagnostics, Saint Paul, MN USA). In addition to spirometry, data were extracted on demographic information including age, sex, body mass index (BMI), pregnancy, *Pseudomonas aeruginosa* colonisation and CF-related diabetes (CFRD). All data points between these dates were extracted on 06/02/2014. The principal investigator collated all data pertaining to the measurements collected as part of this study.

Spirometry had been performed according to American Thoracic Society guidelines for repeatability and acceptability (152), and pre- and post-bronchodilator results were used where available. Health professionals who performed spirometry had previously received training from a certified respiratory technician, and the spirometer was calibrated at the commencement of each CF clinic. All spirometry was reviewed by respiratory scientist following completion. The magnitude of decline in spirometry over these 3 years was compared between adults with CF living in metropolitan Perth vs. those living in rural and remote WA (defined as living greater than 100km from the geographical centre of Perth).

3.1.3 Analyses

Statistical analyses were conducted by a statistician using Stata® (StataCorp. 2015. Stata Statistical Software: Release 14. College Station, Texas, USA). Data are expressed as mean and standard deviation (SD) unless otherwise stated. Magnitude of decline (i.e. slope of line for variables plotted over time) in FEV₁, ppFEV₁, FVC and ppFVC was determined using mixed effects models. Adjusted mixed effects models were implemented to obtain FEV₁ and FVC outcomes by location and months since baseline visit, after adjusting for age, sex, BMI, pregnancy, *Pseudomonas aeruginosa* colonisation and CFRD. Further adjusted mixed effects models analysis was completed stratifying by age group (18 to 29 years, \geq 30 years) (as age has been shown to impact the magnitude of decline in spirometry). Locally Weighted Scatterplot Smoothing (LOWESS) was used to present figures of ppFEV₁ and ppFVC data. Alpha was set at 0.05.

3.2 **Results**

A total of 156 participants met the study criteria and contributed data to the analyses (Table 3.1). In total, 2,748 ((mean [range] 219 (8 [2 to 24]) and (21 [3 to 58]))data points (spirometry) were available for participants living in Perth metropolitan (n = 130) and rural and remote (n = 26) areas, respectively.

3.2.1 Magnitude of decline of spirometry

There was no clear difference between participants living in the Perth metropolitan and rural and remote areas for the magnitude of decline in FEV₁ (mean difference [95% CI] 0.00L [-0.01 to 0.01], p = 0.16), ppFEV₁ (0.07% [-0.07 to 0.21], p = 0.31), FVC (0.00L [-0.01 to 0.01], p = 0.31) or ppFVC (0.05% [-0.10 to 0.20], p = 0.53) (Figure 3.1). Results were unchanged when stratified for age (18 to 29 years or ≥ 30 years).

	Perth metropolitan	Rural and remote WA
Female, n (%)	57 (44)	16 (62)
Age, yr	27 (8)	30 (11)
BMI, kg/m ²	22.9 (3.7)	24.3 (6.5)
FEV ₁ , L	2.42 (0.97)	2.46 (1.00)
$ppFEV_1$	62 (20)	67 (19)
FVC, L	3.55 (1.19)	3.53 (1.19)
ppFVC	76 (18)	81 (17)
Pseudomonas aeruginosa, n (%)	120 (92)	23 (88)
CF-related diabetes, n (%)	19 (16)	1 (4)
Pregnancy during audit period, n (%)	3 (2)	0 (0)

Table 3.1 Participant characteristics at the time of their first occasion of spirometry for those living in the Perth metropolitan area (n = 130) and rural and remote Western Australia (n = 26)

Data are presented as mean (standard deviation) unless otherwise stated. BMI = body mass index, FEV_1 = forced expiratory volume in 1 second, FVC = forced vital capacity, pp = % predicted. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted ppFEV₁ and ppFVC.



Figure 3.1 Locally Weighted Scatterplot Smoothing for percentage predicted forced expiratory volume in 1 second (a) and percentage predicted forced vital capacity (b) for participants living in metropolitan and rural and remote areas of Western Australia

 $ppFEV_1 = percentage predicted forced expiratory volume in 1 second, ppFVC = percentage predicted forced vital capacity. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted ppFEV₁ and ppFVC. All data points for each participant are provided, with timepoint '0' on the x axis indicating the first occasion of spirometry, with subsequent data points in months from baseline according to each participant's CF clinic appointments.$

Chapter 3: Study 1

3.3 Discussion

This study aimed to determine whether there was a difference in the magnitude of decline in spirometry between adults with CF living in the Perth metropolitan and rural and remote areas in WA. The study design was pragmatic given the historical nature of the data, and it was beyond the scope of this research to undertake data collection prospectively. No clear effect on the magnitude of decline of spirometry was demonstrated between groups. Stratifying the analysis for age did not change the results.

While unexpected, the most likely reason no difference was detected in spirometry relates to the limited sample size available for these analyses. That is, data were available for only 26 participants living in rural and remote WA and these participants averaged only eight spirometry recordings over the 3-year study period. Further, some of these participants contributed as few as two (n = 2) or three (n = 4) data points (spirometry), of which the spirometry values were highly variable over the three-year period. Taken together, these factors made it difficult to estimate the true magnitude of decline in measures of spirometry in the rural and remote group, with any precision. This, in turn, compromised the power to detect betweengroup differences. Using data extracted retrospectively from the CF data registry may give rise to missing data points, however the SCGH CF centre had a dedicated employee entering the data (including spirometry) close to the time of each CF clinic, reducing the chance of this occurring.

It is also possible that there was no difference because of the relatively short time period (3 years) and a longer period of surveillance (i.e. 5 to 10 years) might be needed to detect a difference in the magnitude of decline in spirometry. Conducting a prospective study will likely improve the standardisation of data collection and potentially reduce variability in the results that was observed in the present study. Notwithstanding these considerations, the data presented in the current study draw attention to the lower disease surveillance in those people with CF living in rural and remote WA. This may impact health outcomes in a similar way to that observed in earlier work (5, 6).

The fewer available measures of spirometry (performed at CF clinics) amongst adults living in rural and remote WA highlights the reduced access to care experienced by this group. Programs aimed at improving access and engagement with the CF centre and multidisciplinary team need to be considered. Telehealth, in the form of videoconferencing, is a potential solution to some of the barriers (geography, time, financial) faced by adults with CF living in these areas.

CHAPTER 4

STUDY 2 (PART A, PART B)

This chapter details the methodology, results and discussion for Study 2 of this programme of research. Study 2 consists of two parts; Part A (described in Section 1.1) investigates the impact of providing cystic fibrosis (CF) clinics via videoconferencing on healthcare utilisation and other health outcomes in adults with CF living in rural and remote Western Australia (WA); and Part B (described in Section 1.2) is a sub-analysis of data from Part A, relating to absenteeism and presenteeism data collected using the World Health Organization (WHO) Health and Work Performance Questionnaire (HPQ). Part A has been published as original research in *Journal of Telemedicine and Telecare* in July 2016 (Appendix 2), and Part B as a letter to the editor in *Journal of Cystic Fibrosis* in July 2016 (Appendix 3). Data were collected for Study 2 from July 2013 to June 2015.

Chapter 4: Study 2

4.1 Part A Design

This was a single group study in which adults with CF who lived in rural and remote WA were invited to utilise telehealth clinics as part of routine care (i.e. attending a minimum of four clinics per year) over a 12-month period. Participants were recruited between July 2013 and August 2014. Approval was obtained from the Sir Charles Gairdner Osborne Park Healthcare Group (2012-209) and Curtin University (83/2013) Human Research Ethics Committees.

4.1.1 Participants

Adults with CF aged 18 years or more and living a minimum of 100km from the SCGH CF Centre were eligible to participate in this study. Exclusion criteria were pregnancy, previous lung transplantation or current listing for lung transplantation, as adults with CF meeting these criteria were already required to travel to Perth more frequently for medical care.

4.1.2 Telehealth clinics

This study aimed to provide telehealth clinics for participants approximately every three months, or sooner if clinically indicated, for reasons including self-reported acute deterioration in respiratory status or early follow-up after a course of intravenous (IV) antibiotics. Telehealth clinics were scheduled to suit the participants, CF team, regional telehealth coordinators, and a health professional local to the participant (physiotherapist, nurse or general practitioner).

On the day of the telehealth clinic, participants attended the nearest regional hospital. The participant's height, weight and spirometry were measured by a designated health professional before the telehealth clinic commenced. Telehealth clinics were held via a videoconferencing using Polycom HDX® series (Polycom, San Jose, USA) equipment at both SCGH and hospitals throughout rural and remote WA. During telehealth clinics, the participant chose whether to be reviewed by each member of the multidisciplinary CF team individually (i.e. consultant physician, nurse practitioner, physiotherapist and dietician) or by the whole team together in a group. A pharmacist and social worker were also available to provide input via telephone at a later date if requested by the participant

or the team. Following the telehealth clinic, prescriptions were faxed or medications were couriered to the participant's local pharmacy.

Traditional 'in-person' clinics at SCGH were still provided if requested by the participant (i.e. as a preference or if more convenient at that point in time), or if clinically indicated as previously described. Participants were encouraged to attend one in-person clinic during the intervention to allow for more comprehensive annual review of their clinical status. If a participant required admission to hospital at any time during the study, this was arranged at the SCGH CF Centre in Perth. Courses of home IV antibiotics were also commenced at the CF centre before the participant returned home. This allowed for intensive medical and allied health input to be provided, as well as access to specialist secondary services such as intravascular access, endocrinology, gastroenterology and immunology.

4.1.3 Study measurements

Participants completed the relevant measurements at baseline and at each CF clinic (telehealth or inperson) until the completion of the study period, unless otherwise stated. The principal investigator performed all assessments (including questionnaires) during CF clinics conducted at SCGH. For telehealth assessments, spirometry and weight were assessed by a health professional local to the participant, with the remaining measures collected by the principal investigator, unless otherwise stated.

4.1.4 **Primary outcome**

4.1.4.1 Uptake of telehealth

Uptake of telehealth was measured via the number of telehealth CF clinics attended throughout the study period. Clinic attendance data, recorded prospectively in the participant's medical record, were verified at the end of the 12-month study period and compared with the 12-month period preceding recruitment to the study.

Chapter 4: Study 2

4.1.5 Secondary outcomes

4.1.5.1 Participant satisfaction

Participant satisfaction with telehealth clinics was measured at the end of the study period using two questionnaires; the Telehealth Satisfaction Scale (TeSS) (153) and a purpose developed satisfaction survey. The TeSS is a 10-item questionnaire originally developed to assess patient satisfaction with telehealth in a rural memory clinic. For the purposes of this study, the term 'memory clinic team' was replaced with 'CF team' and the final statement requiring a response "How well the staff answered your questions about the equipment", was omitted as the participants in this study were not required to operate the telehealth equipment. The second survey contained six questions that were answered using a 5-point Likert scale (responses ranging from strongly agree to strongly disagree, with higher scores indicating stronger agreement), allowing the participant to provide further feedback on the telehealth clinics including preference for either telehealth or traditional in-person clinics. This survey was piloted in adults with CF (n = 5) living in the metropolitan area and minor changes were made to optimise its readability. The final version of this purpose-built survey and the modified TeSS are available as Appendix 4. The surveys were mailed to participants, and were returned by mail in pre-addressed, postage paid envelopes. The content of the surveys did not allow the researchers to identify the participants.

4.1.5.2 Healthcare utilisation

Healthcare utilisation data (HCU) (total number of CF clinics attended, courses and days of IV antibiotics, courses of inhaled and oral antibiotics, hospital admissions and admission days) were extracted from each participant's medical record at the end of the study period and compared with the 12-month period preceding recruitment to the study.

4.1.5.3 Spirometry

Spirometry (forced expiratory volume in 1 second [FEV₁] and forced vital capacity [FVC]) was measured according to accepted guidelines (152) using either CPFS/D USB[™] (MGC Diagnostics Corporation, St Paul, USA) or Easyone[™] (NDD Medical Technologies, Andover, USA) spirometers. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted FEV₁ (ppFEV₁) and percentage predicted FVC (ppFVC). Health professionals who performed spirometry had previously received training from a certified respiratory technician.

4.1.5.4 Nutritional status

Height (Seca, Hamburg, Germany) and weight (A&D Company, Tokyo, Japan) were measured and used to calculate body mass index (BMI).

4.1.5.5 Health-related quality of life

Health-related quality of life (HRQoL) was measured using the Cystic Fibrosis Questionnaire – Revised (CFQ-R) (107) (discussed in Section 2.3.2.2) This self-completed questionnaire comprises 50 questions answered on a 4-point Likert scale, and covers 12 domains comprising physical, vitality, emotion, treatment burden, health, social, body, role, weight, respiratory, digestion and eating. The CFQ-R requires the user to recall information from the previous 2 weeks.

4.1.6 Analyses

Statistical analyses were conducted using Stata® (StataCorp. 2015. Stata Statistical Software: Release 14. College Station, TX: StataCorp LP). The distribution of data was checked for normality. Satisfaction data were reported using descriptive statistics. Healthcare utilisation data were compared pre- and during- the 12-month study period with Poisson or negative binomial regression. Paired t-tests and Wilcoxon signed rank tests were used to compare spirometry, weight, BMI, and HRQoL data pre- and during the 12-month study period as appropriate. As this study took the pragmatic approach of maximising the inclusion of all adults with CF living in rural and remote WA, with the primary aim of reporting the uptake of telehealth clinics amongst these people, no sample size calculations were undertaken. Alpha was set at 0.05. All data from participants who completed the study period were included in the analysis irrespective of their uptake of telehealth.

Chapter 4: Study 2

4.2 Part A Results

Twenty-three out of a possible 28 (82%) adults with CF living in rural and remote WA were recruited for this study (Table 4.1). Reasons for not participating were regular travel to Perth for other medical reasons (n = 3) and intention to move interstate or overseas within the study period (n = 2). The distance participants were required to travel to their nearest regional hospital was a median (range) of 5 (1 to 22) km (Table 4.1). Data from two participants were excluded from the analysis as they moved residence during the study period (one to Perth and one internationally). One other participant chose to travel to Perth for all CF clinics and did not utilise the telehealth intervention, therefore they were not asked to complete the satisfaction surveys.

4.2.1 Uptake of the telehealth clinics

The 21 participants included in the final analyses attended a total of 100 clinics during the study period, of which 66 (66%) were provided via telehealth (Table 4.2). Four participants (19%) utilised telehealth for all of their clinics, however three of these people also had hospital admissions at SCGH during the study period, and subsequent in-person contact with CF team at this time. The median (range) number of clinics per participant during the study period increased from 2 (0 to 6) in the preceding 12 months to 5 (2 to 8) (p < 0.001, Table 4.2). A total of 19 of the 21 participants (90%) attended four or more CF clinics (telehealth or in-person) during the study period, compared with 4 (19%) in the previous year. On two occasions, participants did not attend their scheduled telehealth clinic and failed to inform the CF team. Of the 34 in-person clinics where the participant attended the SCGH CF Centre, 19 (56%) were requested by the CF team for clinical indications.

4.2.2 Participant satisfaction

The TeSS and purpose developed participant satisfaction survey data were available for 17 (81%) participants. The participants' rated all nine items on the TeSS as either 'good' or 'excellent', with the exception of one participant's response to the item relating to the voice quality of the equipment (Table 4.3).

Participants, n	23
Female, n (%)	14 (61)
Age, yr	31.4 (10.2)
BMI, kg/m ²	21.1 (18.7 to 58.0) ^a
FEV ₁ , L	2.08 (0.78)
ppFEV ₁ , %	60 (20)
Distance from specialist CF centre, km	417 (100 to 2,567) ^a
Distance from nearest regional hospital, km	5 (1 to 22) ^a
Pseudomonas aeruginosa, n (%)	21 (91)
CF-related diabetes, n (%)	4 (17)
Pancreatic insufficiency, n (%)	21 (91)

Table 4.1 Participant (n = 23) characteristics at commencement of the study

Data are presented as mean (SD) unless otherwise stated. BMI = body mass index, CF = cystic fibrosis, $FEV_1 =$ forced expiratory volume in 1 second, pp = percentage predicted. The Global Lung Function Initiative 2012 equations (151) were used to calculate ppFEV₁ and ppFVC. ^a data are presented as median (range).

	Pre-telehealth		During telehealth				
	Nº	per participant	Nº	per participant	IRR	Std. err.	р
CF clinics	46	2 (0 to 6)	100	5 (2 to 8)	2.2	0.4	< 0.01
IV antibiotic courses	14	0 (0 to 3)	25	1 (0 to 4)	1.8	0.3	0.08
IV antibiotic days	170	8 (0 to 32) ^b	350	17 (0 to 61) ^b	2.3	0.8	0.03 ^a
Hospital admissions	8	0 (0 to 2)	19	1 (0 to 3)	2.4	1.0	0.04
Admission days	56	3 (0 to 15) ^b	209	10 (0 to 36) ^b	3.7	0.4	0.01 ^a
Oral antibiotic courses	24	1 (0 to 3)	29	1 (0 to 3)	1.2	0.3	0.5
Inhaled antibiotic courses	18	1 (0 to 2)	31	2 (0 to 3)	1.7	0.5	0.07

Table 4.2 Healthcare utilisation data for participants (n = 21) who completed the intervention, comparing the 12-
month periods prior to and during telehealth

Data are presented as number (N°) and median (range). IRR = incidence rate ratio, IV = intravenous, Std. err. = standard error. Analyses performed using Poisson regression or ^a negative binomial regression.

^b data are presented as mean (range).

Overall, how satisfied were you with:	Score
The voice quality of the equipment	4 (1 to 4)
The visual quality of the equipment	4 (3 to 4)
Your personal comfort using the telehealth system	4 (3 to 4)
The ease of getting to the telehealth department	4 (3 to 4)
The length of time with the CF team	4 (3 to 4)
The explanation of your treatment by the CF team	4 (3 to 4)
The thoroughness, carefulness and skilfulness of the CF team	4 (3 to 4)
The courtesy, respect, sensitivity, and friendliness of the CF team	4 (3 to 4)
How well your privacy was respected	4 (3 to 4)

Table 4.3 Participant (n = 17) responses to questions from the Telehealth Satisfaction Scale

Data are presented as median (range). Questions were answered using a 4-point scale: Excellent = 4, Good = 3, Fair = 2, Poor = 1.

Results from the purpose developed satisfaction survey (Table 4.4) demonstrated that 16 (94%) participants strongly agreed or agreed that telehealth clinics were a good way to manage CF care. Ten (59%) participants preferred telehealth clinics for their CF clinics.

4.2.3 Healthcare utilisation

During the study period, there was an increase in the number of IV antibiotics days, hospital admissions and admission days per participant (Table 4.2) (all p < 0.05). The number of courses of IV antibiotics, oral antibiotics, and inhaled antibiotics per participant also increased, however these differences were not clear (Table 4.2). Of the 25 respiratory exacerbations requiring IV antibiotics during the study period, 20 (80%) were detected during telehealth clinics.

4.2.4 Spirometry

No changes were seen in spirometry (FEV₁, ppFEV₁, FVC and ppFVC) following the intervention (Table 4.5).

4.2.5 Nutritional status

In the 20 participants for whom weight gain was a goal, there was an increase in median (interquartile range) BMI from 20.9 (4.7) to 21.7 (3.4) at the end of the study period, however this was not significant (p = 0.12). One participant was excluded from the BMI analysis as they were morbidly obese and weight loss was being encouraged. Fourteen (70%) participants gained weight during the intervention, with a mean (standard deviation) increase of 2.2 (2.1) kg, or 3.7 (3.6) %.

4.2.6 Health-related quality of life

There was an improvement in the vitality domain (p = 0.04) of the CFQ-R, and a trend toward improvement in the weight domain (p = 0.06) (Table 4.6). No clear changes were observed in other domains of the CFQ-R.

	SA	А	U	D	SD
The organisation of the clinics met my expectations	5 (29)	11 (65)	1 (6)	0 (0)	0 (0)
The timing of the telehealth clinic appointments were convenient for me	8 (47)	9 (53)	0 (0)	0 (0)	0 (0)
The healthcare I received during the telehealth clinics met my expectations	6 (35)	10 (59)	1 (6)	0 (0)	0 (0)
I was able to communicate effectively with the CF team using this technology	6 (35)	10 (59)	1 (6)	0 (0)	0 (0)
I prefer telehealth clinics over traditional 'face to face' clinics	3 (18)	7 (41)	5 (29)	2 (12)	0 (0)
Telehealth clinics are a good way to manage my CF care	3 (18)	13 (76)	1 (6)	0 (0)	0 (0)

Table 4.4 Participant (n = 17) responses to questions from the purpose developed satisfaction survey, with data presented as number (percentage) of participants

Data are presented as number (percentage). Questions were answered using a 5-point Likert scale: SA = Strongly Agree, A = Agree, U = Unsure, D = Disagree, SD = Strongly Disagree.

	Prior to telehealth	Following telehealth	Difference	95% CI
FEV ₁ (L)	2.02 (0.76)	1.96 (0.69)	-0.06	-0.19 to 0.08
$ppFEV_1$ (%)	58 (19)	56 (18)	-2	-4.91 to 1.87
FVC (L)	3.18 (0.90)	3.15 (-0.03)	-0.03	-0.17 to 0.11
ppFVC (%)	75 (17)	74 (15)	-1	-3.44 to 2.78

Table 4.5 Spirometry measured prior to telehealth and at the end of the 12-month telehealth study period for participants (n = 21) who completed the study

Data are presented as mean (standard deviation). $CI = confidence interval, FEV_1 = forced expiratory volume in 1 second, FVC = forced vital capacity, pp = percentage predicted. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted ppFEV₁ and ppFVC.$

CFQ-R domain	Prior to telehealth	Following telehealth	Mean diff. (95% CI)	р
Physical functioning ^a	71 (33 to 42)	75 (42 to 90)	-	0.64 ^b
Vitality	44 (21)	55 (20)	11 (1 to 22)	0.04
Emotional functioning	69 (23)	71 (19)	2 (-9 to 12)	0.75
Eating ^a	100 (61 to 100)	100 (67 to 100)	-	0.43 ^b
Treatment burden	58 (17)	56 (20)	-2 (-9 to 4)	0.51
Health perceptions	53 (23)	58 (19)	5 (-7 to 17)	0.39
Social	63 (20)	61 (18)	-2 (-8 to 5)	0.56
Body image	71 (26)	68 (28)	-3 (-14 to 8)	0.56
Role	78 (16)	77 (21)	-1 (-10 to 9)	0.86
Weight ^a	67 (33 to 100)	100 (50 to 100)	-	0.06 ^b
Respiratory symptoms	55 (24)	64 (19)	9 (-3 to 20)	0.13
Digestive symptoms ^a	89 (67 to 100)	89 (67 to 100)	-	0.97 ^b

Table 4.6 Participant (n = 21) responses to the Cystic Fibrosis Questionnaire-Revised prior to and following telehealth intervention

Data are presented as mean (standard deviation) for normally distributed data, or ^a median (IQR) for non-normally distributed data. An increase in score represents improvement, with a maximum score of 100. Analysis performed using paired t test or ^b Wilcoxon signed rank test. CI = confidence interval, diff = difference.

Chapter 4: Study 2

4.3 Part A Discussion

This study appears to be the first in Australia to evaluate the uptake of telehealth clinics implemented as part of routine care in people with CF living in rural and remote areas. The results have demonstrated that telehealth technology increases access to care for adults with CF and resulted in 90% of participants meeting the recommended number of CF clinic attendances (33) during the study period. Although both telehealth and traditional in-person CF clinics were offered, participants utilised telehealth for more than two-thirds of their clinics. One participant did not attend the CF centre (for an in-person clinic or hospital admission) as it was not convenient for them during the study period. This is a potential limitation of telehealth, as certain assessments (e.g. bone mineral density, general respiratory function testing and chest computed tomography) are not readily available in all rural and remote areas. While one participant chose not to utilise telehealth, the inclusion of their data in measures related to HCU provides a more conservative and real-world estimate of the effect telehealth had on these outcomes.

The observed increase in antibiotic use and hospital admissions reflects increased surveillance of the participants, and a corresponding increased detection of respiratory exacerbations. Although this short-term increase in HCU was observed, the increased frequency of CF clinics attended and contact with the clinical team as a result of this intervention may lead to increased treatment adherence and improved longer-term health outcomes. Specifically, more frequent contact with the CF team (154) and earlier detection of respiratory exacerbations that will allow treatment to be initiated sooner (19), and it is a broadly held view of clinicians that this may slow the magnitude of decline of FEV₁, and potentially improve nutritional status and HRQoL.

Recruitment for this study was good (over 80%), with only five potential participants not enrolling because they were required to travel to Perth for other medical reasons, or had intentions to move during the study period. The most likely explanation for the high recruitment rate is that potential participants could envisage the reduction in travel burden that would result from telehealth, as most of the participants were required to travel many hours by car or plane to

Perth to attend in-person clinics. It is also likely that recruitment was influenced by the mean age (31.4 years) of the study sample, as younger adults may be more open to the idea of integrating modern technology into their care. This study adds support to previous studies reporting that people with CF are willing to utilise telehealth technology (142).

Responses to the TeSS reflected an excellent level of satisfaction with telehealth clinics. The only item given a score less than three (out of 4) by a participant was regarding the voice quality of the equipment, which could be attributed to the voice delay occasionally experienced with the system. The TeSS responses demonstrated that participants perceived telehealth clinics to be a highly acceptable method to provide regular healthcare review. However, responses to the purpose developed participant satisfaction survey indicated that some participants (12%) preferred in-person clinics over telehealth and some had no preference (29%). This is understandable given that in-person clinics have been offered for many years and have advantages such as the ability for physical examination, access to on-site specialist secondary services (e.g. intravascular access, endocrinology, immunology and hepatology) and immediate admission to hospital if required.

In keeping with the high level of satisfaction observed in this study, a meta-analysis of telehealth for chronic obstructive pulmonary disease found that people were largely satisfied with receiving care via telehealth at home, as long as face to face consultations were still available on request (154). The positive view of telehealth amongst participants in this study is consistent with previous telehealth studies in CF (142), and provides merit to the continuation of its use as part of routine CF care.

It is difficult to make comparisons between the HCU results of this study and other telehealth studies in CF due to the differences in the interventions and study designs. In another study, weekly videoconferencing with the CF team has previously been utilised by a small number of people with CF (n = 7) awaiting lung transplantation over a six-month period, and no differences in HCU were found when compared to usual care (26). Videoconferencing has also been used effectively as a means of assessing functional exercise capacity (27). In 19 people

with CF from adult and paediatric centres, the monitoring of spirometry and respiratory symptoms on a daily basis at home over a 6-month period resulted in no difference in the number of respiratory exacerbations per participant when compared to the same period prior to the study; however an increase in courses of oral antibiotics was noted during the intervention (25). Another intervention comparing daily monitoring of spirometry and oximetry (n = 17) to usual care (n = 28) over seven months found no difference in the rate of hospitalisation, but an increase in CF clinic attendance during the study period was observed (21). Thus, despite the differences in study designs across these studies, many observed an increase in certain measures of HCU.

As this study was unlikely to be powered to detect small differences in the change in magnitude of spirometry (FEV₁ and FVC), the lack of difference in these measures were not unexpected. Future studies may address this lack of demonstrated change in spirometry with larger sample sizes and/or longer surveillance periods, given the variability of this outcome seen in people with CF. The increased treatment of respiratory exacerbations may explain the improvement seen in the vitality domain of the CFQ-R, which consists of four questions relating to energy levels. Although the minimal clinically important difference for the vitality domain of the CFO-R has not been established, the magnitude of change in this domain demonstrated in our study (11 out of 100 points) exceeded the magnitude of change (8 points) reported following 28 days of inhaled Aztreonam Lysine in people with CF colonised with *Pseudomonas aeruginosa* in another study (156). As the change in HROoL alongside the improvement in lung function was considered to be clinically meaningful to the participants in this other study, it is likely that the improvement seen in the present study would also be perceived as clinically meaningful (157). The trend towards improvement in the weight domain of the CFQ-R, alongside the weight gain in 14 (67%) of the participants, may reflect an increase in dietetic input delivered during the study period, as well as the improved management of respiratory exacerbations. It is well established that in people with CF, a loss of appetite and weight are associated with respiratory exacerbations (87) and earlier management of such episodes may have minimised weight loss during the study period.

This study demonstrated that telehealth clinics are a feasible way to deliver care for people with CF living in rural and remote areas, and as a result this service will continue to be offered to adults with CF living in WA. Adults with CF in these areas are willing to utilise this technology and most prefer telehealth over in-person clinics offered at the SCGH CF Centre. There were limitations to this study, such as a small sample and lack of control group, which are also common limitations among many of the published telehealth studies in CF to date (142). Furthermore, other clinical measures commonly used in CF care such as sputum microbiology, inflammatory and diabetic markers, and chest radiography were not included as outcomes as they were beyond the scope of this study.

Given the differences in geography, CF centre resources and government health service provision between states and countries throughout the world, it would be difficult to conduct a multicentre randomised controlled trial in this area. Therefore, future studies for similar cohorts should consider focusing on the evaluation of the longer-term implementation of telehealth services as part of routine care, and their impact on a wider range of health outcomes.

4.4 Part B Design

The study was a sub-analysis of data collected during Study 2, relating to absenteeism and presenteeism data collected from the WHO HPQ (158).

4.4.1 Study measurements

The principal investigator collated all data pertaining to the measurements presented in this study, which were collected at the same time as the measurements described in Part A of this chapter.

4.4.1.1 Absenteeism and presenteeism

During CF clinics (telehealth and in-person) throughout a 12-month period, participants completed the absenteeism and presenteeism questions of the WHO HPQ (158). This selfreported 11-item questionnaire collects information regarding; i) expected and actual hours worked during the previous 7 and 28 days; ii) part or entire days of work missed (for health or holiday); iii) overtime worked; and, iv) perceived work performance (rated out of 10) for the participant and colleagues undertaking similar work. Responses were used to calculate relative absenteeism and relative presenteeism. Relative absenteeism is the proportion of time absent from work. A maximum score of 1 indicates 100% absenteeism; a score of 0 indicates no absenteeism. Negative scores indicate the hours worked exceeded employer's expectations. Relative presenteeism is work performance compared to others in similar employment. A score of 1 represents equal performance, 0 represents worst performance, and a maximum score of 2 represents performance at least twice as good as colleagues.

Chapter 4: Study 2

4.5 Part B Results

Twenty-three adults with CF completed the study (Table 4.1). Data were collected during a total of 91 CF clinics attended. Participants were engaged in paid work on 61 (67%) of these occasions. Twenty-two (96%) participants were employed on at least one occasion, and 11 (48%) were employed on all occasions throughout the entire 12 months. Only one participant, a fulltime carer, had no paid work in this time.

For those employed, the median (range) number of hours worked per week was 38 (10 to 80). The mean (range) number of entire work days missed in the previous 28 days because of problems with physical or mental health was 2 (0 to 21) days, while the number of partial work days missed in this period for the same reason was to 1 (0 to 11) days. Participants worked longer hours than expected on 2 (0 to 28) days during the previous 28 days. Overall, relative absenteeism (mean [range]) in those employed was 0.1 (-1.2 to 0.9).

Participants reported that their work performance over the past one to two years was median (range) 8 (5 to 10) out of 10, which was similar to how they rated the performance of colleagues in similar employment, also being 8 (5 to 10). However, when asked to rate their work performance over the past 28 days, the participants' responses were more variable (8 [0 to 10]). Overall, median (range) relative presenteeism in those employed was 1.0 (0.0 to 1.8).
Chapter 4: Study 2

4.6 Part B Discussion

These data provide evidence of the potential impact of CF on a person's ability to work. Despite having a chronic disease that is associated with a significant treatment burden, most adults with CF living in rural and remote WA engaged in paid work during the 12-month study period. Absenteeism scores were variable, and demonstrated that CF affected the work attendance of many, but not all participants. Presenteeism scores, while also variable, demonstrated that on average participants felt they performed as well as their colleagues at work. However, with many reporting that CF does have has a large impact on their work performance, further research is required in larger samples including both metropolitan and rural and remote areas to understand the full impact of CF on presenteeism. Measuring absenteeism and presenteeism using the HPQ provides a useful insight into the disease-related impact on work attendance and performance in adults with CF, and this outcome measure should be considered for use in future studies in this population.

CHAPTER 5

STUDY 3 (PART A, PART B)

This chapter presents the methodology, results and discussion for Study 3 (described in Section 1.3), which investigates the usability of a smartphone application (app) amongst adults with cystic fibrosis (CF) (Part A), and the agreement of clinical decision making between clinicians (experienced in providing CF care) interpreting the app data (Part B). These are preliminary studies undertaken to demonstrate feasibility prior to the commencement of Study 4 (Chapter 6). The methods and results for Parts A and Part B are presented separately, followed by a combined discussion relevant to both parts. Study 3 was published as original research in the *Journal of Telemedicine and Telecare* in November 2017 (Appendix 5).

Chapter 5: Study 3

5.1 Part A Design

This was a single group study exploring usability over four weeks duration, conducted at the Sir Charles Gairdner Hospital (SCGH) CF Centre. Approval was obtained from the Sir Charles Gairdner Osborne Park Healthcare Group (2015-209) and Curtin University (HR123/2015) Human Research Ethics Committees.

5.1.1 Recruitment

While attending a CF clinic, 10 adults with CF who owned a smartphone with iOS or Android operating systems were invited to participate. Recruitment was consecutive and targeted (ceasing recruitment of each sex at five participants) to ensure equal numbers of male and female participants. In total, 190 adults with CF attended the centre at the commencement of the study, and at this time 183 (96%) owned a smartphone.

5.1.2 Smartphone application

Experienced clinicians who were also the investigators in this study formulated the 12 questions used in the app (described further below). Members of the research team reviewed the questions for face validity including completing the questions several times to assess readability. Eight of the respiratory symptom questions were derived from exacerbation scales previously used in CF research (18, 83) (Table 5.1) (see Section 2.3.1) and two further symptom questions (i.e. those pertaining to wheeze and chest tightness) were included as these symptoms have been identified by adults with CF as indicators of a respiratory exacerbation (95). Finally, two questions pertaining to emotional wellbeing were included, because in people with CF, greater feelings of anxiety and depression have been associated with the greater respiratory symptoms, as well as poorer lung function, HRQoL and physical functioning (159). All of the app questions were answered yes or no, and each time an answer was selected the app automatically moved to the next question. The font size of the text in the app was set at a minimum of 18 to ensure all users could read the questions.

Table 5.1 Smartphone application questions

A prototype of the app was designed by a software engineer, and internal testing was then undertaken by several members of the research team to ensure the application was functioning as intended. The final version of the app was then made available free of charge on iTunes and the Google Play Store, allowing participants to install it on their smartphone after enrolment in the study.

5.1.3 Procedure

At enrolment in the study, participants installed the app and practised answering the questions once in the presence of the principal investigator. They were then required to answer the questions once a week over four consecutive weeks. Four occasions (over a 4-week period) of use was considered sufficient by the research team, given another study that explored the usability of a telehealth system in CF based their conclusion on a single use (27). Participants in the current study were asked to set a recurring alert on their smartphone, which prompted them to answer the questions on a pre-specified day and time each week of their choosing. To login securely, each participant had a unique four-digit user code. Once all questions were answered, the responses were emailed in binary code (1 = yes, 0 = no), along with the user code, to an account established for the purpose of the study. To maintain data security, only the principal investigator could identify each participant based on their user code. Participants were informed that for the purpose of this study, symptom responses were not being used by the CF team to detect respiratory exacerbations, and they were instructed to contact the CF clinic by phone if they required an appointment. Participants were not reviewed in the CF clinic during the fourweek period, and therefore no comparison was undertaken regarding the app answers and clinical status.

5.1.4 Study measurements

The principal investigator administered all assessments at the follow-up visit, 4 weeks following enrolment.

Chapter 5: Study 3

5.1.4.1 System usability

Usability was measured at the end of the study period with the System Usability Scale (SUS) (160). This scale comprises 10 items that ask about the usability aspects of the new technology, such as its complexity and functionality, and the participant's confidence in using it. Each item was answered using a 5-point Likert scale with responses that ranged from 5 (strongly agree) to 1 (strongly disagree). Scores from each of the 10 items were scaled to provide a score out of four, then combined and multiplied by 2.5 to provide a total score out of 100. Scores from the SUS were also grouped into two subscales: usability (8 items) and learnability (two items), both of which were also scored out of 100. The SUS has been widely used in the information technology industry, and more recently in studies evaluating the usability of telehealth technologies in CF (27) and chronic obstructive pulmonary disease (137).

5.1.4.2 Participant satisfaction

Participant satisfaction was assessed at the end of the study period using a purpose-developed survey which comprised seven items relating to the design and use of the app. Items alternated between positive and negative wording, and were answered with a 5-point Likert scale with responses that ranged from 5 (strongly agree) to 1 (strongly disagree).

5.1.5 Analyses

Statistical analyses were conducted using SPSS® (Statistical Package for Social Sciences, version 22.0, IBM corp., New York, USA). Data from both the System Usability Scale and satisfaction survey were reported using descriptive statistics.

5.2 Part A Results

Ten participants with CF completed the study (Table 5.2). Each participant used the app once per week over the four-week study period. Seven (70%) participants used iPhones, and three (30%) used phones with Android operating systems. At recruitment, the observed time taken to complete the app questions following installation was less than 2 minutes for all participants.

5.2.1 System usability

The mean (standard deviation) raw SUS score was 94 (6) (Table 5.3). When summed as subscales, participants scored the smartphone application as 92 (7) and 100 (0) for usability and learnability, respectively.

5.2.2 Participant satisfaction

Nine (90%) participants responded that they believed the app was a good method for reporting symptoms to the CF team. All participants (100%) found the instructions in the app easy to follow (Table 5.4).

Male, n (%)	5 (50)
Age, yr	32 (11)
BMI, kg/m ²	23 (5)
ppFEV ₁ , %	49 (27)
Rural or remote, n (%)	2 (20)

Table 5.2 Participant (n = 10) characteristics

Data are presented as number (percentage) or mean (standard deviation). BMI = body mass index; $ppFEV_1$ = percentage predicted forced expiratory volume in one second. The Global Lung Function Initiative 2012 equations (151) were used to calculate percent predicted $ppFEV_1$.

Item	SA	А	U	D	SD
I think that I would like to use this system frequently	5 (50)	4 (40)	1 (10)	0 (0)	0 (0)
I found the system unnecessarily complex	0 (0)	0 (0)	0 (0)	1 (10)	9 (90)
I thought the system was easy to use	9 (90)	1 (10)	0 (0)	0 (0)	0 (0)
I think that I would need the support of a technical person to be able to use this system	0 (0)	0 (0)	0 (0)	0 (0)	10 (100)
I found the various functions in this system were well integrated	4 (40)	3 (30)	3 (30)	0 (0)	0 (0)
I thought there was too much inconsistency in this system	0 (0)	1 (10)	0 (0)	2 (20)	7 (70)
I would imagine that most people would learn to use this system very quickly	10 (100)	0 (0)	0 (0)	0 (0)	0 (0)
I found the system very cumbersome to use	0 (0)	0 (0)	0 (0)	2 (20)	8 (80)
I felt very confident using the system	9 (90)	0 (0)	1 (10)	0 (0)	0 (0)
I needed to learn a lot of things before I could get going with this system	0 (0)	0 (0)	0 (0)	0 (0)	10 (100)

Table 5.3 Participant (n = 10) responses to items of the System Usability Scale

Data are presented as number (percentage of participants). Each item is answered using a 5-point Likert scale: SA = Strongly Agree, A = Agree, U = Unsure, D = Disagree, SD = Strongly Disagree.

Item	SA	А	U	D	SD
I liked the graphic design (layout, colours etc.) of the app	0 (0)	8 (80)	2 (20)	0 (0)	0 (0)
It was easy to follow the instructions on the app	8 (80)	2 (20)	0 (0)	0 (0)	0 (0)
It was difficult to know what the app was asking me to do next	0 (0)	0 (0)	0 (0)	6 (60)	4 (40)
The questions I was asked about my symptoms were easy to understand	5 (50)	3 (30)	0 (0)	2 (20)	0 (0)
The questions I was asked about how I felt were difficult to understand	0 (0)	1 (10)	0 (0)	5 (50)	4 (40)
Using the app was a good way for me to report my symptoms to the CF team	4 (40)	5 (50)	1 (10)	0 (0)	0 (0)
Using the app to report my symptoms was an inconvenience or intrusion into my daily life	0 (0)	0 (0)	0 (0)	3 (30)	7 (70)

|--|

Data are presented as percentage of participants. SA = Strongly Agree, A = Agree, U = Unsure, D = Disagree, SD = Strongly Disagree.

5.3 Part B Design

This was an observer agreement study, investigating the agreement between experienced senior clinicians interpreting scenarios containing simulated app responses.

5.3.1 Study measurements

The principal investigator collated all data pertaining to the measurement collected as part of this study.

5.3.1.1 Observer agreement

A physician and nurse practitioner, both of whom were experienced in providing CF care, were presented with an identical set of 45 clinical scenarios, developed by the principal investigator. A nurse practitioner was included to reflect the growing utilisation of this role as a first point of contact in CF centres internationally. These scenarios were created by pairing one of three case studies, with one of 15 different combinations of app responses (45 in total). The three case studies described clinical presentations of an adult with mild, moderate or severe CF (i.e. each clinical presentation provided 15 differing combinations of app responses) and were reviewed for authenticity by a respiratory physician with extensive experience in CF, who was not otherwise involved in the study. Each scenario contained a different combination of possible responses to the app questions used in Part A. For each scenario, the two clinicians were asked to indicate whether or not they would; i) initiate telephone contact to discuss symptoms further, and/or ii) request the person attend an outpatient clinic for further assessment. The clinicians were blinded to each other's responses. An example scenario is provided as Appendix 6.

5.3.2 Analyses

Observer agreement between clinicians was assessed using a Cohen's kappa statistic. A sample size of 45 responses was adequate to detect a difference in proportion between chance and anticipated agreement of 0.5, assuming a relative error of 30%. Alpha was set at 0.05.

88

Chapter 5: Study 3

5.4 Part B Results

5.4.1 Observer agreement

Observer agreement between clinicians was perfect for the question regarding whether or not to initiate contact with the person to further discuss symptoms ($\kappa = 1.0$, p < 0.001). Specifically, both clinicians indicated that they would initiate contact for 42 (93%) scenarios, but not for three (7%) scenarios. There was near-perfect agreement for whether or not the person should be advised to attend the CF clinic ($\kappa = 0.86$, p < 0.001). Specifically, both clinicians indicated that they would request the person attend the CF clinic for 24 (53%) scenarios, but not for 18 (40%) scenarios. The two clinicians disagreed on the same scenario across all three case studies; this scenario contained a 'yes' response to only one question, which asked "in the past week have you had increased cough or new pain on coughing?"

5.5 Part A and Part B Discussion

This study has shown that an app that can be used to report symptoms that may be suggestive of a respiratory exacerbation had excellent usability in adults with CF. Furthermore, the symptom responses from the app were interpreted in a similar way by different senior clinicians in the process of clinical decision making.

The integration of telehealth with traditional care in CF is an expanding area of research. To date, feasibility in CF has been demonstrated in interventions such as using Skype and WhatsApp as a method of communication (143), monitoring of lung function and symptoms at home (21, 22, 97, 145), assessing exercise capacity (27), and promoting exercise (28, 147) and physical activity (29). While the evidence supporting the role of telehealth in CF is relatively limited, the aim of such interventions in the longer-term will be to improve health outcomes the earlier detection of respiratory exacerbations and increased treatment adherence. Strategies similar to those used in CF have also been employed in chronic diseases such as type 1 diabetes mellitus (161), with demonstrated improvements in access to routine care. A systematic review into mobile health interventions in diabetes mellitus and other cardiovascular and respiratory diseases found mixed evidence supporting the effectiveness of these interventions on treatment adherence and health outcomes (162).

The use of out-of-hospital monitoring systems to detect changes in symptoms and physiological parameters suggestive of a respiratory exacerbation in CF is a growing area of interest (21, 22, 96, 145). One shortcoming of these previous studies was poor adherence to the monitoring system, which resulted, at least in part, from the equipment being large and bulky. The app described in the present study overcame this limitation, as it was small and portable, and integrated easily with the lifestyle of adults with CF. Further, in contrast with earlier work which required daily use of the monitoring system by all participants, those in the current study were asked to use the system less frequently (i.e. once a week); a factor which may also have contributed to the excellent adherence observed. While the optimal frequency of reporting in order to detect a respiratory exacerbation is unknown, daily reporting in other studies probably

90

contributed to the observed poor adherence. Therefore, weekly reporting appears to be a pragmatic alternative which was well tolerated in this small study. In clinical practice, to minimise the possibility of missing the onset of a respiratory exacerbation that occurred between the weekly use of the app, people with CF could be instructed to use the app at any time between scheduled uses if they felt their respiratory symptoms had worsened.

The high SUS scores indicated that the app achieved an 'A' rating for usability. This rating is at the top of the "acceptable" usability range, and is described on the SUS adjective rating scale as "best imaginable" usability (163). In addition, raw scores of over 90 on the SUS indicate that a system is in the 99th percentile of usability for the tested demographic. Finally, a very high score like this also indicates that users would be more likely to recommend this technology to a friend.

The SUS scores observed in this study were similarly high to those reported for interventions in adults with CF investigating the feasibility of an internet based tool to promote physical activity (29), and using a videoconferencing system to assess exercise capacity (27). This was not surprising given the uncomplicated design and functionality of the app, including asking for yes/no responses to questions and requiring minimal time to answer the questions (less than 2 minutes). Similarly, participant satisfaction, as measured using the purpose developed survey was very high. All of the participants owned smartphones prior to the study, and therefore the high level of usability and satisfaction observed may be a function of their pre-existing comfort with such technology. This was expected, as 74% of adults in Australia owned a smartphone in 2015 at the time of this study (164).

Although the sample size for Part A of this study was modest, participants were recruited across a broad range of ages and severity of lung function, and there was equal representation from both genders. Participants living in rural and remote areas (n = 2) also participated in the study. Seven (70%) had received treatment for a respiratory exacerbation in the 12 months prior to commencing the study. Participants were employed in a wide variety of occupations, including administration, retail, information technology and healthcare, and one was a fulltime university

91

student. This allowed the investigators to gather usability data from people who were representative of the broader adult CF population.

In Part B, near-perfect agreement was observed in clinical decision making between two experienced clinicians. This suggests that others may be able to use this app to arrive at similar clinical decisions regarding whether the person requires further assessment for a potential respiratory exacerbation. However, having only two clinicians participating in the observer agreement study was a limitation. It is also a limitation that the scenarios were also simulated and did not reflect true clinical practise. It is possible that for the question with poor agreement, the ambiguity related to asking the clinician to interpret responses regarding two different symptoms (increased cough and new pain on coughing). This ambiguity might be overcome by dividing this into two separate questions: one regarding increased cough and a second regarding chest pain. This will allow people with CF to report chest pain irrespective of whether it is related to coughing.

This app, while designed to detect respiratory exacerbations, clearly did not include spirometry. This is a potential limitation as FEV₁ is regarded as one of the key measures of clinical status in CF (52). However, the provision of a spirometer would have greatly increased cost and reduced the portability of the reporting system. A system was developed that required no additional equipment to the participant's own smartphone, leading to high usability. The app questions, which provided the participant with a means to report respiratory symptoms, will allow the CF team to determine whether spirometry is required. It is acknowledged however, that the inclusion of home spirometry as part of this system may provide more accurate diagnosis of a respiratory exacerbation prior to the person being contacted by the CF team or attending the CF clinic.

The results of this study have demonstrated that amongst adults with CF, an app used to report symptoms suggestive of a respiratory exacerbation has excellent usability, and provides symptom responses that can be interpreted consistently between clinicians who are experienced in the management of this clinical population. A randomised controlled trial is now warranted to investigate whether the use of this app can reduce the delay in time taken by an adult with CF to report respiratory symptoms to a CF team, and measure its impact on health-related outcomes.

CHAPTER 6

STUDY 4

This chapter presents the methodology, results and discussion for Study 4 (described in Section 1.3), which was a prospective randomised controlled trial (RCT) investigating the impact of the weekly use of the smartphone application (app) on healthcare utilisation and other health outcomes in adults with cystic fibrosis (CF). The methods from Study 4 were published as a protocol in the *BMJ Open* in April 2018 (Appendix 7). After completion, Study 4 was published in the *Journal of Cystic Fibrosis* in September 2019 (Appendix 8). Further, baseline data from Study 4 was published in *Respiratory Care* in March 2019 by a different author following an Honours degree project (see 'Publications arising using data from this thesis'). Additional results and discussion regarding unpublished data relating to secondary outcome measures have been incorporated into this chapter.

6.1 Study design

This was a prospective 12-month, single blinded RCT. Ethical approval was obtained from the Sir Charles Gairdner Osborne Park Healthcare Group (2015 - 030) and Curtin University (HR212/2015) Human Research Ethics Committees. This trial was registered with the Australian and New Zealand Clinical Trials Registry (ACTRN12615000599572), and reported in accordance with the Consolidated Standards of Reporting Trials (CONSORT) statement (165) (Appendix 7).

6.1.1 Participant screening and selection

Eligible participants were identified at routine outpatient clinic appointments at the Sir Charles Gairdner Hospital (SCGH) CF Centre in Perth.

6.1.1.1 Inclusion criteria

Adults who met the following criteria were eligible to participate: diagnosis of CF; age ≥ 18 years; under the care of the SCGH CF Centre; had a respiratory exacerbation requiring treatment with intravenous (IV) antibiotics in the preceding 12 months; were in a period of clinical stability, defined as no signs of a respiratory exacerbation for the previous 2 weeks; and were able to understand written and spoken English.

6.1.1.2 Exclusion criteria

Criteria to exclude people from participating were: previous lung transplantation or current listing for lung transplantation; inability to undertake a prescribed treatment regimen due to cognitive reasons, or inability to use or not being in possession of a smartphone.

6.1.2 Randomisation and allocation concealment

Participants were randomly allocated to the intervention (usual care plus experimental intervention) or control (usual care) group using the University of Sydney National Health Medical Research Council Clinical Trials Centre randomisation service. Randomisation was stratified according to sex, location (metropolitan vs. rural and remote) and whether or not the participant was prescribed Ivacaftor or Lumicaftor/Ivacaftor, as these medications can reduce the rate of respiratory exacerbations and improve lung function, weight and health-related quality of life (HRQoL) (39, 166).

6.1.3 Study measurements

Participants had assessments performed prior to randomisation (i.e. at baseline), and at 6 and 12 months (i.e. final), unless otherwise stated. Assessments were performed when the participant was in a period of clinical stability, defined as no signs of a respiratory exacerbation for the previous 2 weeks. If the participant had signs of a respiratory exacerbation at their scheduled follow-up time, the assessments were completed at the subsequent CF clinic when they were clinically stable. The principal investigator performed all assessments during follow-up visits conducted at SCGH. For telehealth assessments, spirometry and weight were assessed by a health professional local to the participant, with the remaining measures collected by the principal investigator.

6.1.4 **Primary outcome**

6.1.4.1 Number of courses of intravenous antibiotics and intravenous antibiotic days

Information regarding the use of IV antibiotics was obtained from the medical records of the participants by a pharmacist blinded to group allocation and verified against SCGH pharmacy dispensing records. This was collected at the final assessment only.

6.1.5 Secondary outcomes

6.1.5.1 Other measures of healthcare utilisation

At the final assessment only, the participant's medical record was reviewed to extract information pertaining to healthcare utilisation (HCU). This included: time (in days) from randomisation to the detection of the first respiratory exacerbation requiring oral and/or IV antibiotics; number of hospital admissions and days; number of courses and days of oral and inhaled antibiotics; number of CF clinics attended; estimated cost of IV, oral and inhaled antibiotic treatment; and estimated cost of hospital admissions. Costs related to HCU were estimated using information provided from the SCGH Pharmacy and Finance departments. All

potential participants received their care at the SCGH CF Centre or via shared care with an external respiratory physician (but not from a general practitioner). Prescribed antibiotic use was documented in the medical record, and participants were not required to keep a diary of antibiotic use. Non-clinical costs relating to participants' travel, parking, time off work and telephone calls were not recorded.

6.1.5.2 Smartphone application adherence

Adherence to the weekly use of the app for participants in the intervention group, expressed as a percentage, was calculated by dividing the number of weeks successfully reported by the total number of weeks enrolled in the study (expected to be 52 weeks), and multiplying this number by 100.

6.1.5.3 Smartphone application symptom responses

Participant responses to the symptom questions were collated and reported at the completion of the study period.

6.1.5.4 System usability – System Usability Scale

The System Usability Scale (SUS) (160) is an assessment tool comprising 10 questions regarding the usability of technological systems, and is answered on a 5-point Likert scale (discussed in Section 5.1.4.1).

6.1.5.5 Spirometry

Spirometry (forced expiratory volume in 1 second [FEV₁] and forced vital capacity [FVC]) was measured using a Medgraphics USB spirometer (MGC Diagnostics, Minnesota, USA), or Easyone spirometer (ndd Medical Technologies, Massachusetts, USA). Each participant performed up to eight maximal forced expiratory manoeuvres to ensure the two best attempts met published standards (152). The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted FEV₁ (ppFEV₁) and percentage predicted FVC (ppFVC).

6.1.5.6 Nutritional status - body mass index

Participants had their height (CE0123, Seca, Hamburg) and weight (UC-321, A&D Company, Tokyo) measured and body mass index (BMI) was calculated.

6.1.5.7 Health-related quality of life – Cystic Fibrosis Questionnaire – Revised

This self-completed questionnaire comprises 50 questions answered on a 4-point Likert scale. It covers 12 domains comprising physical, vitality, emotion, treatment burden, health, social, body, role, weight, respiratory, digestion and eating (107).

6.1.5.8 Feelings of anxiety and depression – Hospital Anxiety and Depression Scale

The Hospital Anxiety and Depression Scale (HADS) (108) contains 14 self-reported questions, evenly divided into anxiety and depression scales (7 each). A 4-point Likert scale (0 to 3) is used to answer questions based on a recall period of 7 days.

6.1.5.9 Treatment adherence – Treatment Adherence Questionnaire – Cystic Fibrosis

Adherence to routine CF treatments was assessed using the Treatment Adherence Questionnaire – Cystic Fibrosis (TAQ-CF) (167). This measure has 12 treatment items and asks the user to report on the frequency of each treatment on a 7-point Likert scale (ranging from "not at all" to "3 or more times per day"), and the duration of treatment on a 6-point Likert scale ("0" to "25+" minutes). The TAQ-CF also asks the participant to report on the barriers to individual treatments. The wording of certain medications and barriers was adjusted to be relevant to an Australian CF cohort (Appendix 10).

6.1.5.10 Absenteeism/presenteeism – World Health Organisation's Heath and Work Performance Questionnaire

Absenteeism is defined as being absent from work or study due to illness whereas presenteeism is defined as the loss of productivity associated with attending work or study when unwell (168). This questionnaire uses the absenteeism and presenteeism questions of the World Health Organisation's Heath and Work Performance Questionnaire (HPQ) (158) (discussed in Section 4.4.1.1).

Chapter 6: Study 4

6.1.6 Intervention group

An app was developed for use by participants in the intervention group to report respiratory symptoms on their own smartphones. The application contained 14 questions (Table 6.4) that are required to be answered 'yes' or 'no'. Twelve questions relate to respiratory symptoms; 10 questions were derived from the Fuchs exacerbation scale (18), with the addition of two symptoms (wheeze and chest tightness) commonly identified by adults with CF as indicators of a respiratory exacerbation (95) and considered important by the research team. The application also asked the participant to answer two questions relating to their feelings of anxiety and depression. The data were then transmitted securely via password protected email to members of the research team. The app was demonstrated to have high system usability in people with CF, as well as good observer agreement between experienced clinicians interpreting the app data (Chapter 5, Study 3).

Participants were asked to use the app once per week on a set day and time of their choosing, for a period of 12 months. A smartphone calendar alert was set by each participant to remind them to answer the symptom questions. Participants could also use the app earlier than the next weekly reporting time if they felt their respiratory symptoms had worsened. If the participant missed one of their weekly reporting times, they were messaged at the end of the week to remind them to answer the app questions the following week. If the participant responded 'yes' to any of the symptom questions, the principal investigator alerted the nurse practitioner, who then phoned the participant to discuss their respiratory symptoms and whether treatment (i.e. antibiotics) were required, in consultation with a physician. The nurse practitioner also determined if a formal review in the CF clinic was required. Both the nurse practitioner and physician were blinded to group allocation. If the participant's respiratory disease had been stable for 3 months (i.e. no increase in any respiratory symptoms), they were asked to attend a CF clinic as per standard of care guidelines (4).

99

Chapter 6: Study 4

6.1.7 Control group

Participants in this group continued to receive usual care, involving routine CF clinic appointments (approximately 3-monthly). Participants were able to contact the nurse practitioner via telephone sooner if there was a change in their clinical status.

6.1.8 Analyses

Statistical analyses were undertaken with Stata (StataCorp, Stata Statistical Software Release: V.14) in accordance with the intention-to-treat principle (169). Categorical data are presented as frequencies and proportions. Continuous data and count are presented as medians and ranges due to skewed distributions. General linear mixed models were used to examine continuous outcomes with normal distributions. Healthcare utilisation data are presented as count data with skewed distributions. Negative binomial regression was used to examine differences between groups for HCU, with this model chosen over Poisson as the data were over dispersed. Models were adjusted for baseline variables of sex, ppFEV₁, and courses of IV antibiotics in the preceding 12 months. Kaplan-Meier and Cox regression were used to compare time to detection of respiratory exacerbation. Where possible, point estimates are reported with their corresponding 95% confidence interval (CI). Alpha was set at 0.05.

6.1.8.1 Sample size calculation

To detect a between-group difference ($\alpha = 0.05$, $1 - \beta = 0.8$) in the number of IV antibiotic courses of one per participant, assuming a standard deviation (SD) of 1.21 (based on IV antibiotic usage at the SCGH CF Centre during 2013 and 2014), a sample size of 23 in each group was required. To account for any loss to follow-up or drop out, it was proposed to recruit seven additional participants to each group, for a total sample size of 60.

6.1.9 Data privacy

Data from the app were transmitted securely via password-protected email to members of the research team at SCGH, and not kept in 'cloud' storage. After transmission from the participant's phone or tablet, the data were automatically deleted, preventing any potential

100

breach of privacy if the device was lost or stolen. The application also requested the user to enter a unique, 4-digit passcode before being able to access the app questions.

6.2 **Results**

6.2.1 Participant characteristics

Of the 180 patients attending the CF centre at the commencement of the study, 70 (39%) met the inclusion criteria and were invited to participate in the study. The most common reason for a potential participant to be ineligible for the study was not having had a respiratory exacerbation requiring IV antibiotics in the preceding 12 months. Sixty participants were enrolled in the study (Table 6.1), with 29 (48%) allocated to the intervention group. One (3%) participant in the control group did not attend the 6- or 12-month study visits, and one participant in the intervention group received lung transplantation prior to the 12-month assessment (Figure 6.1).

	Intervention	Control
Participants, n (%)	29 (48)	31 (52)
Female, n (%)	17 (59)	12 (39)
Age, yr	31 (10)	31 (8)
$FEV_1(L)$	2.07 (0.85)	2.41 (0.82)
$ppFEV_1$	58 (18)	61 (17)
FVC (L)	3.06 (1.04)	3.68 (1.03)
ppFVC	71 (16)	78 (14)
BMI, kg/m ²	22.4 (5.0)	23.4 (2.9)
CFQ-R respiratory domain	67 (17)	71 (16)
IV antibiotics courses in past 12 months, n	60	45
Living in a rural / remote region, n (%)	4 (14)	6 (19)
Pseudomonas aeruginosa, n (%)	29 (100)	30 (97)
CF-related diabetes, n (%)	5 (17)	13 (42)
Pancreatic insufficiency, n (%)	23 (79)	29 (93)
Ivacaftor or Lumacaftor/Ivacaftor, n (%)	8 (28)	9 (29)

Table 6.1 Participants' characteristics at baseline (n = 60)

Data are presented as mean (SD) unless otherwise stated. BMI = body mass index, CFQ-R = Cystic Fibrosis Questionnaire – Revised, FEV_1 = forced expiratory volume in 1 second, FVC = forced vital capacity, IV = intravenous, pp = % predicted. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted ppFEV₁ and ppFVC.



Figure 6.1 Consolidated Standards of Reporting Trials (CONSORT) diagram

6.2.2 Number of courses of intravenous antibiotics and intravenous antibiotic days

Following adjustment for the courses of IV antibiotics in the preceding 12 months, sex, and $ppFEV_1$, there was no clear effect of the intervention on the number of courses and days of IV antibiotics per participant over the 12 month study period (Table 6.2).

6.2.3 Secondary outcomes

6.2.3.1 Other measures of healthcare utilisation

Over the 12 month study period, there was no clear effect of the intervention on the time to detection of a respiratory exacerbation requiring IV antibiotics, being a median (interquartile range [IQR]) of 186 (298) vs. 273 (184) days in the control group (p = 0.20; Figure 6.2). The time to detection of a respiratory exacerbation requiring oral or IV antibiotics was shorter in the intervention group compared to the control group with a median (IQR) of 70 (123) versus 141 (140) days, respectively (p = 0.02; Figure 6.2).

There was evidence to support an increase in the prescription of oral antibiotics in the intervention group, but not in the prescription of inhaled antibiotics (Table 6.2). There was no clear effect of the intervention on the number of hospital admissions, admission days and number of CF clinics attended (Table 6.2). Participants in the intervention group communicated via phone with the nurse practitioner a median (IQR) of 7 (8) times during the study, compared to 3 (4) times (p = 0.01) for participants in the control group. After adjusting for baseline variables, there was no clear evidence to support a difference in HCU costs (antibiotics, admission days and CF clinics) between the two groups (Table 6.3).

	Intervention (n = 29)		Control $(n = 31)$			
	Nº	per participant	Nº	per participant	IRR	95% CI
IV antibiotic courses	42	1 (0 to 4)	30	1 (0 to 6)	1	0.6 to 1.7
IV antibiotic days	609	14 (0 to 64)	468	14 (0 to 105)	1.1	0.5 to 2.6
Oral antibiotic courses	71	2 (0 to 8)	46	1 (0 to 5)	1.5	1.0 to 2.2
Inhaled antibiotic courses	24	0 (0 to 4)	18	0 (0 to 4)	1	0.7 to 1.4
Hospital admissions	27	1 (0 to 4)	21	0 (0 to 5)	1	0.5 to 2.0
Admission days	244	4 (0 to 44)	214	0 (0 to 57)	0.9	0.5 to 2.0
CF clinics attended	249	9 (1 to 21)	216	6 (1 to 22)	1.1	0.8 to 1.4

 Table 6.2 Participants' (n = 60) healthcare utilisation data

Data are presented as number (N°) and median (range) per participant. Analyses performed using negative binomial regression, adjusted for sex, forced expiratory volume in 1 second percentage predicted, and courses of IV antibiotics in the preceding 12 months, measured prior to randomisation. CI = confidence interval. IRR = incidence rate ratio. IV = intravenous.



Figure 6.2 Time to detection of respiratory exacerbation requiring a) intravenous antibiotics; and b) oral or intravenous antibiotics plotted against time (days).

Data are presented as the proportion (%) of participants. Analyses performed using Kaplan-Meier, Cox regression.

a) No difference in time to detection of respiratory exacerbation requiring intravenous (IV) antibiotics between the intervention (n = 29) and control (n = 31) groups (median ([IQR] of 186 [298] vs. 273 [184] days, respectively), p = 0.20;

b) Shorter time to detection of respiratory exacerbation requiring oral or IV antibiotics between the intervention and control groups (median [IQR] of 70 [123] vs. 141 [140] days, respectively), p = 0.02.

Table 6.3 Healthcare utilisation costs

Intervention $(n = 29)$		Control $(n = 31)$				
	Cost	per participant	Cost	per participant	IRR	95% CI
Antibiotics	\$111,871	\$3,858 (\$6 to \$29,452)	\$48,939	\$1,579 (\$0 to \$12,598)	1.1	0.6 to 2.1
Admission days	\$319,303	\$11,010 (\$0 to \$57,579)	\$280,044	\$9,034 (\$0 to \$75,591)	0.8	0.3 to 2.2
CF clinics	\$164,741	\$5,681 (\$662 to \$13,894)	\$142,908	\$4610 (\$662 to \$14,555)	1.0	0.7 to 1.4

Data are presented as total or mean (range) in Australian dollars. Antibiotic costs were provided by the Sir Charles Gairdner Hospital (SCGH) Pharmacy Department. Admission days and CF clinic costs were provided by the SCGH Department of Activity Based Costing. Analyses performed using Poisson regression, adjusted for sex, forced expiratory volume in 1 second and courses of intravenous antibiotics in the preceding 12 months, measured prior to randomisation. CI = confidence interval, IRR = incident rate ratio.

6.2.3.2 Smartphone application adherence

Adherence to the weekly use of the app was a mean (range) of 77 (25 to 100)%, with 1,136 of a possible 1,483 reporting occasions completed. Subgroup analysis demonstrated an increase in oral antibiotic prescription and reduction in IV antibiotic prescription compared with the 12 months preceding the study for participants in the intervention group with adherence \geq 80% (n = 15) (Figure 6.3).

6.2.3.3 Smartphone application symptom responses

Of the completed reporting occasions whilst using the app, participants in the intervention group reported an increase in at least one symptom 399 (35%) times (median [range] 10 [1 to 44]) per participant). Oral or IV antibiotics were subsequently prescribed on 113 (28%) of these occasions. The most commonly reported symptoms were increased cough, increased fatigue or lethargy, and worsening sputum volume or colour (Table 6.4).



Figure 6.3 Intravenous and oral antibiotic use in the 12 months prior to randomisation and during the study for a) participants in the intervention group who were $\geq 80\%$ (n = 15); and b) < 80% adherent (n = 14) to the weekly use of the smartphone application

Data are presented as the total number of courses. Analyses performed using negative binomial regression. IV = intravenous.

a) An increase in oral antibiotic prescription (p = 0.02) and reduction in IV antibiotic prescription (p = 0.03) was observed during the intervention;

b) No change in oral or IV antibiotic prescription during the intervention.

In the past week, have you had:	Nº	per participant
Worsening sputum volume or change in colour?	144	3 (0 to 22)
New or increased blood in your sputum?	50	0 (0 to 10)
Increased cough?	184	5 (1 to 28)
New or increased chest pain?	53	1 (0 to 20)
New or increased wheeze?	101	2 (0 to 25)
New or increased chest tightness?	124	2 (0 to 20)
Increased shortness of breath or difficulty breathing?	124	3 (0 to 19)
Increased fatigue or lethargy?	171	4 (0 to 28)
A fever?	39	0 (0 to 18)
Loss of appetite or weight?	52	1 (0 to 7)
Sinus pain or tenderness?	75	1 (0 to 19)
In the past week do you feel that your health has worsened?	143	4 (0 to 22)
In the past week, have you felt:		
Low in mood?	78	1 (0 to 19)
Worried?	79	1 (0 to 24)

Table 6.4 Smartphone application questions and response data for participants in the intervention group (n = 29)

Questions required a 'yes/no' response. Data are presented as the total number (N^{o}) of times each question had a 'yes' response, and the median (range) per participant.

6.2.3.4 Other secondary outcomes

The mean (SD) SUS score for the app at 6 and 12 months was 87 (13) and 89 (13), respectively. There was no clear evidence that use of the app affected spirometry (Table 6.5), nutritional status (Table 6.5), HRQoL (Table 6.6), or feelings of anxiety and depression (Table 6.7).

There was no clear evidence that use of the app affected treatment adherence (Table 6.8). The most common reasons reported by all participants for 'What is getting in the way?' of treatment were "I couldn't find the time" (34 [57%]), "I forgot to do it" (34 [57%]), "I don't think I need it" (18 [30%]), "I don't want to do it" (12 [20%]) and "I experience side effects" (9 [15%]).

There was no clear evidence that use of the app affected absenteeism and presenteeism (

Table 6.9). Most participants (87%) were engaged in employment on at least one occasion throughout the 12-month study period. The proportion of participants engaged in employment on at least one occasion was similar between the intervention (90%) and control groups (84%).
	Intervention		Control		
	Prior to randomisation	12 months following randomisation	Prior to randomisation	12 months following randomisation	Mean diff. in change (95% CI)
	(n = 29)	(n = 28)	(n = 31)	(n = 30)	
FEV ₁ , L	2.07 (0.85)	2.03 (0.88)	2.41 (0.82)	2.28 (0.86)	0.08 (-0.08 to 0.15)
$ppFEV_1$	58 (18)	56 (19)	61 (17)	58 (18)	1 (-2 to 4)
FVC, L	3.06 (1.04)	3.07 (1.11)	3.68 (1.03)	3.50 (1.11)	0.17 (-0.05 to 0.27)
ppFVC	71 (16)	71 (16)	77 (14)	74 (15)	3 (-1 to 6)
BMI, kg/m ²	22.4 (5.0)	21.9 (6.3)	23.4 (2.9)	24.0 (2.7)	-1.1 (-2.4 to 0.2)

Table 6.5 Spirometry and body mass index measured prior to randomisation and 12 months following randomisation

Data are presented as mean (SD). Analyses performed using general linear mixed models. BMI = body mass index, CI = confidence interval, diff. = difference, FEV_1 = forced expiratory volume in 1 second, FVC = forced vital capacity, pp = percentage predicted. The Global Lung Function Initiative 2012 equations (151) were used to calculate percentage predicted ppFEV₁ and ppFVC.

	Intervention		Control		
	Prior to randomisation (n = 29)	12 months following randomisation (n = 28)	Prior to randomisation (n = 31)	12 months following randomisation (n = 30)	Mean diff. in change (95% CI)
Physical functioning	70 (26)	73 (22)	70 (26)	74 (25)	-3 (-11 to 5)
Vitality	55 (16)	55 (17)	58 (17)	52 (20)	6 (-3 to 15)
Emotional functioning	79 (14)	82 (14)	77 (17)	79 (15)	1 (-6 to 7)
Eating	92 (13)	92 (16)	93 (12)	96 (11)	-3 (-8 to 2)
Treatment burden	56 (22)	58 (19)	57 (16)	58 (18)	1 (-8 to 9)
Health perceptions	62 (19)	64 (18)	63 (20)	63 (19)	2 (-7 to 9)
Social	75 (15)	71 (16)	65 (16)	67 (18)	-6 (-13 to 1)
Body Image	66 (29)	67 (29)	75 (24)	80 (24)	-4 (-12 to 4)
Role	83 (14)	84 (17)	79 (20)	81 (18)	-1 (-9 to 6)
Weight	67 (41)	62 (42)	82 (28)	89 (20)	-12 (-22 to 3)
Respiratory symptoms	67 (17)	65 (17)	71 (16)	65 (18)	4 (-5 to 11)
Digestive symptoms	82 (19)	85 (15)	86 (15)	88 (12)	1 (-8 to 9)

 Table 6.6 Participant responses to the Cystic Fibrosis Questionnaire-Revised measured prior to randomisation and 12 months following randomisation

Data are presented as mean (SD). An increase in score represents improvement, with a maximum score of 100. Analyses performed using general linear mixed models. CI = confidence interval, CFQ-R = Cystic Fibrosis Questionnaire - Revised, Diff. = difference.

	Intervention		Control		
	Prior to randomisation (n = 29)	12 months following randomisation (n = 28)	Prior to randomisation (n = 31)	12 months following randomisation (n = 30)	Mean diff. in change (95% CI)
Feelings of anxiety	5 (6)	3.5 (6)	4 (5)	4.5 (4)	2 (-2 to 1)
Feelings of depression	2 (4)	2 (3)	3 (4)	2 (4)	-1 (-1 to 1)

 Table 6.7 Participant responses to the Hospital Anxiety and Depression Scale prior to randomisation and 12 months following randomisation

Data are presented as mean (SD). A decrease in score represents improvement, with a maximum score of 21. Analyses performed using general linear mixed models. BMI = body mass index. CI = confidence interval, diff. = difference.

	Intervention $(n = 26)$		Control $(n = 30)$		
	Prior to randomisation	12 months following randomisation	Prior to randomisation	12 months following randomisation	Mean diff. in change (95% CI)
Treatments/week (n)	89 (25 to 132)	85 (17 to 133)	85 (42 to 161)	77 (23 to 133)	4 (-9 to 15)
Total time/week (min)	1582 (0 to 6290)	1858 (20 to 4340)	1640 (260 to 7765)	1332 (60 to 3465)	584 (-283 to 1446)

Table 6.8 Treatment Adherence Questionnaire – Cystic Fibrosis scores for participants with complete data prior to and 12 months following randomisation

Data are presented as number (range) or minutes (range). Treatments/week represents the total number of treatments completed over a one-week period. Total time/week represents the number of treatments per week multiplied by the time taken to complete each treatment according to participants' responses. Analyses performed using general linear mixed models.

	Intervention $(n = 2)$	26)	Control $(n = 26)$		
	Prior to randomisation	12 months following randomisation	Prior to randomisation	12 months following randomisation	Mean diff. in change (95% CI)
Relative absenteeism	0.0 (0.8)	0.0 (0.8)	0.1 (0.3)	0.1 (0.2)	0.0 (-0.5 to 0.4)
Relative presenteeism	1.1 (0.4)	1.0 (0.1)	1.1 (0.4)	1.1 (0.2)	-0.1 (-0.4 to 0.0)

Table 6.9 Absenteeism and presenteeism scores for participants with complete data prior to and 12 months following randomisation

Data are presented as mean (SD). Relative absenteeism is the percentage of time absent from work. A maximum score of 1 indicates 100% absenteeism; a score of 0 indicates no absenteeism. Negative scores indicate the hours worked exceeded employer's expectations. Relative presenteeism is self-rated work performance compared to others in similar employment. A score of 1 represents equal performance, 0 represents worst performance, and a maximum score of 2 represents performance at least twice as good as colleagues. Analyses performed using general linear mixed models. CI = confidence interval, diff. = mean difference.

Chapter 6: Study 4

6.3 Discussion

For adults with CF, a novel app was developed for reporting symptoms suggestive of a respiratory exacerbation its effect on the number of respiratory exacerbations requiring IV antibiotics and other health outcomes was investigated. Adherence to the weekly use of the app by participants in the intervention group was much greater than reported in previous telehealth studies in people with CF (21, 22, 24, 25, 96), and system usability was maintained above 85% at both 6 and 12 months. The 95% CI for between-group differences in the number of respiratory exacerbations requiring IV antibiotics suggested there was no effect of the intervention on this outcome. However, respiratory exacerbations requiring oral or IV antibiotics were detected earlier, and there was an increased prescription of oral antibiotics in the intervention group. There was no clear effect of the use of the app on other aspects of HCU, spirometry, nutritional status, HRQoL or feelings of anxiety and depression, treatment adherence or absenteeism and presenteeism.

The level of adherence (77%) to the use of the app was somewhat better than earlier work that has investigated home monitoring interventions for people with CF, in which adherence has ranged from 10% to 67% (21, 22, 24, 25, 96). Any intervention that is inconvenient and time consuming will likely have poorer adherence, given it is in addition to the already large treatment burden that people with CF experience (71). Adherence in the present study did vary though, with 14 (48%) participants using the app less than 80% of the required weeks. Sub-group analyses suggested that those participants in the intervention group who were more adherent (\geq 80%) to the weekly use of the app experienced an increase in oral antibiotic use and a concurrent reduction in IV antibiotic use. Further investigation is needed to determine the characteristics of participants likely to have better adherence to apps and other monitoring interventions so that prospective studies can determine the effectiveness in this sub-group.

The high usability of this app is further supported by the low attrition from the intervention group (3%). This is considerably less than earlier studies of telehealth interventions in a CF population, in which attrition ranged from 24% to 63% (21, 22, 24, 25). The level of adherence

and low attrition in this study is likely attributed to the use of smartphones, a small and portable technology already incorporated into the lifestyle of the participants, and the once weekly reporting requirement. The usability scores at both 6 and 12 months reflected those observed in the previous usability study for this app (Chapter 5, Study 3, Part A). High usability was maintained through to 12 months, indicating that the app is a potential longer-term solution within routine clinical care.

The earlier detection of respiratory exacerbations and increased provision of oral antibiotics in the intervention group did not translate into an important reduction in the number of courses of IV antibiotics. The greater number of courses of IV antibiotics prescribed in the intervention group was consistent with the number of courses seen in the preceding 12 months, and was adjusted for in the analysis. This does raise the question as to whether the early provision of oral antibiotic treatment alone can translate into a reduction in the need for IV antibiotics, and also highlights the limited treatment options available for clinicians when treating respiratory exacerbations. Likewise, as noted for the difference in IV antibiotics, these data demonstrate no differences between groups in secondary outcomes such as aspects of healthcare costs, spirometry, nutritional status, HRQoL, feelings of anxiety and depression, treatment adherence and absenteeism and presenteeism. Despite this finding, it is encouraging that the app provided a means of earlier detection of respiratory exacerbations, and if implemented over a longer surveillance period (e.g. 3 years), this may provide clearer answers regarding the potential longer-term benefits such as slowing the magnitude of decline in lung function.

The most common respiratory symptoms reported in the present study were cough, sputum, fatigue, shortness of breath and chest tightness, mirroring those previously reported as being triggers for seeking treatment for a respiratory exacerbation by children and adults with CF (95). Most of the symptoms included in the app questions were adapted from Fuchs exacerbation scale (18), but of note the symptom 'chest tightness' was included as it has been previously reported by people with CF as being commonly associated with respiratory exacerbations (95). As there is no universal agreement on how to diagnose a respiratory exacerbation in this

population (82, 83, 86), the pragmatic decision was made to mirror clinical practice and rely on symptom reporting alone as the first line of detection.

It was surprising to see that, when compared with the 12 months preceding enrolment, there was a reduction in the number of courses of IV antibiotics in both groups during the 12-month study period. While this may simply reflect a spurious finding, it is likely that participating in a study of this kind led to increased contact with the CF centre and potentially offered greater opportunity for people to collect prescriptions from the hospital pharmacy. Participants in both groups attended a higher number of CF clinics than are currently recommended (i.e. 4 per year) (4); therefore involvement in the study seems to have improved access to care for all participants. The single centre design of this study is a limitation in the ability to generalise the results to other CF centres, and a larger multicentre study is now warranted.

Monitoring with smartphone spirometry was not included because of the increased burden on the participant, which would most likely have compromised adherence. Future studies planning to add additional technology to supplement data obtained from the app must carefully consider the benefit vs. the risk to adherence. The use of other readily available smartphone technologies, such as measurements of daily steps may be a useful addition as physical activity may decline during respiratory exacerbations, and these data can be collected with minimal additional burden to the user.

It is also possible that once weekly reporting of symptoms was insufficient and that increased frequency may be helpful. That is, with more frequent reporting of symptoms, respiratory exacerbations may be detected earlier than in the present study. Given the high level of adherence observed with weekly reporting, there is scope to increase this to two or three times per week; however, the impact of increasing the frequency of reporting on adherence will need careful evaluation.

In summary, this is the first app used by adults with CF to report their respiratory symptoms directly to the CF team. The high level of adherence is attributed to the low reporting burden and use of technology already owned by participants. Provision of the app facilitated the earlier

detection of respiratory exacerbations and treatment in the form of oral antibiotics. While the data did not demonstrate an effect on the number of courses of IV antibiotics for the participants over a 12-month period, this study supports the continued development and investigation of similar apps and technologies used to identify respiratory exacerbations, with the aim of improving long-term health outcomes.

CHAPTER 7

SUMMARY AND CONCLUSIONS

This chapter summarises the findings and draws conclusions from the studies conducted within this programme of research. The summary and conclusions are presented in three sections, corresponding with those presented in the Introduction (Chapter 1). Finally, future directions and the implications for further research in this field are discussed.

7.1 Telehealth for adults with cystic fibrosis living in rural and remote Western Australia

In this study, the difference in magnitude of decline in forced expiratory volume in 1 second (FEV_1) was investigated between adults with cystic fibrosis (CF) living in rural and remote Western Australia (WA) and their metropolitan counterparts. An audit was conducted over a 3-year period for all eligible adults with CF living in WA (n = 156), with the magnitude of decline calculated based on spirometry collected at CF clinic attended during this period.

No clear effect on the magnitude of decline of spirometry measures was demonstrated, however this was most likely explained by the modest sample size (n = 156), especially for those living in rural and remote WA (n = 26). Despite this, the results of this study highlighted the poor access to care and clinic attendance amongst adults with CF in rural and remote WA, with the mean (range) number of CF clinics attended per participant over 3 years (8 [2 to 24]) falling well below the Australian CF Standards of Care guideline of a minimum of four CF clinics per year (i.e. 12 over 3 years) (4). In contrast, for those living in the Perth metropolitan area, this recommendation was exceeded with the number of CF clinics attended per participant over 3 years (21 [3 to 58]). The reduced level of surveillance observed for those in rural and remote areas may lead to the CF team missing opportunities to identify respiratory exacerbations and provide essential antibiotic treatment. This provides a sound argument for the need for improved access to care and disease monitoring via telehealth in adults with CF living in rural and remote WA.

Telehealth clinics (via videoconferencing) were implemented as part of the routine outpatient care for people with CF living in rural and remote WA, and uptake and satisfaction was evaluated. Uptake was high, with CF clinic attendance increasing during the study period (median = 5) to meet the recommended standards of care. Satisfaction was also high amongst participants, and most expressed a desire to continue using telehealth as part of routine clinical care. This is not surprising, given the reduced travel burden and time away from home associated with utilising telehealth in this manner.

Telehealth facilitated improved surveillance of participants, resulting in increased antibiotic use and hospital admissions. Rather than telehealth being of detriment to the participants, these data reflected the increased ability of the CF team to detect respiratory exacerbations. The improved detection of respiratory exacerbations may lead to improved longer-term health outcomes (FEV₁, nutritional status and health-related quality of life [HRQoL]) if maintained over a much longer timeframe (i.e. 5 to 10 years).

People with CF have been previously found to be willing to utilise telehealth technology (142), which is supported by the results of this study. Nevertheless, clinicians need to recognise that services provided by telehealth are unable to replicate all aspects of an in-person clinic. First, telehealth removes the ability of the clinician to perform physical assessments such as palpation and auscultation. Second, participants in rural and remote areas have historically had limited access to on-site specialist secondary services (e.g. intravascular access, endocrinology, immunology and hepatology), and may also have no access to imaging equipment such as computed tomography and magnetic resonance imaging. Third, a service that can provide spirometry may not be available in many areas of low population density throughout the world, potentially limiting the information available for clinical decision making.

Telehealth has a promising role in the delivery of care to people with CF in rural and remote areas. By removing the need to travel to the CF centre for all CF clinics, people with CF can reduce their travel, time and financial burden. If utilised by the CF team in a manner that recognises both the strengths and limitations of this type of system, the use of telehealth in the provision of CF clinics will be beneficial to both people with CF and the healthcare system.

7.2 The working profile of adults with cystic fibrosis living in rural and remote Western Australia

This study was an opportunistic sub-analysis of data collected from those participants who enrolled in the study investigating the provision of CF clinics via telehealth. The results demonstrated that most adults with CF living in rural and remote WA engaged in paid work during the 12-month study period.

The impact of CF on work attendance (absenteeism) and performance (presenteeism) was variable. While this may be due to the variable impact of CF on the participants in general, the results may also have been affected by the small sample size and variability in employment undertaken (e.g. administration, retail, information technology, and healthcare). It is encouraging that, on average, participants felt that they performed as well as their colleagues when at work. It would be useful to know further details about the impact of CF on the participants who reported high absenteeism and low presenteeism.

The use of absenteeism and presenteeism is a novel outcome measure in CF, and its measurement can provide a useful insight into the disease-related impact on work attendance and performance. If appropriate, future studies should consider capturing these data to measure the impact of interventions on the ability of people with CF to engage in employment, especially considering the increasing proportion of adults living with CF (31, 33).

7.3 A smartphone application for reporting respiratory symptoms in adults with cystic fibrosis

Studies that have explored the effect of home monitoring interventions in CF have been plagued by sub-optimal adherence to the monitoring system. This has been a result of the high reporting burden (up to daily), and the monitoring equipment being large and not mobile outside of the home. The smartphone application (app) described in this study overcame this limitation, as it was small and portable, required only weekly use, and integrated easily with the lifestyle of adults with CF. The system did not include spirometry which may improve the accuracy of diagnosis of respiratory exacerbations, however this would have increased costs, and increased the reporting burden amongst participants. The inclusion of spirometry may potentially compromise adherence in future versions of the system.

The usability of the app was rated as high by participants, and importantly, the symptom responses from the app were interpreted in a similar way by clinicians experienced in the process of clinical decision making within CF care.

The use of an app for the purpose of reporting symptoms suits the demographic profile of the adult CF population. A quick-to-use, convenient and low-burden tool is also ideal given the heavy treatment and lifestyle burden that people with CF already experience. While the usability study was of short duration with a small sample, the results provided encouragement for further investigation of the impact of the app on health outcomes. Specifically, it was important to ascertain whether using the app can improve the detection of respiratory exacerbations and reduce the time taken to provide antibiotic treatment.

The randomised controlled trial (RCT) undertaken was the first to examine the impact of an app used by adults with CF to report respiratory symptoms to the CF team. The aim was to improve the detection of respiratory exacerbations, facilitating the earlier provision of treatment, which in turn may reduce the severity of exacerbations. Despite good adherence (77%) to the weekly use of the app, and the faster detection of respiratory exacerbations requiring oral or IV antibiotics for those in the intervention group, there was no effect of the intervention on the

number of exacerbations requiring IV antibiotics. The question needs to be raised as to whether the provision of oral antibiotic treatment alone as the first line of respiratory exacerbation treatment can translate into a reduction in the need for IV antibiotics and improved health outcomes in the longer-term. There was also no clear effect of the app on other outcomes including other aspects of HCU, spirometry, nutritional status, HRQoL or feelings of anxiety and depression, treatment adherence or absenteeism and presenteeism.

The level of adherence observed with use of the app was somewhat higher than that seen in earlier work in this area, and this matched the high levels of usability demonstrated throughout the entire study period. Further, attrition was extremely low in the intervention group (3%), highlighting that this system was low-burden and acceptable to the adults with CF using it. The app questions took approximately two minutes to complete, and therefore it is unlikely that the perceived reporting burden associated with this would increase in the future. The optimal frequency of reporting has not been determined, and twice per week may be warranted, however care should be taken not to compromise adherence by increasing the required frequency.

The app questions were formulated from respiratory exacerbation criteria used in a previous clinical trial (18) in combination with clinician and patient reported outcomes (95). The definition of a respiratory exacerbation is a contentious issue and not universally agreed (82, 83, 86, 87), and the pragmatic approach taken with this app, and criteria used to define a respiratory exacerbation, mirrors clinical practice around the world (84, 86).

This is the first app designed for use by adults with CF to report their respiratory symptoms directly to the CF team. While the data did not demonstrate a clear effect on the number of courses of IV antibiotics for the participants over a 12-month period, this study supports the continued development and investigation of similar apps and technologies used to identify respiratory exacerbations, with the aim of improving long-term health outcomes.

7.4 Limitations

The studies comprising this work have several limitations that could be addressed for future research efforts. The sample sizes are generally small and the research is single centre in nature. In the present work this was difficult to address given the pragmatic approach and unique CF model of care in Western Australia. The telehealth interventions investigated had better adherence than previously seen in CF studies, and show promise technologies that can be incorporated into routine clinical care; however, they need to be explored in much larger numbers of participants and in different CF clinics, countries and cultures. The level of adherence to the interventions may have been improved by the fact this was research, and that the investigators had a prior relationship with the participants. Longer term studies conducted over several years will provide further clarification on the adherence to, and effectiveness of these interventions in a setting that reflects real life more closely.

The detection of respiratory exacerbations is CF is a contentious topic given the lack of consensus around the definition of an exacerbation. Given that the identification and treatment of an exacerbation in CF is clinician driven, this will influence the interpretation of outcome sin both clinical practice and research. It is therefore important to continue this work in different CF clinics in different areas of the world, to further refine the system so it can provide benefit to all people with CF, and not just those in WA.

7.5 Future directions

This programme of research has demonstrated that telehealth can improve access to care for adults with CF living in rural and remote areas via videoconferencing, and for all adults with CF via an app for reporting respiratory symptoms. The widespread adoption of such technologies in CF care is still in a nascent period, and ongoing evaluation is required to examine their impact on health outcomes, and the cost benefit (i.e. staffing, technology costs, impact on HCU) over longer periods of time. The enthusiasm that can be generated with newer technologies also needs to be met with a sufficient evidence base prior to any widespread adoption. The research presented in this thesis adds to this evidence base with pragmatic solutions to the difficulties experienced with access to care and detecting exacerbations in CF.

One of the major challenges is the difficulty faced in evaluating technologies that are rapidly evolving, as conducting rigorous research using traditional methods (i.e. RCTs) can often take many years and by the time the research is completed the technology may well have been superseded. Despite this, the growing number of technologies and devices available that have the potential to be integrated into CF care warrants enough interest to continue to evaluate their role in the management of this disease. It will be important to not only develop research programs with sufficient numbers of participants and surveillance periods, but to also ensure that the methodologies are appropriate for the technologies being investigated. Considering the role of education and behaviour change as a complimentary mode of treatment to home monitoring technology may also improve the impact of these types of services. Further, utilising similar technologies in other chronic disease groups, such as asthma and chronic obstructive pulmonary disease, will strengthen support for their use in clinical care. This work supports the integration of modern technologies such as videoconferencing and apps with traditional care by CF teams when planning future service delivery.

REFERENCES

1. Australian Institute of Health and Welfare. Australia's Health 2014. Canberra (Australia): Australian Institute for Health and Welfare; 2014 [cited 2015, July 15]. Report No.: AUS 178. Available from: <u>https://www.aihw.gov.au/getmedia/d2946c3e-9b94-413c-898c-aa5219903b8c/16507.pdf.aspx?inline=true</u>

2. Bywood P KR, Lunnay B. Disparities in primary health care utilisation: Who are the disadvantaged groups? How are they disadvantaged? What interventions work? Adelaide: Primary Health Care Research and Information Service; 2011 [cited 2015, July 15]. Available from:

https://dspace.flinders.edu.au/xmlui/bitstream/handle/2328/26598/PIR%20June%2011%20Full.pdf?sequence=1&isAllowed=y

3. Elborn JS, Bell SC, Madge SL, Burgel PR, Castellani C, Conway S, et al. Report of the European Respiratory Society/European Cystic Fibrosis Society task force on the care of adults with cystic fibrosis. Eur Respir J. 2016;47(2):420-8.

4. Bell SC. Cystic Fibrosis standards of care, Australia 2008. Sydney (Australia): Cystic Fibrosis Australia; 2008 [cited 2012, November 12]. Available from: <u>https://www.thoracic.org.au/journal-</u> publishing/command/download_file/id/20/filename/CF_standardsofcare_Australia_2008.pdf

5. Lebecque P, Leonard A, De Boeck K, De Baets F, Malfroot A, Casimir G, et al. Early referral to cystic fibrosis specialist centre impacts on respiratory outcome. J Cyst Fibros. 2009;8(1):26-30.

6. Mahadeva R, Webb K, Westerbeek RC, Carroll NR, Dodd ME, Bilton D, et al. Clinical outcome in relation to care in centres specialising in cystic fibrosis: cross sectional study. BMJ. 1998;316(7147):1771-5.

7. Hogg M, Braithwaite M, Bailey M, Kotsimbos T, Wilson JW. Work disability in adults with cystic fibrosis and its relationship to quality of life. J Cyst Fibros. 2007;6(3):223-7.

8. Havermans T, Colpaert K, Vanharen L, Dupont LJ. Health related quality of life in cystic fibrosis: To work or not to work? J Cyst Fibros. 2009;8(3):218-23.

9. Targett K, Bourke S, Nash E, Murphy E, Ayres J, Devereux G. Employment in adults with cystic fibrosis. Occup Med (Lond). 2014;64(2):87-94.

10. Frangolias DD, Holloway CL, Vedal S, Wilcox PG. Role of exercise and lung function in predicting work status in cystic fibrosis. Am J Respir Crit Care Med. 2003;167(2):150-7.

11. Sanders DB, Bittner RC, Rosenfeld M, Redding GJ, Goss CH. Pulmonary exacerbations are associated with subsequent FEV_1 decline in both adults and children with cystic fibrosis. Pediatr Pulmonol. 2011;46(4):393-400.

12. Ouyang L, Grosse SD, Amendah DD, Schechter MS. Healthcare expenditures for privately insured people with cystic fibrosis. Pediatr Pulmonol. 2009;44(10):989-96.

13. Britto MT, Kotagal UR, Hornung RW, Atherton HD, Tsevat J, Wilmott RW. Impact of recent pulmonary exacerbations on quality of life in patients with cystic fibrosis. Chest. 2002;121(1):64-72.

14. Flume PA, Suthoff ED, Kosinski M, Marigowda G, Quittner AL. Measuring recovery in health-related quality of life during and after pulmonary exacerbations in patients with cystic fibrosis. J Cyst Fibros. 2019;18(5):737-42.

15. Ward N, White D, Rowe H, Stiller K, Sullivan T. Physical activity levels of patients with cystic fibrosis hospitalised with an acute respiratory exacerbation. Respir Med. 2013;107(7):1014-20.

16. Riekert KA, Bartlett SJ, Boyle MP, Krishnan JA, Rand CS. The association between depression, lung function, and health-related quality of life among adults with cystic fibrosis. Chest. 2007;132(1):231-7.

17. de Boer K, Vandemheen KL, Tullis E, Doucette S, Fergusson D, Freitag A, et al. Exacerbation frequency and clinical outcomes in adult patients with cystic fibrosis. Thorax. 2011;66(8):680-5.

18. Fuchs HJ, Borowitz DS, Christiansen DH, Morris EM, Nash ML, Ramsey BW, et al. Effect of aerosolized recombinant human DNase on exacerbations of respiratory symptoms and on pulmonary function in patients with cystic fibrosis. The Pulmozyme Study Group. N Engl J Med. 1994;331(10):637-42.

19. Sanders DB, Bittner RC, Rosenfeld M, Hoffman LR, Redding GJ, Goss CH. Failure to recover to baseline pulmonary function after cystic fibrosis pulmonary exacerbation. Am J Respir Crit Care Med. 2010;182(5):627-32.

20. Zemanick ET, Harris JK, Conway S, Konstan MW, Marshall B, Quittner AL, et al. Measuring and improving respiratory outcomes in cystic fibrosis lung disease: opportunities and challenges to therapy. J Cyst Fibros. 2010;9(1):1-16.

21. Bella S, Murgia F, Tozzi AE, Cotognini C, Lucidi V. Five years of telemedicine in cystic fibrosis disease. Clin Ter. 2009;160(6):457-60.

22. Jarad NA, Sund ZM. Telemonitoring in chronic obstructive airway disease and adult patients with cystic fibrosis. J Telemed Telecare. 2011;17(3):127-32.

23. Lechtzin N, West N, Allgood S, Wilhelm E, Khan U, Mayer-Hamblett N, et al. Rationale and design of a randomized trial of home electronic symptom and lung function monitoring to detect cystic fibrosis pulmonary exacerbations: the Early Intervention in Cystic fibrosis Exacerbation (eICE) trial. Contemp Clin Trials. 2013;36(2):460-9.

24. Murgia F, Bianciardi F, Solvoll T, Tagliente I, Bella F, Carestia A, et al. Telemedicine home program in patients with cystic fibrosis: Results after 10 years. Clin Ter. 2015;166(6):e384-8.

25. Sarfaraz S, Sund Z, Jarad N. Real-time, once-daily monitoring of symptoms and FEV in cystic fibrosis patients--a feasibility study using a novel device. Clin Respir J. 2010;4(2):74-82.

26. Wilkinson OM, Duncan-Skingle F, Pryor JA, Hodson ME. A feasibility study of home telemedicine for patients with cystic fibrosis awaiting transplantation. J Telemed Telecare. 2008;14(4):182-5.

27. Cox NS, Alison JA, Button BM, Wilson JW, Holland AE. Assessing exercise capacity using telehealth: a feasibility study in adults with cystic fibrosis. Respir Care. 2013;58(2):286-90.

28. Tomlinson OW, Shelley J, Trott J, Bowhay B, Chauhan R, Sheldon CD. The feasibility of online video calling to engage patients with cystic fibrosis in exercise training. J Telemed Telecare. 2019:1357633X19828630.

29. Cox NS, Alison JA, Button BM, Wilson JW, Holland AE. Feasibility and acceptability of an internet-based program to promote physical activity in adults with cystic fibrosis. Respir Care. 2015;60(3):422-9.

30. Ratjen F, Doring G. Cystic fibrosis. Lancet. 2003;361(9358):681-9.

31. Ruseckaite R, Ahern S, Ranger T, Dean J, Gardam M, Bell S, et al. The Australian Cystic Fibrosis Data Registry Annual Report 2017. Melbourne: Monash University Department of Epidemiology and Preventative Medicine; 2019. Available from: https://www.cysticfibrosis.org.au/getmedia/24e94d66-29fa-4e3f-8e65-21ee24ed2e5a/ACFDR-2017-Annual-Report_highres_singles.pdf.aspx.

32. Australian Bureau of Statistics. Population Clock. Canberra (Australia): Australian Bureau of Statistics; 2020 [cited 2020, March 27]. Available from: https://www.abs.gov.au/AUSSTATS/abs@.nsf/Web+Pages/Population+Clock?opendocument& ref=HPKI.

33. Bell SC, Mall MA, Gutierrez H, Macek M, Madge S, Davies JC, et al. The future of cystic fibrosis care: a global perspective. Lancet Respir Med. 2020;8(1):65-124.

34. Stephenson AL, Sykes J, Stanojevic S, Quon BS, Marshall BC, Petren K, et al. Survival comparison of patients with cystic fibrosis in Canada and the United States: A population-based cohort study. Ann Intern Med. 2017;166(8):537-46.

35. Grasemann H. CFTR modulator therapy for cystic fibrosis. N Engl J Med. 2017;377(21):2085-8.

36. Chassagnon G, Hubert D, Fajac I, Burgel PR, Revel MP, investigators. Long-term computed tomographic changes in cystic fibrosis patients treated with ivacaftor. Eur Respir J. 2016;48(1):249-52.

37. Quittner A, Suthoff E, Rendas-Baum R, Bayliss MS, Sermet-Gaudelus I, Castiglione B, et al. Effect of ivacaftor treatment in patients with cystic fibrosis and the G551D-CFTR mutation: patient-reported outcomes in the STRIVE randomized, controlled trial. Health Qual Life Outcomes. 2015;13:93.

38. Middleton PG, Mall MA, Drevinek P, Lands LC, McKone EF, Polineni D, et al. Elexacaftor-Tezacaftor-Ivacaftor for cystic fibrosis with a single Phe508del allele. N Engl J Med. 2019;381(19):1809-19.

39. Ramsey BW, Davies J, McElvaney NG, Tullis E, Bell SC, Drevinek P, et al. A CFTR potentiator in patients with cystic fibrosis and the G551D mutation. N Engl J Med. 2011;365(18):1663-72.

40. van Gool K, Norman R, Delatycki MB, Hall J, Massie J. Understanding the costs of care for cystic fibrosis: an analysis by age and health state. Value Health. 2013;16(2):345-55.

41. Hassan M, Bonafede MM, Limone BL, Hodgkins P, Sawicki GS. The burden of cystic fibrosis in the Medicaid population. Clinicoecon Outcomes Res. 2018;10:423-31.

42. Pharmaceutical Benefits Scheme. Ivacaftor. Canberra (Australia): Australian Government Department of Health; 2020 [cited 2020, January 14]. Available from: https://www.pbs.gov.au/medicine/item/10170G-10175M-11097C-11098D-11105L-11109Q.

43. Ratjen F, Bell SC, Rowe SM, Goss CH, Quittner AL, Bush A. Cystic fibrosis. Nat Rev Dis Primers. 2015;1:15010.

44. Flume PA. Pulmonary complications of cystic fibrosis. Respir Care. 2009;54(5):618-27.

45. Mitchell I, Nakielna E, Tullis E, Adair C. Cystic fibrosis. End-stage care in Canada. Chest. 2000;118(1):80-4.

46. Chin M, Aaron SD, Bell SC. The treatment of the pulmonary and extrapulmonary manifestations of cystic fibrosis. Presse Med. 2017;46(6 Pt 2):e139-e64.

47. Duff AJ, Abbott J, Cowperthwaite C, Sumner C, Hurley MA, Quittner A, et al. Depression and anxiety in adolescents and adults with cystic fibrosis in the UK: a cross-sectional study. J Cyst Fibros. 2014;13(6):745-53.

48. Backstrom-Eriksson L, Sorjonen K, Bergsten-Brucefors A, Hjelte L, Melin B. Anxiety and depression in adults with cystic fibrosis: a comparison between patients and the general population in Sweden and three other European countries. BMC Pulm Med. 2015;15:121.

49. Cruz I, Marciel KK, Quittner AL, Schechter MS. Anxiety and depression in cystic fibrosis. Semin Respir Crit Care Med. 2009;30(5):569-78.

50. Quittner AL, Barker DH, Snell C, Grimley ME, Marciel K, Cruz I. Prevalence and impact of depression in cystic fibrosis. Curr Opin Pulm Med. 2008;14(6):582-8.

51. Stanbrook MB, Corey M, Tullis DE. The repeatability of forced expiratory volume measurements in adults with cystic fibrosis. Chest. 2004;125(1):150-5.

52. Davies JC, Alton EW. Monitoring respiratory disease severity in cystic fibrosis. Respir Care. 2009;54(5):606-17.

53. Kerem E, Viviani L, Zolin A, Macneill S, Hatziagorou E, Ellemunter H, et al. Factors associated with FEV_1 decline in cystic fibrosis: analysis of the data of the ECFS Patient Registry. Eur Respir J. 2014;43(1):125-33.

54. Waters V, Stanojevic S, Atenafu EG, Lu A, Yau Y, Tullis E, et al. Effect of pulmonary exacerbations on long-term lung function decline in cystic fibrosis. Eur Respir J. 2012;40(1):61-6.

55. Harness-Brumley CL, Elliott AC, Rosenbluth DB, Raghavan D, Jain R. Gender differences in outcomes of patients with cystic fibrosis. J Womens Health. 2014;23(12):1012-20.

56. Sanders DB, Li Z, Laxova A, Rock MJ, Levy H, Collins J, et al. Risk factors for the progression of cystic fibrosis lung disease throughout childhood. Ann Am Thorac Soc. 2014;11(1):63-72.

57. Bakker EM, Borsboom GJ, van der Wiel-Kooij EC, Caudri D, Rosenfeld M, Tiddens HA. Small airway involvement in cystic fibrosis lung disease: routine spirometry as an early and sensitive marker. Pediatr Pulmonol. 2013;48(11):1081-8.

58. Gustafsson PM, De Jong PA, Tiddens HA, Lindblad A. Multiple-breath inert gas washout and spirometry versus structural lung disease in cystic fibrosis. Thorax. 2008;63(2):129-34.

59. Logan PM, O'Laoide RM, Mulherin D, O'Mahony S, FitzGerald MX, Masterson JB. High resolution computed tomography in cystic fibrosis: correlation with pulmonary function and assessment of prognostic value. Ir J Med Sci. 1996;165(1):27-31.

60. Ramsey KA, Rosenow T, Turkovic L, Skoric B, Banton G, Adams AM, et al. Lung clearance index and structural lung disease on computed tomography in early cystic fibrosis. Am J Respir Crit Care Med. 2016;193(1):60-7.

61. Sanders DB, Li Z, Brody AS. Chest computed tomography predicts the frequency of pulmonary exacerbations in children with cystic fibrosis. Ann Am Thorac Soc. 2015;12(1):64-9.

62. Kilcoyne A, Lavelle LP, McCarthy CJ, McEvoy SH, Fleming H, Gallagher A, et al. Chest CT abnormalities and quality of life: relationship in adult cystic fibrosis. Ann Transl Med. 2016;4(5):87.

63. Loeve M, Hop WC, de Bruijne M, van Hal PT, Robinson P, Aitken ML, et al. Chest computed tomography scores are predictive of survival in patients with cystic fibrosis awaiting lung transplantation. Am J Respir Crit Care Med. 2012;185(10):1096-103.

64. Hebestreit H, Hulzebos EHJ, Schneiderman JE, Karila C, Boas SR, Kriemler S, et al. Cardiopulmonary exercise testing provides additional prognostic information in cystic fibrosis. Am J Respir Crit Care Med. 2019;199(8):987-95.

65. Kuo W, Ciet P, Tiddens HA, Zhang W, Guillerman RP, van Straten M. Monitoring cystic fibrosis lung disease by computed tomography. Radiation risk in perspective. Am J Respir Crit Care Med. 2014;189(11):1328-36.

66. O'Neill K, Tunney MM, Johnston E, Rowan S, Downey DG, Rendall J, et al. Lung clearance index in adults and children with cystic fibrosis. Chest. 2016;150(6):1323-32.

67. Conway S, Balfour-Lynn IM, De Rijcke K, Drevinek P, Foweraker J, Havermans T, et al. European Cystic Fibrosis Society Standards of Care: Framework for the Cystic Fibrosis Centre. J Cyst Fibros. 2014;13 Suppl 1:S3-22.

68. Thomas C, Mitchell P, O'Rourke P, Wainwright C. Quality-of-life in children and adolescents with cystic fibrosis managed in both regional outreach and cystic fibrosis center settings in Queensland. J Pediatr. 2006;148(4):508-16.

69. Weber HC, Robinson PF, Saxby N, Beggs SA, Els I, Ehrlich RI. Do children with cystic fibrosis receiving outreach care have poorer clinical outcomes than those treated at a specialist cystic fibrosis centre? Aust J Rural Health. 2017;25(1):34-41.

70. Bregnballe V, Schiotz PO, Boisen KA, Pressler T, Thastum M. Barriers to adherence in adolescents and young adults with cystic fibrosis: a questionnaire study in young patients and their parents. Patient Prefer Adherence. 2011;5:507-15.

71. Sawicki GS, Sellers DE, Robinson WM. High treatment burden in adults with cystic fibrosis: challenges to disease self-management. J Cyst Fibros. 2009;8(2):91-6.

72. Burker EJ, Sedway J, Carone S. Psychological and educational factors: better predictors of work status than FEV_1 in adults with cystic fibrosis. Pediatr Pulmonol. 2004;38(5):413-8.

73. Gillen M, Lallas D, Brown C, Yelin E, Blanc P. Work disability in adults with cystic fibrosis. Am J Respir Crit Care Med. 1995;152(1):153-6.

74. Laborde-Casterot H, Donnay C, Chapron J, Burgel PR, Kanaan R, Honore I, et al. Employment and work disability in adults with cystic fibrosis. J Cyst Fibros. 2012;11(2):137-43.

75. Flume PA, Mogayzel PJ, Jr., Robinson KA, Goss CH, Rosenblatt RL, Kuhn RJ, et al. Cystic fibrosis pulmonary guidelines: treatment of pulmonary exacerbations. Am J Respir Crit Care Med. 2009;180(9):802-8.

76. Flume PA, O'Sullivan BP, Robinson KA, Goss CH, Mogayzel PJ, Jr., Willey-Courand DB, et al. Cystic fibrosis pulmonary guidelines: chronic medications for maintenance of lung health. Am J Respir Crit Care Med. 2007;176(10):957-69.

77. Kettler LJ, Sawyer SM, Winefield HR, Greville HW. Determinants of adherence in adults with cystic fibrosis. Thorax. 2002;57(5):459-64.

78. Bucks RS, Hawkins K, Skinner TC, Horn S, Seddon P, Horne R. Adherence to treatment in adolescents with cystic fibrosis: the role of illness perceptions and treatment beliefs. J Pediatr Psychol. 2009;34(8):893-902.

79. Dziuban EJ, Saab-Abazeed L, Chaudhry SR, Streetman DS, Nasr SZ. Identifying barriers to treatment adherence and related attitudinal patterns in adolescents with cystic fibrosis. Pediatr Pulmonol. 2010;45(5):450-8.

80. Myers LB, Horn SA. Adherence to chest physiotherapy in adults with cystic fibrosis. J Health Psychol. 2006;11(6):915-26.

81. Eakin MN, Riekert KA. The impact of medication adherence on lung health outcomes in cystic fibrosis. Curr Opin Pulm Med. 2013;19(6):687-91.

82. Bilton D, Canny G, Conway S, Dumcius S, Hjelte L, Proesmans M, et al. Pulmonary exacerbation: towards a definition for use in clinical trials. Report from the EuroCareCF Working Group on outcome parameters in clinical trials. J Cyst Fibros. 2011;10 Suppl 2:S79-81.

83. Dakin C, Henry RL, Field P, Morton J. Defining an exacerbation of pulmonary disease in cystic fibrosis. Pediatr Pulmonol. 2001;31(6):436-42.

84. Kraynack NC, Gothard MD, Falletta LM, McBride JT. Approach to treating cystic fibrosis pulmonary exacerbations varies widely across US CF care centers. Pediatr Pulmonol. 2011;46(9):870-81.

85. Sanders DB, Solomon GM, Beckett VV, West NE, Daines CL, Heltshe SL, et al. Standardized Treatment of Pulmonary Exacerbations (STOP) study: Observations at the initiation of intravenous antibiotics for cystic fibrosis pulmonary exacerbations. J Cyst Fibros. 2017;16(5):592-9. 86. West NE, Beckett VV, Jain R, Sanders DB, Nick JA, Heltshe SL, et al. Standardized Treatment of Pulmonary Exacerbations (STOP) study: Physician treatment practices and outcomes for individuals with cystic fibrosis with pulmonary Exacerbations. J Cyst Fibros. 2017;16(5):600-6.

87. Rabin HR, Butler SM, Wohl ME, Geller DE, Colin AA, Schidlow DV, et al. Pulmonary exacerbations in cystic fibrosis. Pediatr Pulmonol. 2004;37(5):400-6.

88. Ramsey BW, Boat TF. Outcome measures for clinical trials in cystic fibrosis. Summary of a Cystic Fibrosis Foundation consensus conference. J Pediatr. 1994;124(2):177-92.

89. Amadori A, Antonelli A, Balteri I, Schreiber A, Bugiani M, De Rose V. Recurrent exacerbations affect FEV_1 decline in adult patients with cystic fibrosis. Respir Med. 2009;103(3):407-13.

90. Solem CT, Vera-Llonch M, Liu S, Botteman M, Castiglione B. Impact of pulmonary exacerbations and lung function on generic health-related quality of life in patients with cystic fibrosis. Health Qual Life Outcomes. 2016;14:63.

91. Rubin JL, Thayer S, Watkins A, Wagener JS, Hodgkins PS, Schechter MS. Frequency and costs of pulmonary exacerbations in patients with cystic fibrosis in the United States. Curr Med Res Opin. 2017;33(4):667-74.

92. Bhatt JM. Treatment of pulmonary exacerbations in cystic fibrosis. Eur Respir Rev. 2013;22(129):205-16.

93. Elkins MR, Robinson M, Rose BR, Harbour C, Moriarty CP, Marks GB, et al. A controlled trial of long-term inhaled hypertonic saline in patients with cystic fibrosis. N Engl J Med. 2006;354(3):229-40.

94. Abbott J, Holt A, Hart A, Morton AM, MacDougall L, Pogson M, et al. What defines a pulmonary exacerbation? The perceptions of adults with cystic fibrosis. J Cyst Fibros. 2009;8(5):356-9.

95. Goss CH, Edwards TC, Ramsey BW, Aitken ML, Patrick DL. Patient-reported respiratory symptoms in cystic fibrosis. J Cyst Fibros. 2009;8(4):245-52.

96. Lechtzin N, Mayer-Hamblett N, West NE, Allgood S, Wilhelm E, Khan U, et al. Home monitoring of patients with cystic fibrosis to identify and treat acute pulmonary exacerbations. eICE Study Results. Am J Respir Crit Care Med. 2017;196(9):1144-51.

97. McCourt F, O'Neill B, Logan I, Abbott J, Plant B, McCrum-Gardner E, et al. Indicators of pulmonary exacerbation in cystic fibrosis: A Delphi survey of patients and health professionals. J Cyst Fibros. 2015;14(1):90-6.

98. Burge S, Wedzicha JA. COPD exacerbations: definitions and classifications. Eur Respir J Suppl. 2003;41:46s-53s.

99. Chung KF, Wenzel SE, Brozek JL, Bush A, Castro M, Sterk PJ, et al. International ERS/ATS guidelines on definition, evaluation and treatment of severe asthma. Eur Respir J. 2014;43(2):343-73.

100. Cystic Fibrosis Foundation. Cystic Fibrosis Foundation Patient Registry: 2017 annual data report. Bethesda, MD; 2018. [cited 2019, December 12]. Available from:

https://www.cff.org/Research/Researcher-Resources/Patient-Registry/2017-Patient-Registry-Annual-Data-Report.pdf

101. Cystic Fibrosis Trust. UK Cystic Fibrosis Registry annual data report 2018. London UK; 2019. [cited 2019, December 12]. Available from: <u>https://www.cysticfibrosis.org.uk/the-work-we-do/ukcf-registry/reporting-and-resources</u>

102. Independent Hospital Pricing Authority. National efficient price determination 2018-19.Sydney (Australia): Australian Government Department of Health; 2018 [cited 2019, December 29]. Available from:

https://www.ihpa.gov.au/sites/default/files/publications/national_efficient_price_determination_ 2018-19.pdf.

103. Sanders DB, Zhao Q, Li Z, Farrell PM. Poor recovery from cystic fibrosis pulmonary exacerbations is associated with poor long-term outcomes. Pediatr Pulmonol. 2017;52(10):1268-75.

104. Block JK, Vandemheen KL, Tullis E, Fergusson D, Doucette S, Haase D, et al. Predictors of pulmonary exacerbations in patients with cystic fibrosis infected with multiresistant bacteria. Thorax. 2006;61(11):969-74.

105. VanDevanter DR, Morris NJ, Konstan MW. IV-treated pulmonary exacerbations in the prior year: An important independent risk factor for future pulmonary exacerbation in cystic fibrosis. J Cyst Fibros. 2016;15(3):372-9.

106. Liou TG, Adler FR, Fitzsimmons SC, Cahill BC, Hibbs JR, Marshall BC. Predictive 5year survivorship model of cystic fibrosis. Am J Epidemiol. 2001;153(4):345-52.

107. Quittner AL, Sawicki GS, McMullen A, Rasouliyan L, Pasta DJ, Yegin A, et al. Erratum to: Psychometric evaluation of the Cystic Fibrosis Questionnaire-Revised in a national, US sample. Qual Life Res. 2012;21(7):1279-90.

108. Zigmond AS, Snaith RP. The hospital anxiety and depression scale. Acta Psychiatr Scand. 1983;67(6):361-70.

109. Bradley JM, Blume SW, Balp MM, Honeybourne D, Elborn JS. Quality of life and healthcare utilisation in cystic fibrosis: a multicentre study. Eur Respir J. 2013;41(3):571-7.

110. Bradley J, McAlister O, Elborn S. Pulmonary function, inflammation, exercise capacity and quality of life in cystic fibrosis. Eur Respir J. 2001;17(4):712-5.

111. Modi AC, Lim CS, Driscoll KA, Piazza-Waggoner C, Quittner AL, Wooldridge J. Changes in pediatric health-related quality of life in cystic fibrosis after IV antibiotic treatment for pulmonary exacerbations. J Clin Psychol Med Settings. 2010;17(1):49-55.

112. Quittner AL, Modi AC, Wainwright C, Otto K, Kirihara J, Montgomery AB. Determination of the minimal clinically important difference scores for the Cystic Fibrosis Questionnaire-Revised respiratory symptom scale in two populations of patients with cystic fibrosis and chronic Pseudomonas aeruginosa airway infection. Chest. 2009;135(6):1610-8.

113. Alicandro G, Bisogno A, Battezzati A, Bianchi ML, Corti F, Colombo C. Recurrent pulmonary exacerbations are associated with low fat free mass and low bone mineral density in young adults with cystic fibrosis. J Cyst Fibros. 2014;13(3):328-34.

114. Shead EF, Haworth CS, Barker H, Bilton D, Compston JE. Osteoclast function, bone turnover and inflammatory cytokines during infective exacerbations of cystic fibrosis. J Cyst Fibros. 2010;9(2):93-8.

115. Savi D, Simmonds N, Di Paolo M, Quattrucci S, Palange P, Banya W, et al. Relationship between pulmonary exacerbations and daily physical activity in adults with cystic fibrosis. BMC Pul Med. 2015;15:151.

116. Cox NS, Alison JA, Button BM, Wilson JW, Morton JM, Holland AE. Accumulating physical activity in at least 10-minute bouts predicts better lung function after 3-years in adults with cystic fibrosis. ERJ Open Res. 2018;4(2).

117. Naon H, Hack S, Shelton MT, Gotthoffer RC, Gozal D. Resting energy expenditure. Evolution during antibiotic treatment for pulmonary exacerbation in cystic fibrosis. Chest. 1993;103(6):1819-25.

118. Abbott J, Holt A, Morton AM, Hart A, Milne G, Wolfe SP, et al. Patient indicators of a pulmonary exacerbation: preliminary reports from school aged children map onto those of adults. J Cyst Fibros. 2012;11(3):180-6.

119. Flume PA, Mogayzel PJ, Jr., Robinson KA, Rosenblatt RL, Quittell L, Marshall BC, et al. Cystic fibrosis pulmonary guidelines: pulmonary complications: hemoptysis and pneumothorax. Am J Respir Crit Care Med. 2010;182(3):298-306.

120. Skolnik K, Quon BS. Recent advances in the understanding and management of cystic fibrosis pulmonary exacerbations. F1000Res. 2018;7:575.

121. Plummer A, Wildman M, Gleeson T. Duration of intravenous antibiotic therapy in people with cystic fibrosis. Cochrane Database Syst Rev. 2016(9):CD006682.

122. Szentpetery S, Flume PA. Optimizing outcomes of pulmonary exacerbations in cystic fibrosis. Curr Opin Pulm Med. 2018;24(6):606-11.

123. Smith AL, Fiel SB, Mayer-Hamblett N, Ramsey B, Burns JL. Susceptibility testing of Pseudomonas aeruginosa isolates and clinical response to parenteral antibiotic administration: lack of association in cystic fibrosis. Chest. 2003;123(5):1495-502.

124. Abbott L, Plummer A, Hoo ZH, Wildman M. Duration of intravenous antibiotic therapy in people with cystic fibrosis. Cochrane Database Syst Rev. 2019(9):CD006682.

125. Waters V, Stanojevic S, Klingel M, Chiang J, Sonneveld N, Kukkar R, et al. Prolongation of antibiotic treatment for cystic fibrosis pulmonary exacerbations. J Cyst Fibros. 2015;14(6):770-6.

126. Hurley MN, Prayle AP, Flume P. Intravenous antibiotics for pulmonary exacerbations in people with cystic fibrosis. Paediatric respiratory reviews. 2015;16(4):246-8.

127. Ryan G, Jahnke N, Remmington T, Smyth A. Inhaled antibiotics for pulmonary exacerbations in cystic fibrosis. Paediatr Respir Rev. 2013;14(1):27-8.

128. Stanojevic S, McDonald A, Waters V, MacDonald S, Horton E, Tullis E, et al. Effect of pulmonary exacerbations treated with oral antibiotics on clinical outcomes in cystic fibrosis. Thorax. 2017;72(4):327-32.

129. Sood S, Mbarika V, Jugoo S, Dookhy R, Doarn CR, Prakash N, et al. What is telemedicine? A collection of 104 peer-reviewed perspectives and theoretical underpinnings. Telemed J E Health. 2007;13(5):573-90.

130. World Health Organisation Global Observatory for eHealth. Telemedicine: opportunities and developments in Member States: report on the second global survey on eHealth. Geneva (Switzerland): World Health Organisation; 2010 [cited 2020, January 6]. Available from: https://www.who.int/goe/publications/goe_telemedicine_2010.pdf

131. Chongmelaxme B, Lee S, Dhippayom T, Saokaew S, Chaiyakunapruk N, Dilokthornsakul P. The effects of telemedicine on asthma control and patients' quality of life in adults: A systematic review and meta-analysis. J Allergy Clin Immunol Pract. 2019;7(1):199-216.e11.

132. McLean G, Murray E, Band R, Moffat KR, Hanlon P, Bruton A, et al. Interactive digital interventions to promote self-management in adults with asthma: systematic review and meta-analysis. BMC Pulm Med. 2016;16(1):83.

133. Cruz J, Brooks D, Marques A. Home telemonitoring effectiveness in COPD: a systematic review. Int J Clin Pract. 2014;68(3):369-78.

134. Franek J. Home telehealth for patients with chronic obstructive pulmonary disease (COPD): an evidence-based analysis. Ont Health Technol Assess Ser. 2012;12(11):1-58.

135. Bolton CE, Waters CS, Peirce S, Elwyn G, Epsrc, Team MRCGC. Insufficient evidence of benefit: a systematic review of home telemonitoring for COPD. J Eval Clin Pract. 2011;17(6):1216-22.

136. Tsai LL, McNamara RJ, Moddel C, Alison JA, McKenzie DK, McKeough ZJ. Homebased telerehabilitation via real-time videoconferencing improves endurance exercise capacity in patients with COPD: The randomized controlled TeleR Study. Respirology. 2017;22(4):699-707.

137. Holland A. Telehealth reduces hospital admission rates in patients with COPD. J Physiother. 2013;59(2):129.

138. Wakefield BJ, Ward MM, Holman JE, Ray A, Scherubel M, Burns TL, et al. Evaluation of home telehealth following hospitalization for heart failure: a randomized trial. Telemed J E Health. 2008;14(8):753-61.

139. Dang S, Dimmick S, Kelkar G. Evaluating the evidence base for the use of home telehealth remote monitoring in elderly with heart failure. Telemed J E Health. 2009;15(8):783-96.

140. Dansky KH, Vasey J, Bowles K. Impact of telehealth on clinical outcomes in patients with heart failure. Clin Nurs Res. 2008;17(3):182-99.

141. Kotb A, Cameron C, Hsieh S, Wells G. Comparative effectiveness of different forms of telemedicine for individuals with heart failure (HF): a systematic review and network metaanalysis. PloS One. 2015;10(2):e0118681.

142. Cox NS, Alison JA, Rasekaba T, Holland AE. Telehealth in cystic fibrosis: a systematic review. J Telemed Telecare. 2012;18(2):72-8.

143. Gur M, Nir V, Teleshov A, Bar-Yoseph R, Manor E, Diab G, et al. The use of telehealth (text messaging and video communications) in patients with cystic fibrosis: A pilot study. J Telemed Telecare. 2017;23(4):489-93.

144. Magrabi F, Lovell NH, Henry RL, Celler BG. Designing home telecare: a case study in monitoring cystic fibrosis. Telemed J E Health. 2005;11(6):707-19.

145. Grzincich G, Gagliardini R, Bossi A, Bella S, Cimino G, Cirilli N, et al. Evaluation of a home telemonitoring service for adult patients with cystic fibrosis: a pilot study. J Telemed Telecare. 2010;16(7):359-62.

146. Bella S, Cotognini C, Alghisi F, Montemitro E. Program of home telemonitoring in patients with cystic fibrosis over a period of 2 years: a contribution to the rationalization of care. Clin Ter. 2013;164(4):e313-7.

147. Chen JJ, Cooper DM, Haddad F, Sladkey A, Nussbaum E, Radom-Aizik S. Teleexercise as a promising tool to promote exercise in children with cystic fibrosis. Front Public Health. 2018;6:269.

148. Nikander K, Denyer J, Dodd M, Dyche T, Webb K, Weller P, et al. The adaptive aerosol delivery system in a telehealth setting: patient acceptance, performance and feasibility. J Aerosol Med Pulm Drug Deliv. 2010;23 Suppl 1:S21-7.

149. Morton RW, Elphick HE, Edwards E, Daw WJ, West NS. Investigating the feasibility of text message reminders to improve adherence to nebulized medication in children and adolescents with cystic fibrosis. Patient Prefer Adherence. 2017;11:861-9.

150. Murgia F, Cotognini C, Montemitro E, Cilli M, Renzetti E, Lucidi V, et al. Evaluation of compliance to telehomecare (THC) in a group of patients with cystic fibrosis (CF) in a period of 2 years. Clin Ter. 2012;163(3):e111-4.

151. Quanjer PH, Stanojevic S, Cole TJ, Baur X, Hall GL, Culver BH, et al. Multi-ethnic reference values for spirometry for the 3-95-yr age range: the global lung function 2012 equations. Eur Respir J. 2012;40(6):1324-43.

152. Graham BL, Steenbruggen I, Miller MR, Barjaktarevic IZ, Cooper BG, Hall GL, et al. Standardization of Spirometry 2019 Update. An Official American Thoracic Society and European Respiratory Society Technical Statement. Am J Respir Crit Care Med. 2019;200(8):e70-e88.

153. Morgan DG, Kosteniuk J, Stewart N, O'Connell ME, Karunanayake C, Beever R. The telehealth satisfaction scale: reliability, validity, and satisfaction with telehealth in a rural memory clinic population. Telemed J E Health. 2014;20(11):997-1003.

154. Johnson C, Butler SM, Konstan MW, Morgan W, Wohl ME. Factors influencing outcomes in cystic fibrosis: a center-based analysis. Chest. 2003;123(1):20-7.

155. McLean S, Nurmatov U, Liu JL, Pagliari C, Car J, Sheikh A. Telehealthcare for chronic obstructive pulmonary disease: Cochrane review and meta-analysis. Br J Gen Pract. 2012;62(604):e739-49.

156. McCoy KS, Quittner AL, Oermann CM, Gibson RL, Retsch-Bogart GZ, Montgomery AB. Inhaled aztreonam lysine for chronic airway Pseudomonas aeruginosa in cystic fibrosis. Am J Respir Crit Care Med. 2008;178(9):921-8.

157. Retsch-Bogart GZ, Quittner AL, Gibson RL, Oermann CM, McCoy KS, Montgomery AB, et al. Efficacy and safety of inhaled aztreonam lysine for airway pseudomonas in cystic fibrosis. Chest. 2009;135(5):1223-32.

158. Kessler RC, Barber C, Beck A, Berglund P, Cleary PD, McKenas D, et al. The World Health Organization Health and Work Performance Questionnaire (HPQ). J Occup Environ Med. 2003;45(2):156-74.

159. Yohannes AM, Willgoss TG, Fatoye FA, Dip MD, Webb K. Relationship between anxiety, depression, and quality of life in adult patients with cystic fibrosis. Respir Care. 2012;57(4):550-6.

160. Brooke J. SUS-A quick and dirty usability scale. In: Jordan PW, Thomas B, Weerdmeester BA, McClelland AL, editors. Usability evaluation in industry. London: Taylor and Francis; 1996. p. 189-94.

161. Wood CL, Clements SA, McFann K, Slover R, Thomas JF, Wadwa RP. Use of telemedicine to improve adherence to American Diabetes Association standards in pediatric type 1 diabetes. Diabetes Technol Ther. 2016;18(1):7-14.

162. Hamine S, Gerth-Guyette E, Faulx D, Green BB, Ginsburg AS. Impact of mHealth chronic disease management on treatment adherence and patient outcomes: a systematic review. J Med Internet Res. 2015;17(2):e52.

163. J S. A practical guide to the system usability scale: Background, benchmarks & best practices. Denver CO: Createspace Independent Publishing Platform; 2011.

164. Australian Communications and Media Authority. Australian Government. Communications report 2014–15; Commonwealth of Australia 2015. [cited 2016, February 12]. Available from: <u>https://www.acma.gov.au/publications/2015-12/report/communications-report-2014-15</u>

165. Schulz KF, Altman DG, Moher D. CONSORT 2010 statement: Updated guidelines for reporting parallel group randomised trials. BMC Med. 2010;8:18.

166. Wainwright CE, Elborn JS, Ramsey BW. Lumacaftor-Ivacaftor in patients with cystic fibrosis homozygous for Phe508del CFTR. N Engl J Med. 2015;373(18):1783-4.

167. Quittner AL, Modi AC, Lemanek KL, Ievers-Landis CE, Rapoff MA. Evidence-based assessment of adherence to medical treatments in pediatric psychology. J Pediatr Psychol. 2008;33(9):916-36.

168. Wang PS, Beck A, Berglund P, Leutzinger JA, Pronk N, Richling D, et al. Chronic medical conditions and work performance in the health and work performance questionnaire calibration surveys. J Occup Environ Med. 2003;45(12):1303-11.

169. McCoy CE. Understanding the intention-to-treat principle in randomized controlled trials. West J Emerg Med. 2017;18(6):1075-8.

Every reasonable effort has been made to acknowledge the owners of copyrighted material. I would be pleased to hear from any copyright owner who has been omitted or incorrectly acknowledged.

Appendix 1

Author attribution statement

Wood, J, Mulrennan, S, Hill, K, Cecins, N, Morey, S, Jenkins, S. Telehealth clinics increase access to

care for adults with cystic fibrosis living in rural and remote Western Australia. J Telemed Telecare.

2017;23(7):673-679.

Conception and Design	Acquisition of Data and Method	Data Conditioning/ Manipulation	Analysis and Statistical Method	Interpretation and Discussion	Final Approval	Total % contribution			
Co-Author 1: Jamie Wood	~	~	~	~	~	40%			
Co-Author 1 Ack I acknowledge the Signed:	Co-Author 1 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 2: Siobhain Mulrennan	~	~		~	~	5%			
Co-Author 2 Ack I acknowledge the Signed:	Co-Author 2 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 3: Kylie Hill	~	~	~	*	*	20%			
Co-Author 3 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:									
Co-Author 4: Nola Cecins	~		~	~	~	5%			
Co-Author 4 Ack I acknowledge the Signed:	Co-Author 4 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 5: Sue Morey	~			~	~	5%			
Co-Author 5 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:									
Co-Author 6: Sue Jenkins	~	~	~	~	*	25%			
Co-Author 6 Ack I acknowledge the Signed:	nowledgment: at these represent my	contribution to the abo	ove research output						
Total %						100%			

Wood J, Jenkins S, Mulrennan S, Hill K. The impact of cystic fibrosis on work attendance and

performance in adults living in rural and remote Western Australia. J Cyst Fibros. 2017;16(2):e1-e2.

Conception and Design	Acquisition of Data and Method	Data Conditioning/ Manipulation	Analysis and Statistical Method	Interpretation and Discussion	Final Approval	Total % contribution	
Co-Author 1: Jamie Wood	~	~	~	~	~	45%	
Co-Author 1 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Co-Author 2: Sue Jenkins	4	4	4	~	~	20%	
Co-Author 2 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Co-Author 3: Siobhain Mulrennan	~	\checkmark		~	~	10%	
Co-Author 3 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Co-Author 4: Kylie Hill	4	~	~	*	*	25%	
Co-Author 4 Acknowledgment: I acknowledge that these represent my contribution to the above research output							
Total %						100%	

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey S, Cecins N, et al. High usability of a smartphone

application for reporting symptoms in adults with cystic fibrosis. J Telemed Telecare.

2017;24(8):547-552.

Conception and Design	Acquisition of Data and Method	Data Conditioning/ Manipulation	Analysis and Statistical Method	Interpretation and Discussion	Final Approval	Total % contribution	
Co-Author 1: Jamie Wood	~	~	~	~	~	40%	
Co-Author 1 Ack I acknowledge the Signed:	nowledgment: at these represent my	contribution to the abo	ove research output				
Co-Author 2: Sue Jenkins	~	~	~	~	~	15%	
Co-Author 2 Ack I acknowledge that Signed:	nowledgment: at these represent my o	contribution to the abo	ove research output				
Co-Author 3: David Putrino	~		~	~	~	10%	
Co-Author 3 Ack I acknowledge the Signed:	Co-Author 3 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:						
Co-Author 4: Siobhain Mulrennan	~			~	~	5%	
Co-Author 4 Ack I acknowledge the Signed:	nowledgment: at these represent my	contribution to the abo	ove research output				
Co-Author 5: Sue Morey	~			~	~	5%	
Co-Author 5 Ack I acknowledge the Signed:	Co-Author 5 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:						
Co-Author 6: Nola Cecins	4			~	~	5%	
Co-Author 6 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Co-Author 7: Kylie Hill	4	✓	~	~	~	20%	
Co-Author 7 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Total %						100%	

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey S, Cecins N, et al. A smartphone application for

reporting symptoms in adults with cystic fibrosis: protocol of a randomised controlled trial. BMJ

Open. 2018;8(4):e021136.

Conception and Design	Acquisition of Data and Method	Data Conditioning/ Manipulation	Analysis and Statistical Method	Interpretation and Discussion	Final Approval	Total % contribution		
Co-Author 1: Jamie Wood	~		~	~	~	40%		
Co-Author 1 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 2: Sue Jenkins	~		~	1	~	15%		
Co-Author 2 Ack I acknowledge the Signed:	Co-Author 2 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Co-Author 3: David Putrino	~		~	~	~	10%		
Co-Author 3 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed								
Co-Author 4: Siobhain Mulrennan	~			~	~	5%		
Co-Author 4 Ack I acknowledge the Signed:	nowledgment: at these represent my	contribution to the abo	ove research output					
Co-Author 5: Sue Morey	~			~	~	5%		
Co-Author 5 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 6: Nola Cecins	~			~	~	5%		
Co-Author 6 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 7: Kylie Hill	~		~	~	~	20%		
Co-Author 7 Ack I acknowledge the Signed:	Co-Author 7 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Total %						100%		

Wood J, Jenkins S, Putrino D, Mulrennan S, Morey S, Cecins N, et al. A smartphone

application for reporting symptoms in adults with cystic fibrosis improves the detection of

exacerbations: Results of a randomised controlled trial. J Cyst Fibros. 2019;S1569-

1993(19)30887-2.

Conception and Design	Acquisition of Data and Method	Data Conditioning/ Manipulation	Analysis and Statistical Method	Interpretation and Discussion	Final Approval	Total % contribution		
Co-Author 1: Jamie Wood	~	~	~	~	~	40%		
Co-Author 1 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 2: Sue Jenkins	~	~	~	~	~	15%		
Co-Author 2 Ack I acknowledge th Signed:	Co-Author 2 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Co-Author 3: David Putrino	~	~	~	~	~	10%		
Co-Author 3 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 4: Siobhain Mulrennan	~			~	~	5%		
Co-Author 4 Ack I acknowledge th Signed:	nowledgment: at these represent my	contribution to the abo	ove research output					
Co-Author 5: Sue Morey	1			~	~	5%		
Co-Author 5 Ack I acknowledge th Signed:	nowledgment: at these represent my	contribution to the abo	ove research output					
Co-Author 6: Nola Cecins	~			~	~	5%		
Co-Author 6 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:								
Co-Author 7: Kylie Hill	~	~	~	~	~	20%		
Co-Author 7 Ack I acknowledge th Signed:	Co-Author 7 Acknowledgment: I acknowledge that these represent my contribution to the above research output Signed:							
Total %						100%		

Appendix 2

Original Article

Telehealth clinics increase access to care for adults with cystic fibrosis living in rural and remote Western Australia

Jamie Wood^{1,2,3}, Siobhain Mulrennan^{3,4}, Kylie Hill^{1,3}, Nola Cecins², Sue Morey^{3,4} and Sue Jenkins^{1,2,3}.

¹School of Physiotherapy and Exercise Science, Curtin University, Perth, Western Australia
 ²Physiotherapy Department, Sir Charles Gairdner Hospital, Perth, Western Australia
 ³Institute for Respiratory Health, Sir Charles Gairdner Hospital, Perth, Western Australia
 ⁴Department of Respiratory Medicine, Sir Charles Gairdner Hospital, Perth, Western Australia

Corresponding author:

Jamie Wood, Physiotherapy Department, A block, Sir Charles Gairdner Hospital, Hospital Avenue, Nedlands, Western Australia, 6009

Email: jamie.wood@health.wa.gov.au
Abstract

Background: A significant proportion (15%, n=28) of the adults with cystic fibrosis (CF) in Western Australia (WA) live in rural and remote areas and have difficulty accessing specialist care at the state adult CF centre, located in Perth. We aimed to increase access by offering telehealth clinics, and evaluate the impact on health outcomes.

Methods: Telehealth clinics were offered via videoconference over a 12-month period, with uptake and satisfaction measured at the end of the intervention. Participants could still attend in person clinics at the CF centre if requested. Other outcomes comprised healthcare utilisation (HCU), spirometry, weight and health-related quality of life.

Results: In 21 participants, total clinic visits increased from 46 (median [range] per participant 2 [0 to 6]) in the 12-month period preceding the study to 100 (5 [2 to 8] p<0.001) during the intervention. Of the 100 clinics in total, 66 were delivered via telehealth. Satisfaction with telehealth was high and most (94%) participants agreed that telehealth is a good way to deliver CF care. An increase in intravenous antibiotic days (incident rate ratio [IRR] 2.3, p=0.03) and hospital admission days (IRR 3.7, p=0.01) was observed. There was an improvement in the vitality domain of the Cystic Fibrosis Questionnaire – Revised (p<0.05).

Conclusion: Telehealth had good uptake and increased clinic attendance in adults with CF living in rural and remote WA, and had high satisfaction amongst participants. The increase in HCU, resulting from increased detection and treatment of exacerbations, may improve long-term outcomes in this population.

Keywords

Telehealth, cystic fibrosis, videoconference, rural and remote, healthcare utilisation.

Introduction

Individuals living in areas of low population density, such as rural and remote areas of Australia, have poorer health outcomes when compared with those living in metropolitan areas.¹ A key contributor to this is reduced access to care, being the inability of an individual to utilise health services that are necessary to improve or maintain their health. This is due in part to the lower numbers of skilled health professionals in these areas, and the large distances needed to travel to specialist healthcare facilities.² Of note, earlier work has demonstrated that individuals with cystic fibrosis (CF), who have complex healthcare needs,³ have poorer health outcomes when not managed in specialist CF centres.⁴, ⁵

In Western Australia (WA), the state adult specialist CF centre is at Sir Charles Gairdner Hospital (SCGH), in Perth. Of the 190 adults with CF who attend this centre, 28 (15%) live outside Perth, with distances to the centre ranging between 100 and 2,567 km. Due to limited specialist CF resources in these rural and remote areas, adults with CF have previously travelled long distances (via car or plane) to attend their scheduled routine clinics at SCGH. The travel and financial burden for these individuals is a major contributor to their low rates of attendance at clinics. The Australian CF Standards of Care⁶ recommend a minimum of four clinic visits per patient per year; however, in the year preceding this study an internal audit revealed that only 22% of adults living in rural and remote WA met this recommendation. Therefore, we sought to explore the use of telehealth in the provision of clinics for adults with CF living in rural and remote WA to improve access to our specialist centre.

Telehealth utilises audio-visual and medical technology to enable remote healthcare, monitoring and rehabilitation of individuals in their home or at a local medical centre. Although previous studies have shown that individuals with CF are willing to utilise telehealth technology,⁷ to the best of our knowledge there are no published studies on the feasibility or efficacy of providing clinics via telehealth as part of routine outpatient care in adults with CF living in rural and remote areas. Therefore, the aim of this study was to report on the uptake and participant satisfaction with telehealth clinics for adults with CF living in rural and remote WA, and to evaluate the effect of this service on health outcomes such as healthcare utilisation (HCU), spirometry, weight and health-related quality of life (HRQoL).

Such information is likely to be important because increased clinic attendance at a specialist CF centre by adults with CF who live in rural and remote WA may assist with earlier detection and treatment of respiratory exacerbations, which in turn may slow the rate of decline in spirometry and optimise HRQoL.

Methods

Study design

This was a single group study in which adults with CF who lived in rural and remote WA were invited to utilise telehealth clinics as part of routine care over a 12-month intervention period. Participants were recruited between July 2013 and August 2014. Approval was obtained from the Sir Charles Gairdner Group and Curtin University and all participants provided written informed consent prior to data collection.

Participants

Adults with CF aged 18 years or more and living a minimum of 100km from the specialist CF centre were eligible to participate in this study. Exclusion criteria were pregnancy, previous lung transplantation or current listing for lung transplantation, as individuals meeting these criteria were already required to travel to Perth more frequently for medical care.

152

Telehealth clinics

We aimed to provide telehealth clinics approximately every three months, or sooner if clinically indicated, for reasons including self-reported acute deterioration in respiratory status or early follow up after a course of intravenous antibiotics (IVAB). Telehealth clinics were scheduled to suit the participants, specialist CF team, regional telehealth coordinators, and a health professional local to the participant (physiotherapist, nurse or general practitioner).

On the day of the telehealth clinic, participants were attended the nearest regional hospital. The participant's height, weight and spirometry were measured by a designated health professional beforehand. Telehealth clinics were held via videoconference using Polycom HDX® series (Polycom, San Jose, USA) equipment at both SCGH and hospitals throughout rural and remote WA. During telehealth clinics, the participant chose whether to be reviewed by each member of the multidisciplinary CF team individually (i.e. consultant physician, nurse practitioner, physiotherapist and dietician) or by the whole team together. A CF pharmacist and social worker were also available to provide input via telephone at a later date if requested by the participant or the team. Following the telehealth clinic, prescriptions were faxed or medications were couriered to the participant's local pharmacy.

Traditional in-person clinics at SCGH were still provided if requested by the participant or if clinically indicated as previously described. We encouraged participants to attend one in-person clinic during the intervention to allow more comprehensive annual review of their clinical status. If a participant required admission to hospital at any time during the study, this was arranged at the specialist CF centre in Perth. Courses of home IVAB were also commenced at the CF centre before the participant returned home. This allowed for intensive medical and allied health input to be provided, as well as access to specialist secondary services such as intravenous access, endocrinology, gastroenterology and immunology.

Outcomes

Uptake of telehealth was measured via the number of telehealth clinic visits. Clinic attendance data, recorded in the participant's medical record, were extracted at the end of the 12-month intervention period and compared with the 12-month period preceding recruitment to the study.

Participant satisfaction with telehealth clinics was measured at the end of the intervention period using two questionnaires; the Telehealth Satisfaction Scale (TeSS)⁸ and a purpose developed satisfaction survey. The TeSS is a 10 item questionnaire originally developed to assess patient satisfaction with telehealth in a rural memory clinic.⁸ For the purposes of this study, the term 'memory clinic team' was replaced with 'CF team' and the final question "How well the staff answered your questions about the equipment", was omitted as the participants in this study were not required to operate the telehealth equipment. The second survey contained six questions that were answered using a 5-point Likert scale (responses ranging from strongly agree to strongly disagree), and provided the participant with the opportunity to provide further feedback on the telehealth clinics including preference for either telehealth or traditional 'in person' clinics. This survey was piloted in adults with CF (n=5) living in the metropolitan area and minor changes were made to optimise its readability. Both surveys were mailed to participants, and were returned by mail in pre-addressed, postage paid envelopes. The content of the surveys did not allow the researchers to identify the participants.

Healthcare utilisation data (total number of clinics, courses and days of IVAB, courses of inhaled and oral antibiotics [AB], hospital admissions and admission days) were extracted from each participant's medical record at the end of the intervention period and compared with the 12-month period preceding recruitment to the study.

Spirometry (FEV₁ and FVC) was measured according to accepted guidelines⁹ using either CPFS/D USBTM (MGC Diagnostics Corporation, St Paul, USA) or EasyoneTM (NDD Medical Technologies, Andover, USA) spirometers. Health professionals who performed spirometry had previously received training from a certified respiratory technician. Weight and height were used to calculate body mass index (BMI). Health-related quality of life was collected at each clinic using the Cystic Fibrosis Questionnaire – Revised (CFQ-R).¹⁰

Analyses

Statistical analyses were conducted using Stata[®] (StataCorp. 2015. Stata Statistical Software: Release 14. College Station, TX: StataCorp LP). All data were checked for normality. Satisfaction data were reported using descriptive statistics. Healthcare utilisation data were compared pre- and during- the 12-month intervention period with Poisson or negative binomial regression. Paired t-tests and Wilcoxon signed rank tests were used to compare spirometry, weight, BMI, and HRQoL data pre- and during- the 12-month intervention period as appropriate. Repeated measures analyses of these secondary outcomes were not undertaken as the study was not powered to detect differences. A p-value of < 0.05 was used to denote statistical significance. As the primary aim of this study was to report the uptake of telehealth clinics amongst all individuals with CF living in rural and remote WA, no sample size calculations were undertaken.

Results

Twenty-three out of a possible 28 (82%) adults with CF living in rural and remote WA were recruited for this study (Table 1). Reasons for not participating were regular travel to Perth for other medical reasons (n=3) and intention to move interstate or overseas within the intervention period (n=2). The distance participants were required to travel to their nearest regional hospital was a median (range) of 5 (1 to 22) km (Table 1). Data from two participants were excluded from the analysis as they moved residence during the intervention period (one to Perth and one internationally). One participant chose to travel to Perth for all clinic visits and did not utilise the telehealth intervention, therefore they were not asked to complete the satisfaction surveys.

Uptake of the telehealth clinics

The 21 participants included in the final analyses attended a total of 100 clinics during the intervention period, of which 66 (66%) were provided via telehealth (Table 2). Four participants (19%) utilised telehealth for all of their clinics, however three of these individuals also had hospital admissions at SCGH during the intervention period, and subsequent in-person contact with CF team at this time. The median (range) number of clinics per participant during the intervention period increased from 2 (0-6) to 5 (2-8) p<0.001 (Table 2). A total of 19 of the 21 participants (90%) attended four or more clinics during the intervention period, compared with 4 (19%) in the previous year. On two occasions, participants did not attend their scheduled telehealth clinic and failed to inform the CF team. Of the 34 in-person clinics where participants attended the specialist CF centre, 19 (56%) were requested by the CF team for clinical indications.

Satisfaction

The TeSS and purpose developed participant satisfaction survey data were available for 17 (81%) participants. The participants' rated all nine items on the TeSS as either 'good' or 'excellent', with the exception of one participant's response to the item relating to the voice quality of the equipment (Table 3).

Results from the purpose developed satisfaction survey (Table 4) showed that 16 (94%) participants strongly agreed or agreed that telehealth clinics were a good way to manage CF care. Ten (59%) participants preferred telehealth clinics for their routine outpatient care.

Healthcare utilisation

During the intervention period, there was an increase in IVAB days, hospital admissions and admission days per participant (Table 2) (all p<0.05). The number of courses of IVAB, oral AB and inhaled AB per participant also increased, however this was not significant (Table 4). Of the 25 exacerbations requiring IVAB during the intervention period, 20 (80%) were detected during telehealth clinics.

Spirometry, BMI and HRQoL

156

No significant changes were seen in spirometry (FEV₁ and FVC) following the intervention. In the 20 participants for whom weight gain was a goal, there was an increase in median (IQR) BMI from 20.9 (4.7) to 21.7 (3.4), however this was not significant (p=0.12). One participant was excluded from the BMI analysis as they were morbidly obese and weight loss was being encouraged. Fourteen (70%) participants gained weight during the intervention, with a mean (SD) increase of 2.2 (2.1) kg, or 3.7 (3.6) %. There was an improvement in the vitality domain (p=0.04) of the CFQ-R, and a trend toward improvement in the weight domain (p=0.06) (Table 5). No changes were observed in other domains of the CFQ-R.

Discussion

This study is the first to evaluate the uptake of telehealth clinics implemented as part of routine outpatient care in individuals with CF living in rural and remote areas. We have demonstrated that telehealth technology increases access to care for these individuals and resulted in 90% of participants meeting the recommended number of clinic visits⁶ during the intervention period. Although both telehealth and traditional in-person clinics were offered during the intervention period, participants chose the telehealth option for more than two-thirds of their clinics. One participant did not attend the CF centre (for a clinic or hospital admission) as it was not convenient for them during the intervention period. This is a potential limitation of telehealth, as certain assessments (e.g. bone mineral density, general respiratory function testing and chest computed tomography) are not readily available in all rural and remote areas. While one participant chose not to use the telehealth intervention, the inclusion of their data in measures related to healthcare utilisation provides a more conservative and real world estimate of the effect telehealth had on these outcomes.

The observed increase in antibiotic use and hospital admissions reflects increased surveillance of the participants, and a corresponding increased detection of respiratory exacerbations. Although this short-term increase in HCU was observed, the increased frequency of clinic visits and contact with the clinical team as a result of this telehealth intervention may lead to increased patient adherence and improved longer term health outcomes. Specifically, more frequent contact with the CF team¹¹ and earlier detection of exacerbations that will allow treatment to be initiated sooner,¹² is likely to slow the rate of decline of FEV₁, and potentially improve nutritional status and HRQoL.

Recruitment for this study was good (over 80%), with only five potential participants not enrolling because they were required to travel to Perth for other medical reasons, or had intentions to move during the intervention period. The most likely explanation for the high recruitment rate is the reduction in travel burden as a result of telehealth, as most of the participants travelled many hours by car or plane to Perth to attend in-person clinics in Perth. It is also likely that our recruitment was influenced by the mean age (31.4 years) of the study sample, as younger adults may be more open to the idea of integrating modern technology into their care. Our study adds support to previous studies reporting that individuals with CF are willing to utilise telehealth technology.⁷

Responses to the TeSS reflected an excellent level of satisfaction with telehealth clinics. The only item given a score less than three (out of four) by a participant was regarding the voice quality of the equipment, which could be attributed to the voice delay occasionally experienced with the system. The TeSS responses demonstrated that participants perceived telehealth clinics to be a highly acceptable method to provide regular healthcare review. However, responses to the purpose developed participant satisfaction survey indicated that some participants (12%) preferred traditional in-person clinics over telehealth clinics and some had no preference (29%). This is understandable given that in-person clinics have been offered for many years and have advantages such as the ability for physical examination, access to on-site specialist secondary services (e.g. intravenous access, endocrinology, immunology and hepatology) and immediate admission to hospital if required.

In keeping with this finding, a meta-analysis of telehealth for Chronic Obstructive Pulmonary Disease found that individuals were largely satisfied with receiving care via telehealth at home, as long as face to face consultations were still available on request.¹³ The positive view of telehealth among our participants is consistent with previous telehealth studies in CF,⁷ and provides merit to the continuation of its use as part routine CF care. In a home monitoring study measuring feasibility and satisfaction,¹⁴ 80% of participants with CF stated they would opt to continue recording of spirometry, oxygen saturation and symptoms on a daily basis for routine management.

It is difficult to make comparisons between the HCU results of this study and other telehealth studies in CF due to the differences in the interventions and study designs. Weekly videoconferencing with the CF team has previously been utilised by a small number of individuals with CF (n=7) awaiting lung transplantation over a six-month period, and no differences in HCU were found when compared to usual care.¹⁵ Videoconferencing has also been used effectively as a means of assessing functional exercise capacity¹⁶ In 19 individuals with CF from adult and paediatric centres, the monitoring of spirometry and symptoms on a daily basis at home over a six-month resulted in no difference in the number of exacerbations per participant when compared to the same period prior to the study; however an increase in courses of oral AB was noted during the intervention.¹⁷ Another intervention¹⁸ comparing daily monitoring of spirometry and oximetry (n=17) to usual care (n=28) over seven months found no difference in the rate of hospitalisation, but an increase in clinic visits during the intervention period was observed. Despite the differences in study designs across these studies, many share an observed increase in certain measures of HCU. This study was not powered to detect changes in spirometry (FEV₁ and FVC), and therefore the lack of significant difference in these measures was not unexpected. The increased treatment of exacerbations may explain the improvement in the vitality domain of the CFQ-R, which consists of four questions relating to energy levels. Although the minimal clinically important difference for the vitality domain of the CFQ-R has not been established, the magnitude of change in this domain demonstrated in our study (11 out of 100 points) exceeded the magnitude of change (8 points) reported following 28 days of inhaled Aztreonam Lysine in individuals with CF colonised with Pseudomonas aeruginosa. As the change in HRQoL alongside the improvement in pulmonary function was considered to be clinically meaningful to the participants in this earlier study, it is likely that the improvement seen in our study would also to be perceived as clinically meangingful.¹⁹ The trend towards improvement in the weight domain of the CFQ-R, alongside the weight gain in fourteen (67%) of the participants may reflect an increase in dietetic input delivered during the intervention period, as well as the improved management of respiratory exacerbations. It is well established that in individuals with CF, a loss of appetite and weight are associated with exacerbations²⁰ and earlier management of such episodes may have minimised weight loss during the intervention period. This study demonstrated that telehealth clinics are a feasible way to deliver care for individuals with CF living in rural and remote areas, and as a result this service will continue to be offered to adults living in Western Australia. Adults with CF in these areas are willing to utilise this technology and most prefer telehealth clinics over traditional in-person clinics offered at the specialist CF centre at SCGH. There are limitations to our study, such as a small sample and lack of control group, which are also common among many of the published telehealth studies in CF to date.⁷ Furthermore, other clinical measures commonly used in CF care such as sputum microbiology, inflammatory and diabetic markers, and chest radiography were not included as outcomes as they were beyond the scope of this study. Given the differences in geography, CF centre resources and government health service provision between states and countries throughout the world, it would be difficult to conduct a multicentre randomised controlled trial in this area. Therefore, future studies can focus on the evaluation of the longer term implementation of telehealth services as part of routine care, and their impact on a wider range of health outcomes.

Acknowledgements

This study was partially funded by the Institute for Respiratory Health Glenn Brown Memorial Grant. We would like to acknowledge the participants in this study, North Metropolitan Health Service Telehealth Department, and WA Country Health Service.

Declaration of Conflicting Interests

Mr. Wood reports grants from Institute for Respiratory Health, during the conduct of the study; grants from Technipro Pulmomed, non-financial support from Novartis, non-financial support from Pharmaxis, outside the submitted work. Clin A/Prof Mulrennan reports grants from Institute for Respiratory Health, during the conduct of the study; personal fees from Vertex, non-financial support from Novartis, outside the submitted work. A/Prof Hill has nothing to disclose. Mrs. Cecins has nothing to disclose. Mrs. Morey has nothing to disclose. A/Prof Jenkins reports grants from Institute for Respiratory Health, during the respiratory Health, during the conduct of the study.

References

1. Australia's Health 2014. Canberra: Australian Institute for Health and Welfare, 2014. http://www.aihw.gov.au/publication-detail/?id=60129547205. Accessed July 15, 2015.

 Bywood P KR, Lunnay B. Disparities in primary health care utilisation: Who are the disadvantaged groups? How are they disadvantaged? What interventions work? PHCRIS Policy Issue Review.
 Adelaide: Primary Health Care Research and Information Service, 2011.

3. Elborn JS, Bell SC, Madge SL, et al. Report of the European Respiratory Society/European Cystic
Fibrosis Society task force on the care of adults with cystic fibrosis. The European respiratory journal
: official journal of the European Society for Clinical Respiratory Physiology. 2016; 47: 420-8.

4. Lebecque P, Leonard A, De Boeck K, et al. Early referral to cystic fibrosis specialist centre impacts on respiratory outcome. J Cyst Fibros. 2009; 8: 26-30.

5. Mahadeva R, Webb K, Westerbeek RC, et al. Clinical outcome in relation to care in centres specialising in cystic fibrosis: cross sectional study. Bmj. 1998; 316: 1771-5.

6. Bell SC. Cystic Fibrosis Standards of Care, Australia. 2008.

http://www.cysticfibrosis.org.au/media/wysiwyg/CF-

Australia/PDF_files/CFA_Standards_of_Care_journal_31_Mar_08.pdf. Accessed November 12, 2012.

7. Cox NS, Alison JA, Rasekaba T and Holland AE. Telehealth in cystic fibrosis: a systematic review.J Telemed Telecare. 2012; 18: 72-8.

8. Morgan DG, Kosteniuk J, Stewart N, O'Connell ME, Karunanayake C and Beever R. The telehealth satisfaction scale: reliability, validity, and satisfaction with telehealth in a rural memory clinic population. Telemedicine journal and e-health : the official journal of the American Telemedicine Association. 2014; 20: 997-1003.

 Miller MR, Hankinson J, Brusasco V, et al. Standardisation of spirometry. Eur Respir J. 2005; 26: 319-38.

10. Quittner AL, Sawicki GS, McMullen A, et al. Psychometric evaluation of the Cystic Fibrosis Questionnaire-Revised in a national sample. Qual Life Res. 2012; 21: 1267-78.

11. Johnson C, Butler SM, Konstan MW, Morgan W and Wohl ME. Factors influencing outcomes in cystic fibrosis: a center-based analysis. Chest. 2003; 123: 20-7.

12. Sanders DB, Bittner RC, Rosenfeld M, Hoffman LR, Redding GJ and Goss CH. Failure to recover to baseline pulmonary function after cystic fibrosis pulmonary exacerbation. American journal of respiratory and critical care medicine. 2010; 182: 627-32.

13. McLean SC and McKinstry B. Meta-analysis on COPD: comment on 'Home telehealth for chronic obstructive pulmonary disease: a systematic review and meta-analysis. by J Polisena et al.'. Journal of telemedicine and telecare. 2012; 18: 242; author reply

14. Grzincich G, Gagliardini R, Bossi A, et al. Evaluation of a home telemonitoring service for adult patients with cystic fibrosis: a pilot study. J Telemed Telecare. 2010; 16: 359-62.

15. Wilkinson OM, Duncan-Skingle F, Pryor JA and Hodson ME. A feasibility study of home telemedicine for patients with cystic fibrosis awaiting transplantation. J Telemed Telecare. 2008; 14: 182-5.

16. Cox NS, Alison JA, Button BM, Wilson JW and Holland AE. Assessing exercise capacity using telehealth: a feasibility study in adults with cystic fibrosis. Respiratory care. 2013; 58: 286-90.

17. Sarfaraz S, Sund Z and Jarad N. Real-time, once-daily monitoring of symptoms and FEV in cystic fibrosis patients--a feasibility study using a novel device. Clin Respir J. 2010; 4: 74-82.

Bella S, Murgia F, Tozzi AE, Cotognini C and Lucidi V. Five years of Telemedicine in Cystic
 Fibrosis Disease. Clin Ter. 2009; 160: 457-60.

19. Retsch-Bogart GZ, Quittner AL, Gibson RL, et al. Efficacy and safety of inhaled aztreonam lysine for airway pseudomonas in cystic fibrosis. Chest. 2009; 135: 1223-32.

20. Rabin HR, Butler SM, Wohl ME, et al. Pulmonary exacerbations in cystic fibrosis. Pediatr Pulmonol. 2004; 37: 400-6.

Table 10. Participant (n=23) characteristics at baseline.

Participants, n	23
Female, n (%)	14 (61)
Age, yr	31.4 (10.2)
BMI, kg/m ²	21.1 (18.7 to 58.0) ^a
FEV ₁ , L	2.08 (0.78)
FEV ₁ , % predicted	60 (20)
Distance from specialist CF centre, km	417 (100 to 2,567) ^a
Distance from nearest regional hospital, km	5 (1 to 22) ^a
Pseudomonas aeruginosa, n (%)	21 (91)
CF-related diabetes, n (%)	4 (17)
Pancreatic insufficiency, n (%)	21 (91)

Data are presented as mean (SD) unless otherwise stated.

^a data are presented as median (range).

Table 2. Healthcare utilisation data for participants (n=21) who completed the intervention,

comparing the 12-month periods pre- and during telehealth.

	Pre -T	elehealth	During	g Telehealth			
	Nº	per participant	Nº	per participant	IRR	Std. err.	p
Clinics	46	2 (0 to 6)	100	5 (2 to 8)	2.2	0.4	< 0.001
IVAB courses	14	0 (0 to 3)	25	1 (0 to 4)	1.8	0.3	0.08
IVAB days	170	8 (0 to 32) ^b	350	17 (0 to 61) ^a	2.3	0.8	0.03 ^a
Hospital admissions	8	0 (0 to 2)	19	1 (0 to 3)	2.4	1.0	0.04
Admission days	56	3 (0 to 15) ^b	209	10 (0 to 36) ^a	3.7	0.4	0.01 ^a
Oral AB courses	24	1 (0 to 3) ^b	29	1 (0 to 3) ^a	1.2	0.3	0.5
Inhaled AB courses	18	1 (0 to 2)	31	2 (0 to 3)	1.7	0.5	0.07

Data are presented as number (N°) and median (range). IRR = incidence rate ratio. Std. err. = standard

error. Analyses performed using Poisson regression or ^a negative binomial regression.

^b data are presented as mean (range).

Table 3. Participant (n=17) responses to questions from the TeSS.

Overall, how satisfied were you with:	
The voice quality of the equipment	4 (1 to 4)
The visual quality of the equipment	4 (3 to 4)
Your personal comfort using the telehealth system	4 (3 to 4)
The ease of getting to the telehealth department	4 (3 to 4)
The length of time with the CF team	4 (3 to 4)
The explanation of your treatment by the CF team	4 (3 to 4)
The thoroughness, carefulness and skilfulness of the CF team	4 (3 to 4)
The courtesy, respect, sensitivity, and friendliness of the CF team	4 (3 to 4)
How well your privacy was respected	4 (3 to 4)

Data are presented as median (range). Questions were answered using a 4-point scale: Excellent = 4, Good = 3, Fair = 2, Poor = 1.

167

Table 4. Participant (n=17) responses to questions from the purpose developed satisfaction survey, with data presented as percentage of participants.

	SA	А	U	D	SD
The organisation of the clinics met my expectations	29	65	6	0	0
The timing of the telehealth clinic appointments were convenient for me	47	53	0	0	0
The healthcare I received during the telehealth clinics met my expectations	35	59	6	0	0
I was able to communicate effectively with the CF team using this technology	35	59	6	0	0
I prefer telehealth clinics over traditional 'face to face' clinics	18	41	29	12	0
Telehealth clinics are a good way to manage my CF care	18	76	6	0	0

Questions were answered using a 5-point Likert scale: SA = Strongly Agree, A = Agree, U = Unsure, D

= Disagree, SD = Strongly Disagree.

CFQ-R domain	Pre -Telehealth	Post -Telehealth	Difference ^c	р
Physical functioning ^a	71 [33 to 42]	75 [42 to 90]	-	0.64 ^b
Vitality	44 (21)	55 (20)	11 (1 to 22)	0.04
Emotional functioning	69 (23)	71 (19)	2 (-9 to 12)	0.75
Eating ^a	100 [61 to 100]	100 [67 to 100]	-	0.43 ^b
Treatment burden	58 (17)	56 (20)	-2 (-9 to 4)	0.51
Health perceptions	53 (23)	58 (19)	5 (-7 to 17)	0.39
Social	63 (20)	61 (18)	-2 (-8 to 5)	0.56
Body Image	71 (26)	68 (28)	-3 (-14 to 8)	0.56
Role	78 (16)	77 (21)	-1 (-10 to 9)	0.86
Weight ^a	67 [33 to 100]	100 [50 to 100]	-	0.06 ^b
Respiratory symptoms	55 (24)	64 (19)	9 (-3 to 20)	0.13
Digestive symptoms ^a	89 [67 to 100]	89 [67 to 100]	-	0.97 ^b

Table 5. Participant (n=21) responses to the CFQ-R pre - and post - telehealth intervention.

Data are presented as mean (SD) (normally distributed data) or ^a median [IQR] (non-normally distributed data). An increase in score represents improvement, with a maximum score of 100. Analysis performed using paired t test or ^b Wilcoxon signed rank test.

^c data are presented as mean (95% CI)

Appendix 3

Letter to the Editor

The impact of cystic fibrosis on work attendance and performance in adults living in rural and remote Western Australia

Jamie Wood^{1,2,3}, Sue Jenkins^{1,2,3}, Siobhain Mulrennan^{3,4} and. Kylie Hill^{1,3}

¹School of Physiotherapy and Exercise Science, Curtin University, Perth, Western Australia

²Physiotherapy Department, Sir Charles Gairdner Hospital, Perth, Western Australia

³Institute for Respiratory Health, Sir Charles Gairdner Hospital, Perth, Western Australia

⁴Department of Respiratory Medicine, Sir Charles Gairdner Hospital, Perth, Western Australia

Corresponding author:

Jamie Wood, Physiotherapy Department, A block, Sir Charles Gairdner Hospital, Hospital Avenue,

Nedlands, Western Australia, 6009

Email: jamie.wood@health.wa.gov.au

Keywords:

Absenteeism; presenteeism; cystic fibrosis; employment; work

Letter to the Editor

The impact of cystic fibrosis on work attendance and performance in adults living in rural and remote Western Australia

Dear Editor,

As the expected age of survival increases, adults with CF will likely stay in the work force for longer. Previous studies have reported that, in adults with CF, employment status is related to age and time spent in hospital [1], lung function and quality of life [2]. While it is known that many (67%) engage in paid employment [3], to our knowledge there is minimal data available regarding the impact that CF has on work attendance (absenteeism) [4] and performance (presenteeism) [5]. At Sir Charles Gairdner Hospital in Perth, we recently evaluated telehealth clinics for adults with CF living in rural and remote Western Australia (WA), and, as part of our study quantified absenteeism and presenteeism.

During outpatient clinic visits (telehealth and in-person) throughout a 12-month period, participants completed the absenteeism and presenteeism questions of the World Health Organisation's Health and Work Performance Questionnaire [6]. This self-reported 11 item questionnaire collects information regarding; (i) expected and actual hours worked during the previous 7 and 28 days, (ii) part or entire days of work missed (for health or holiday), (iii) overtime worked and (iv) perceived work performance (rated out of 10) for the participant and colleagues undertaking similar work. Responses are used to calculate relative absenteeism and relative presenteeism. Relative absenteeism is the percentage of time absent from work. A maximum score of 1 indicates 100% absenteeism; a score of 0 indicates no absenteeism. Negative scores indicate the hours worked exceeded employer's expectations. Relative presenteeism is work performance compared to others in similar employment. A score of 1 represents equal performance, 0 represents worst performance, and a maximum score of 2 represents performance at least twice as good as colleagues.

172

Twenty-three adults with CF (14 female, aged mean \pm SD 31 \pm 10 years, forced expiratory volume in one second 60 \pm 20% predicted, living median [range] 417 [100 to 2567] km from Perth) completed the study. Data were collected during a total of 91 clinic visits. Participants were engaged in paid work on 61 (67%) of these occasions. Twenty two (96%) participants were employed on at least one occasion, and 11 (48%) were employed on all occasions throughout the entire 12 months. Only one participant, a fulltime carer, had no paid work in this time.

For those employed, the median (range) number of hours worked per week was 38 (10 to 80). The mean (range) number of entire work days missed in the previous 28 days because of problems with physical or mental health was 2 (0 to 21) days, while the number of part work days missed in this period for the same reason was to 1 (0 to 11) days. Participants worked longer hours than expected on 2 (0 to 28) days during the previous 28 days. Overall, relative absenteeism (mean [range]) in those employed was 0.1 (-1.2 to 0.9).

Participants' reported that their work performance over the past one to two years was median (range) 8 (5 to 10) out of 10, which was similar to how they rated the performance of colleagues in similar employment, also being 8 (5 to 10). However when asked to rate their work performance over the past 28 days, the participants' responses were more variable (8 [0 to 10]). Overall, relative presenteeism in those employed was 1.0 (0.0 to 1.8).

These data provide further evidence of the potential impact of cystic fibrosis on an individual's ability to work. Despite having a chronic disease that is associated with a significant treatment burden, most adults with CF living in rural and remote WA engaged in paid work during the 12-month study period. Absenteeism scores were variable, and demonstrated that CF affected the work attendance of many, but not all participants. Presenteeism scores, while also variable, demonstrated that on average participants felt they performed as well as their colleagues at work. However with many reporting that CF does have has a large impact on their work performance, further work is required in larger samples to understand the full impact of CF on presenteeism. Measuring absenteeism and presenteeism provides a useful insight into the disease-related impact on work attendance and performance in adults with CF, and this outcome measure can be considered for use in future studies in this population.

Acknowledgements

This study was part funded by the Institute for Respiratory Health Glenn Brown Memorial Grant.

References

[1] Hogg M, Braithwaite M, Bailey M, Kotsimbos T, Wilson JW. Work disability in adults with cystic fibrosis and its relationship to quality of life. Journal of Cystic Fibrosis. 2007; 6: 223-7.

[2] Havermans T, Colpaert K, Vanharen L, Dupont LJ. Health related quality of life in cystic fibrosis: To work or not to work? Journal of Cystic Fibrosis. 2009; 8: 218-23.

[3] Cystic Fibrosis Australia. Cystic Fibrosis in Australia 2014 17th Annual Report Australian Cystic Fibrosis Data Registry. North Ryde: Cystic Fibrosis Australia. 2016. Accessed on November 26 2016 from https://www.cysticfibrosis.org.au/media/wysiwyg/CF-Australia/medical-documents/CFA_DataRegistryReport_2014_Final.pdf

[4] Targett K, Bourke S, Nash E, Murphy E, Ayres J, Devereux G. Employment in adults with cystic fibrosis. Occupational Medicine. 2014; 64(2):87-94.

[5] Frangolias DD, Holloway CL, Vedal S, Wilcox PG. Role of Exercise and Lung Function in Predicting Work Status in Cystic Fibrosis. American Journal of Respiratory and Critical Care Medicine. 2003; 167(2):150-157.

[6] Kessler CR, Barber DC, Beck BA, Berglund BP, Cleary BP, McKenas BD, et al. The World Health Organization Health and Work Performance Questionnaire (HPQ). Journal of Occupational and Environmental Medicine. 2003; 45: 156-74.

SIR CHARLES GAIRNDER HOSPITAL TELEHEALTH CF CLINICS: PARTICIPANT SATISFACTION SURVEY & TELEHEALTH SATISFACTION SCALE

These 2 short surveys are designed to gather and evaluate your feedback on the new telehealth CF clinics that have been implemented as part of your outpatient care.

Please provide your feedback on the following items:

1) The organisation of the tele	health clinics met	my expectations		
Strongly Agree Disagree	Agree	Unsure	Disagree	Strongly
Comments:				
2) The timing of telehealth clin	ic appointments w	ere convenient for	me	
Strongly Agree Disagree	Agree	Unsure	Disagree	Strongly
Comments:				
3) The healthcare I received d	uring the telehealt	h clinics met my ex	pectations	
Strongly Agree Disagree	Agree	Unsure	Disagree	Strongly
Comments:				
4) I was able to communicate	effectively with the	e CF team using this	s technology	
Strongly Agree Disagree	Agree	Unsure	Disagree	Strongly
Comments:				
5) I prefer telehealth clinics ov	er traditional 'face	to face' clinics		
Strongly Agree Disagree	Agree	Unsure	Disagree	Strongly
Comments:				
6) Telehealth clinics are a goo	d way to manage	my CF care		
Strongly Agree Disagree	Agree	Unsure	Disagree	Strongly
Comments:				
7) Would you like to add any c	other comments at	oout your experienc	e with CF telehealt	h clinics?

The Telehealth Satisfaction Scale - overall, how satisfied were you with:

	Excellent	Good	Poor/Fair	
Comments:				
2) The visual quality of the	e equipment			
	Excellent	Good	Poor/Fair	
Comments:				
3) Your personal comfort	using the telehealth system			
	Excellent	Good	Poor/Fair	
Comments:				
4) The ease of getting to t	he telehealth department			
	Excellent	Good	Poor/Fair	
Comments:				
5) The length of time with	the CF team			
	Excellent	Good	Poor/Fair	
Comments:				
6) The explanation of your	r treatment by the CF team			
	Excellent	Good	Poor/Fair	
Comments:				
7) The thoroughness, care	efulness and skilfulness of t	he CF team		
	Excellent	Good	Poor/Fair	
Comments:				
8) The courtesy, respect, sensitivity, and friendliness of the CF team				
	Excellent	Good	Poor/Fair	
Comments:				
9) How well your privacy was respected				
	Excellent	Good	Poor/Fair	
Comments:				

1) The voice quality of the equipment

Appendix 5

Original article

High usability of a smartphone application for reporting symptoms in adults with cystic fibrosis Jamie Wood^{1,2,3}, Sue Jenkins^{1,2,3}, David Putrino^{4,5}, Siobhain Mulrennan^{3,6}, Sue Morey^{3,6}, Nola Cecins², Kylie Hill^{1,3}.

 ¹School of Physiotherapy and Exercise Science, Curtin University, Perth, Western Australia
 ²Physiotherapy Department, Sir Charles Gairdner Hospital, Perth, Western Australia
 ³Institute for Respiratory Health, Sir Charles Gairdner Hospital, Perth, Western Australia
 ⁴Department of Telemedicine and Virtual Rehabilitation, Burke Institute of Medical Research, White Plains, New York

⁵Department of Rehabilitation Medicine, Weill Cornell Medicine, New York, New York

⁶Department of Respiratory Medicine, Sir Charles Gairdner Hospital, Perth, Western Australia

Corresponding author:

Jamie Wood, Physiotherapy Department, A block, Sir Charles Gairdner Hospital, Hospital Avenue, Nedlands, Western Australia, Australia, 6009

Email: jamie.wood@health.wa.gov.au

Key Words: Cystic fibrosis; exacerbations; smartphone application; telemedicine; telehealth

Main body word count: 2,464

Abstract word count: 250

Abstract

Introduction/Aim: In cystic fibrosis (CF), exacerbations impair lung function and health-related quality of life, increase healthcare costs and reduce survival. Delayed reporting of worsening symptoms can result in more severe exacerbations and worse clinical outcomes; therefore there is a need for a novel approach to facilitate the early identification and treatment of exacerbations in this population. This study investigated the usability of a smartphone application to report symptoms in adults with CF, and the observer agreement in clinical decision making between senior clinicians interpreting smartphone application responses.

Methods: Adults with CF used the smartphone application weekly for four weeks. The application comprised 10 yes/no questions regarding respiratory symptoms and two regarding emotional wellbeing. Usability was measured with the System Usability Scale (SUS); Observer agreement was tested by providing a CF physician and a nurse practitioner with 45 clinical scenarios. For each scenario the clinicians, who were blinded to each other's responses, were asked to indicate whether or not they would: (i) initiate telephone contact, and/or (ii) request a clinic visit for the individual.

Results: Ten participants (5 female), aged mean (SD) 33 (11) years, FEV₁ 49 (27)% predicted completed the study. The mean (SD) SUS score was 94 (6). There was perfect agreement between clinicians for initiating contact with the participant (κ =1.0, p<0.001), and near-perfect for requesting a clinic visit (κ =0.86, p<0.001).

Conclusion: The use of a smartphone application for reporting symptoms in adults with CF has excellent usability and near-perfect agreement between senior clinicians when interpreting the application responses.

Introduction

Cystic fibrosis (CF) is the most common lethal genetic disease affecting Caucasians, occurring in approximately one in 3,600 live births¹. Although it is a multi-system disease, respiratory failure in those with poor lung function is the most common cause of mortality². Exacerbations are an important contributor to the decline in lung function^{3, 4} with approximately 25% of individuals who require intravenous antibiotics (IVAB) for an exacerbation failing to return to their pre-exacerbation (i.e. baseline) level of lung function following initial treatment^{5, 6}. Of note, more than half (58%) of these patients still do not reach their previous baseline level of lung function after 12 months⁵. Earlier work has demonstrated that recovery of lung function following an exacerbation is dependent, at least in part, on the time between exacerbation onset and the initiation of treatment. Specifically, the number of days elapsed prior to commencing IVAB after the diagnosis of an exacerbation is a predictor of lung function outcomes after treatment cessation^{5,7}. These studies reveal that early detection and treatment of exacerbations are of critical importance in the clinical care in this population. Nevertheless, it is well-established that the identification of exacerbations can be delayed in individuals with CF as they may not attend a CF clinic or seek medical advice as soon as they develop worsening symptoms. The cause of this delayed presentation is multi-factorial and includes competing issues such as study, work or family commitments, the travel and financial burden involved in attending a CF clinic, a lack of understanding of the disease process, or feelings of anxiety and depression^{8, 9}.

There is a need for a novel approach to facilitate the early identification and treatment of exacerbations in this population. For this reason, we have developed a smartphone application that can be used by adults with CF to immediately report symptoms suggestive of an exacerbation to the CF team.

The aim of this study was to determine the: (i) usability and satisfaction of this smartphone application to report symptoms suggestive of an exacerbation in adults with CF and, (ii) observer agreement between two senior clinicians in clinical decision making based on the symptom responses obtained using this technology.

Methods

Study design

This was a single group usability study of four weeks duration, conducted at the adult CF centre at Sir Charles Gairdner Hospital. Adults with CF who owned a smartphone with iOS or Android operating systems were invited to participate at routine outpatient clinic appointments, with recruitment targeted to ensure equal numbers of male and female participants. In total, 190 adults with CF attend the centre, and at the time of recruitment 183 (96%) owned a smartphone. Approval was obtained from the Human Research Ethics Committees for Sir Charles Gairdner Group and Curtin University.

Smartphone application

Experienced CF clinicians who were investigators in this study formulated the twelve questions used in the smartphone application. Members of the research team reviewed the questions for face validity. Ten of the respiratory symptom questions were derived from exacerbation scales previously used in CF research^{10, 11} (Table 1) and two questions (i.e. those pertaining to wheeze and chest tightness) were added as these symptoms have been identified by adults with CF as indicators of an exacerbation¹². Finally, two questions pertaining to emotional wellbeing were included, because in individuals with CF, greater feelings of anxiety and depression have been associated with the greater respiratory symptoms, as well as poorer lung function, HRQoL and physical functioning¹³. All smartphone application questions were answered yes or no, and each time an answer was selected the application automatically moved to the next question. The font size of the text in the application was set at a minimum of 18 to ensure all users could read the questions.

181

A prototype of the application was designed by a smartphone application engineer, and internal testing was then undertaken by several members of the research team to ensure the application was functioning as intended. The final version of the application was then made available free of charge on iTunes and the Google Play Store, allowing participants to install it on their smartphone after enrolment in the study.

Following enrolment in the study and installation of the application, participants were required to answer the questions once a week over four consecutive weeks. Four occasions of use was considered sufficient by the research team, given another study that explored the usability of a telehealth system in CF based their conclusion on a single use¹⁴. Participants in the current study were asked to set a recurring alert on their smartphone, which prompted them to answer the questions on a pre-specified day and time each week of their choosing. To login securely, each participant had a unique four digit user code. Once all questions were answered, the responses and user code were emailed in binary code (1 = yes, 0 = no), along with the login used to generate each set of responses, to an account established for the purpose of the study. To maintain data security, only the principal investigator could identify each participant based on their user code. Participants were informed that for the purpose of this study, symptom responses were not being used by the CF team to detect exacerbations and were instructed to contact the CF clinic by phone if they required an appointment. Participants were not reviewed in the CF clinic during the four-week period, and therefore no comparison was undertaken regarding the smartphone application answers and clinical status.

Outcomes

System usability

Usability was measured at the end of the intervention period with the System Usability Scale (SUS)¹⁵. This scale comprises 10 items that ask about the usability aspects of the new technology, such as its complexity and functionality, and the participant's confidence in using it. Each item was answered using a 5 point Likert scale with responses that ranged from 5 (strongly agree) to 1 (strongly disagree). Scores from each of the 10 items were scaled to provide a score out of four, then combined and multiplied by 2.5 to provide a total score out of 100. Scores from the SUS were also grouped into two subscales: usability (8 items) and learnability (two items), both of which were also scored out of 100. The SUS has been widely used in the information technology industry, and more recently in studies evaluating the usability of telehealth technologies used in CF¹⁴ and chronic obstructive pulmonary disease¹⁶.

Participant satisfaction

Participant satisfaction was assessed at the end of the intervention period using a purpose-developed survey which comprised seven items relating to the design and use of the smartphone application. Items alternated between positive and negative wording, and were answered with a 5-point Likert scale with responses that ranged from 5 (strongly agree) to 1 (strongly disagree).

Observer agreement

A physician and nurse practitioner who specialise in CF were presented with an identical set of 45 clinical scenarios. A nurse practitioner was included to reflect the growing utilisation of this role as a first point of contact in CF centres internationally. These scenarios were created by pairing one of three case studies, with one of 15 different combinations of smartphone application responses. The three case studies described clinical presentations of an adult with mild, moderate or severe CF and were reviewed for authenticity by a respiratory physician with extensive experience in CF, who was not otherwise involved in the study. For each scenario, the two clinicians were asked to indicate whether or not they would; (i) initiate telephone contact to discuss symptoms further, and/or (ii) request the individual attend an outpatient clinic for further assessment. The clinicians were blinded to each other's responses.

183

Data analysis

Statistical analyses were conducted using SPSS® (Statistical Package for Social Sciences, version 22.0, IBM corp., New York, USA). System Usability Scale data were reported using descriptive statistics. Observer agreement between clinicians was assessed using a Cohen's kappa statistic. A sample size of 45 responses was adequate to detect a difference in proportion between chance and anticipated agreement of 0.5, assuming a relative error of 30%.

Results

Ten participants with CF completed the study (Table 2). Each participant used the smartphone application once per week over the four-week period. Seven (70%) participants used iPhones, and three (30%) used phones with Android operating systems.

System usability

The mean (SD) raw SUS score was 94 (6) (Table 3). When summed as subscales, participants scored the smartphone application as 92 (7) and 100 (0) for usability and learnability, respectively.

Participant satisfaction

Nine (90%) participants responded that the smartphone application was a good method for reporting symptoms to the CF team. All participants (100%) found the instructions in the application easy to follow (Table 4).

Observer agreement

Observer agreement between clinicians was perfect for the question regarding whether or not to initiate contact with the patient to further discuss symptoms (κ =1.0, p<0.001). Specifically, both clinicians indicated that they would initiate contact for 42 (93%) scenarios, but not for three (7%) scenarios. There was near-perfect agreement for whether or not the patient should visit the outpatient clinic (κ =0.86, p<0.001). Specifically, both clinicians indicated that they would request the patient visit the outpatient clinic for 24 (53%) scenarios, but not for 18 (40%) scenarios. Clinicians disagreed on three (7%) scenarios, which was for the same scenario across all three case studies. This asked "in the past week have you had increased cough or new pain on coughing?"

Discussion

This study has shown that a smartphone application that can be used to report symptoms that may be suggestive of an exacerbation has excellent usability in adults with CF. Furthermore the symptom responses from the application were interpreted in a similar way by different experienced CF clinicians in the process of clinical decision making.

The integration of telehealth with usual care in CF is an expanding area of research. To date, feasibility has been demonstrated in interventions such as using Skype and WhatsApp to communicate with patients¹⁷, videoconferencing to deliver routine outpatient clinics¹⁸, monitoring of lung function and symptoms at home¹⁹⁻²¹, and assessing exercise capacity within the patient's home¹⁴. While the evidence supporting the role of telehealth in CF is relatively limited, the aim of such interventions in the longer term will be to improve health outcomes via increased treatment adherence and the earlier detection of exacerbations. Strategies similar to those used in CF have also been employed in chronic diseases such as type 1 diabetes²², with demonstrated improvements in access to routine care. A recent systematic review²³ into mobile health interventions in diabetes mellitus and other cardiovascular and respiratory diseases found mixed evidence supporting the effectiveness of these interventions on treatment adherence and health outcomes.
The use of out-of-hospital monitoring systems to detect changes in symptoms and physiological parameters suggestive of an exacerbation is a growing area of interest¹⁹⁻²¹. One shortcoming of these previous studies was poor adherence to the monitoring system, which resulted, at least in part, from the equipment being large and bulky. The smartphone application described in this study overcame this limitation, as it was small and portable, and integrated easily with the lifestyle of adults with CF. Further, in contrast with earlier work which required daily use of the monitoring system by all participants, those in the current study were asked to use the system less frequently (i.e. once a week); a factor which may also have contributed to the excellent adherence. While the optimal frequency of reporting in order to detect an exacerbation is unknown, daily reporting in other studies likely contributed to the observed poor adherence. Therefore weekly reporting appears to be a pragmatic alternative which was well tolerated. In clinical practice, to minimise the possibility of missing the onset of an exacerbation that occurred between weekly use of the smartphone application, patients could be instructed to use the smartphone application anytime between scheduled uses if they felt their symptoms had worsened.

The high SUS scores indicate that the smartphone application achieved an 'A' rating for usability. This rating is at the top of the "acceptable" usability range, and is described on the SUS adjective rating scale as "best imaginable" usability²⁴. In addition, raw scores of over 90 on the SUS indicate that a system is in the 99th percentile of usability for the tested demographic²⁴. Finally, this score also indicates that users would be more likely to recommend this technology to a friend²⁴. The SUS scores observed in this study are similar to those reported for interventions in adults with CF investigating the feasibility of an internet based tool to promote physical activity²⁵, and using a videoconference system to assess exercise capacity¹⁴. This was not surprising given the uncomplicated design and functionality of the smartphone application, including asking for yes/no responses to questions and requiring minimal time to answer the questions (less than two minutes). Similarly, participant satisfaction, as measured using the purpose developed survey was very high. All of the participants owned smartphones prior to the study, and therefore the high level of usability and satisfaction observed may be a function of their pre-existing comfort with technology. This was expected, as 74% of adults in Australia owned a smartphone in 2015²⁶.

We observed near-perfect agreement in clinical decision making between experienced CF clinicians. This suggests that others will be able to use this application to arrive at similar clinical decisions regarding whether the patient requires further assessment for a potential exacerbation, however having only two clinicians participating in the observer agreement study is a limitation. It is possible that for the question with poor agreement, the ambiguity related to asking the clinician to interpret responses regarding two different symptoms (increased cough and new pain on coughing). This ambiguity might be overcome by dividing this into two separate questions: one regarding increased cough and a second regarding chest pain. This will allow patients to report chest pain irrespective of whether it is related to coughing.

This system, while designed to detect exacerbations, did not include spirometry. This is a potential limitation as FEV₁ is regarded as one of the key measures of clinical status in CF²⁷. However the provision of a spirometer would have greatly increased cost and reduced the portability of the reporting system. We aimed to develop a system that required no additional equipment to the participant's own smartphone, allowing us to achieve high usability. This system, which was based entirely on patient reported symptoms, will allow the CF team to determine whether spirometry is required. We do acknowledge however, that the inclusion of home spirometry as part of our system may provide more accurate diagnosis of an exacerbation before the patient needed to attend the CF clinic."

Although the sample size for this study was modest, participants were recruited across a broad range of ages and severity of lung function, and there was equal representation from both genders. Participants living in rural and remote areas (n=2) also participated in the study. Seven (70%) had treatment for an exacerbation in the 12 months prior to commencing the study. Participants were employed in a wide variety of occupations, including administration, retail, information technology and healthcare, and one was a fulltime university student. This allowed the investigators to gather usability data from people who were representative of the broader adult CF population.

The results of this study have demonstrated that amongst adults with CF, a smartphone application used to report symptoms suggestive of an exacerbation has excellent usability, and provides symptom responses that can be interpreted consistently between clinicians who are experienced in the management of this clinical population. A randomised controlled trial is now warranted to investigate whether this system can reduce the delay in time taken by a patient to report symptoms to a CF clinic, and measure the impact on health-related outcomes in adults with CF.

Acknowledgements

We would like to acknowledge the participants in this study. This study received funding from the Sir Charles Gairdner Group Research Advisory Council.

Declaration of conflicting interests

Mr. Wood reports grants from Sir Charles Gairdner Group Research Advisory Council, during the conduct of the study; grants from Technipro Pulmomed, non-financial support from Novartis, non-financial support from Pharmaxis, outside the submitted work. A/Prof Jenkins reports grants from Sir Charles Gairdner Group Research Advisory Council, during the conduct of the study. A/Prof Putrino has nothing to disclose. Clin A/Prof Mulrennan reports grants from Sir Charles Gairdner Group Research Advisory Council, during the study; personal fees from Vertex, non-financial support from Novartis, outside the submitted work. Mrs. Morey has nothing to disclose. Mrs. Cecins has nothing to disclose. A/Prof Hill has nothing to disclose.

References

Australia CF. Cystic Fibrosis in Australia 2014 17th Annual Report: Australian Cystic
 Fibrosis Data Registry. North Ryde 2016. https://www.cysticfibrosis.org.au/media/wysiwyg/CF Australia/medical-documents/CFA_DataRegistryReport_2014_Final.pdf. Accessed 17th January 2017.

2. Bell SC, Bye PT, Cooper PJ, et al. Cystic fibrosis in Australia, 2009: results from a data registry. The Medical journal of Australia. 2011; 195: 396-400.

3. Britto MT, Kotagal UR, Hornung RW, Atherton HD, Tsevat J and Wilmott RW. Impact of recent pulmonary exacerbations on quality of life in patients with cystic fibrosis. Chest. 2002; 121: 64-72.

4. Ward N, White D, Rowe H, Stiller K and Sullivan T. Physical activity levels of patients with cystic fibrosis hospitalised with an acute respiratory exacerbation. Respiratory medicine. 2013; 107: 1014-20.

5. Sanders DB, Bittner RC, Rosenfeld M, Hoffman LR, Redding GJ and Goss CH. Failure to recover to baseline pulmonary function after cystic fibrosis pulmonary exacerbation. American journal of respiratory and critical care medicine. 2010; 182: 627-32.

6. Sanders DB, Bittner RC, Rosenfeld M, Redding GJ and Goss CH. Pulmonary exacerbations are associated with subsequent FEV1 decline in both adults and children with cystic fibrosis. Pediatric pulmonology. 2011; 46: 393-400.

7. Smith AL, Fiel SB, Mayer-Hamblett N, Ramsey B and Burns JL. Susceptibility testing of Pseudomonas aeruginosa isolates and clinical response to parenteral antibiotic administration: lack of association in cystic fibrosis. Chest. 2003; 123: 1495-502.

8. Berlinski A, Chambers MJ, Willis L, Homa K and Com G. Redesigning care to meet national recommendation of four or more yearly clinic visits in patients with cystic fibrosis. BMJ quality & safety. 2014; 23 Suppl 1: i42-9.

9. Zemanick ET, Harris JK, Conway S, et al. Measuring and improving respiratory outcomes in cystic fibrosis lung disease: opportunities and challenges to therapy. Journal of cystic fibrosis : official journal of the European Cystic Fibrosis Society. 2010; 9: 1-16.

 Bilton D, Canny G, Conway S, et al. Pulmonary exacerbation: towards a definition for use in clinical trials. Report from the EuroCareCF Working Group on outcome parameters in clinical trials.
 Journal of cystic fibrosis : official journal of the European Cystic Fibrosis Society. 2011; 10 Suppl 2: S79-81.

11. Fuchs HJ, Borowitz DS, Christiansen DH, et al. Effect of aerosolized recombinant human DNase on exacerbations of respiratory symptoms and on pulmonary function in patients with cystic fibrosis. The Pulmozyme Study Group. The New England journal of medicine. 1994; 331: 637-42.

12. Goss CH, Edwards TC, Ramsey BW, Aitken ML and Patrick DL. Patient-reported respiratory symptoms in cystic fibrosis. Journal of cystic fibrosis : official journal of the European Cystic Fibrosis Society. 2009; 8: 245-52.

13. Yohannes AM, Willgoss TG, Fatoye FA, Dip MD and Webb K. Relationship between
anxiety, depression, and quality of life in adult patients with cystic fibrosis. Respiratory care. 2012;
57: 550-6.

14. Cox NS, Alison JA, Button BM, Wilson JW and Holland AE. Assessing exercise capacity using telehealth: a feasibility study in adults with cystic fibrosis. Respiratory care. 2013; 58: 286-90.

Brooke J. SUS: A "quick and dirty" usability scale. In: P. W. Jordan BT, B. A.
 Weerdmeester, & A. L. McClelland, (ed.). Usability Evaluation in Industry. London: Taylor and
 Francis., 1996, p. 189-94.

16. Holland A. Telehealth reduces hospital admission rates in patients with COPD. Journal of physiotherapy. 2013; 59: 129.

17. Gur M, Nir V, Teleshov A, et al. The use of telehealth (text messaging and video communications) in patients with cystic fibrosis: A pilot study. J Telemed Telecare. 2017; 23: 489-93.

18. Wood J, Mulrennan S, Hill K, Cecins N, Morey S and Jenkins S. Telehealth clinics increase access to care for adults with cystic fibrosis living in rural and remote Western Australia. J Telemed Telecare. 2016.

19. Bella S, Cotognini C, Alghisi F and Montemitro E. Program of home telemonitoring in patients with cystic fibrosis over a period of 2 years: a contribution to the rationalization of care. La Clinica terapeutica. 2013; 164: e313-7.

20. Grzincich G, Gagliardini R, Bossi A, et al. Evaluation of a home telemonitoring service for adult patients with cystic fibrosis: a pilot study. J Telemed Telecare. 2010; 16: 359-62.

21. Jarad NA and Sund ZM. Telemonitoring in chronic obstructive airway disease and adult patients with cystic fibrosis. J Telemed Telecare. 2011; 17: 127-32.

22. Wood CL, Clements SA, McFann K, Slover R, Thomas JF and Wadwa RP. Use of Telemedicine to Improve Adherence to American Diabetes Association Standards in Pediatric Type 1 Diabetes. Diabetes Technol Ther. 2016; 18: 7-14.

23. Hamine S, Gerth-Guyette E, Faulx D, Green BB and Ginsburg AS. Impact of mHealth chronic disease management on treatment adherence and patient outcomes: a systematic review. J Med Internet Res. 2015; 17: e52.

24. Sauro J. A Practical Guide to the System Usability Scale: Background, Benchmarks & Best Practices. Denver CO2011.

25. Cox NS, Alison JA, Button BM, Wilson JW and Holland AE. Feasibility and acceptability of an internet-based program to promote physical activity in adults with cystic fibrosis. Respiratory care. 2015; 60: 422-9.

26. Government A. Communications report 2014-15. In: Authority ACaM, (ed.). Pyrmont2015,

p. 3.

27. Davies JC and Alton EW. Monitoring respiratory disease severity in cystic fibrosis.

Respiratory care. 2009; 54: 606-17.

Smartphone application questions.

In the past week, have you had:

An increase in or sputum volume or change in colour?

New or increased blood in your sputum?

Increased cough, or new pain on coughing?

New or increased wheeze?

New or increased chest tightness?

Increased shortness of breath or difficulty breathing?

Increased fatigue or lethargy?

Fever?

Loss of appetite or weight?

Sinus pain or tenderness?

In the past week, have you felt:

Low in mood?

Worried?

Participant (n = 10) characteristics.

Male, n (%)	5 (50)
Age, yr	32 (11)
BMI, kg/m^2	23 (5)
FEV ₁ , % predicted	49 (27)
Location, n (%)	
Metropolitan	8 (80)
Rural or remote	2 (20)

Data are presented as number (%) or mean (SD). BMI = body mass index; FEV_1 = forced expiratory

volume in one second.

i diticipant (n=10) responses to items of the bystem osubility beater	Participant (n=1	0) responses	to items	of the Syster	n Usability Scale
---	------------------	--------------	----------	---------------	-------------------

Item	SD				SA
I think that I would like to use this system frequently	0	0	10	40	50
I found the system unnecessarily complex	90	10	0	0	0
I thought the system was easy to use	0	0	0	10	90
I think that I would need the support of a technical person to be able to use this system	100	0	0	0	0
I found the various functions in this system were well integrated	0	0	30	30	40
I thought there was too much inconsistency in this system	70	20	0	10	0
I would imagine that most people would learn to use this system very quickly	0	0	0	0	100
I found the system very cumbersome to use	80	20	0	0	0
I felt very confident using the system	0	0	10	0	90
I needed to learn a lot of things before I could get going with this system	100	0	0	0	0

Data are presented as percentage of participants. Each item is answered using a 5-point Likert scale from strongly disagree (SD) to strongly agree (SA).

SA	А	U	D	SD
0	80	20	0	0
80	20	0	0	0
0	0	0	60	40
50	30	0	20	0
0	10	0	50	40
40	50	10	0	0
0	0	0	30	70
	SA 0 80 0 50 0 40 0	SA A 0 80 80 20 0 0 50 30 0 10 40 50 0 0	SA A U 0 80 20 80 20 0 0 0 0 50 30 0 0 10 0 40 50 10 0 0 0	SA A U D 0 80 20 0 80 20 0 0 80 20 0 0 0 0 0 0 0 0 0 60 50 30 0 20 0 10 0 50 40 50 10 0 0 0 0 30

Participant (n = 10) responses to the purpose developed satisfaction survey.

Data are presented as percentage of participants. SA = Strongly Agree, A = Agree, U = Unsure, D = Disagree,

SD = Strongly Disagree

Case Study 3

- 29 year old female with CF
- Lives in Bunbury, married, works part time, one daughter
- Best FEV₁ in past 12 months = 1.9L (61% predicted)
- Sputum: Pseudomonas aeruginosa
- Pancreatic insufficient, osteopenia
- Meds: Creon, Pulmozyme, HTS, VitABDECK
- Last course of IVABs = 7 months ago

Scenario 3

In the past week have you had:	
An increase in sputum volume or change in colour	No
New or increased blood in your sputum	No
Increased cough, or new pain on coughing	No
New or increased wheeze	Yes
New or increased chest tightness	No
Increased shortness of breath or difficulty breathing	No
Increased fatigue or lethargy	No
Fever	No
Loss of appetite or weight	No
Sinus pain or tenderness	Yes
In the past week have you:	
Felt low in mood	No
Felt worried	No

Would you initiate phone contact with this person? YES NO

CONSORT diagram



Protocol

A smartphone application for reporting symptoms in adults with cystic fibrosis: protocol of a randomised controlled trial

Jamie Wood^{1,2,3}, Sue Jenkins^{1,2,3}, David Putrino^{4,5}, Siobhain Mulrennan^{3,6}, Sue Morey^{3,6}, Nola Cecins², Kylie Hill^{1,3}.

¹School of Physiotherapy and Exercise Science, Curtin University, Perth, Western Australia

²Physiotherapy Department, Sir Charles Gairdner Hospital, Perth, Western Australia

³Institute for Respiratory Health, Sir Charles Gairdner Hospital, Perth, Western Australia

⁴Abilities Research Center, Mount Sinai Health System, New York, New York

⁵Department of Rehabilitation Medicine, Icahn School of Medicine at Mount Sinai, New York, New York

⁶Department of Respiratory Medicine, Sir Charles Gairdner Hospital, Perth, Western Australia

Corresponding author:

Jamie Wood, Physiotherapy Department, A block, Sir Charles Gairdner Hospital, Hospital Avenue, Nedlands, Perth, Western Australia, Australia, 6009

Email: jamie.wood@health.wa.gov.au

Keywords: Cystic fibrosis; Exacerbation; Smartphone; Telehealth; Telemedicine

ABSTRACT

Introduction: In people with cystic fibrosis (CF), exacerbations have been shown to have profound and prolonged negative effects such as reducing physical activity and health-related quality of life, increasing the rate of decline of lung function and healthcare costs, and ultimately increasing the risk of mortality. Delayed initiation of treatment following the signs of an exacerbation has been shown to be associated with failure to recover to baseline. Therefore, the late identification and treatment of an exacerbation due to delayed presentation will potentially worsen short and long-term outcomes. We have developed a smartphone application, containing questions which require yes or no responses relating to symptoms suggestive of a respiratory exacerbation, and allow the CF team to initiate treatment sooner, thereby potentially reducing the risk of severe exacerbations which require intravenous antibiotics (IVAB) and often a hospital admission.

Methods: We will undertake a randomised controlled trial (RCT). Sixty adults with CF will be recruited and randomised to either the intervention or control group. The intervention group will use the smartphone application weekly for 12 months, or earlier than the next weekly reporting time if they feel their symptoms have worsened. The control group will continue to receive usual care, involving regular (approximately 3-monthly) CF outpatient clinic appointments. The primary outcome measure will be courses and days of IVAB.

Ethics and dissemination: Approval was obtained from the Sir Charles Gairdner Group Human Research Ethics Committee for WA Health (2015 - 030), and Curtin University Human Research Ethics Committee (HR212/2015), and has been registered with the Australian and New Zealand Clinical Trials Registry (ACTRN12615000599572). Results of this study will be presented at international conferences and published in peer-reviewed journals in accordance with the CONSORT (Consolidated Standards of Reporting Trials) statement.

Strengths and limitations of this study

This is the first long term randomised controlled trial investigating a smartphone application for reporting symptoms in adults with cystic fibrosis.

The pharmacist collecting primary outcome data, and the physicians prescribing intravenous antibiotics, are blinded to the participants' group allocation.

Selecting questions to include in the smartphone application was difficult due to a lack of consensus of what defines an exacerbation in cystic fibrosis.

It was not feasible to include spirometry in the intervention in this study which may improve clinical interpretation of reported symptoms.

INTRODUCTION

Cystic fibrosis (CF) is a disease characterised by periods of clinical stability, interspersed with periods of acute deterioration, known as exacerbations. Respiratory exacerbations have been shown to have profound and prolonged negative effects on people with CF, such as reducing physical activity¹ and health-related quality of life (HRQoL),² increasing the rate of decline of lung function³ and healthcare costs,⁴ and ultimately increasing the risk of mortality.⁵ The decline in lung function may also contribute to increased feelings of depression.⁶ In people with CF, an exacerbation has been defined as a change in four or more disease-specific criteria comprising cough, sputum, haemoptysis, dyspnoea, fatigue, sinus pain and discharge, fever, worsening airflow obstruction (i.e. reduced forced expiratory volume in one second [FEV₁]), anorexia or weight loss, and changes in physical examination of the chest or chest radiography.⁷

Of concern, of those individuals who experience an exacerbation requiring intravenous antibiotics (IVAB), approximately 25% have a persistent reduction in FEV₁ at the end of their treatment,⁸ and more than half (58%) of these individuals do not regain their pre-exacerbation level of FEV₁ after 12 months.⁸ Delayed initiation of treatment has been shown to be associated with this failure to recover to baseline.⁸ Therefore, the late identification and treatment of an exacerbation due to delayed presentation will potentially worsen short and long-term outcomes. People with CF may delay their presentation to a CF clinic for management of an exacerbation due to study, work or family commitments, the travel and financial burden involved in attending a CF clinic, a lack of understanding of the disease process, or if affected by anxiety and depression.⁹

Telehealth has been evaluated in studies in people with CF using interventions such as monitoring of spirometry,¹⁰⁻¹² oxygen saturation^{10 13} and symptoms from home,¹⁰⁻¹² using videoconferencing to measure exercise capacity¹⁴ and provide health assessment in the home,¹⁵ and to provide routine outpatient clinics.¹⁶ Although telehealth, in the form of home monitoring, appears to be feasible and readily accepted by people with CF,¹⁷ to date, studies have not demonstrated a significant impact of telehealth on health outcomes such as exacerbation rates, lung function, HRQoL and healthcare utilisation (HCU). One reason for this may relate to the burden associated with the telehealth interventions. That is, earlier work often required participants to record measurements as often as daily and did not offer the opportunity to complete these measures outside the home (i.e. there was no use of mobile technology), which compromised participant adherence to the telehealth intervention.

We aim to overcome this shortcoming by investigating the effects of using of an application for reporting symptoms that is completely mobile and stored on the participant's own smartphone. This novel approach is intended to facilitate the early identification of symptoms suggestive of a respiratory exacerbation, and allow the CF team to initiate treatment sooner in the form of oral and inhaled antibiotics, and increased mucolytic therapy and airway clearance. This may potentially reduce the risk of severe exacerbations which require IVAB and often a hospital admission.

METHODS AND ANALYSIS

This will be a prospective 12-month, single blinded randomised controlled trial (RCT).

Participant screening and selection

Eligible participants will be identified by the principal investigator and recruited from routine outpatient clinic appointments at the adult CF centre at Sir Charles Gairdner Hospital (SCGH), Perth. The Patient Information and Consent Form will be provided to potential participants, who will then be contacted by the principal investigator approximately 48 hours later to discuss their willingness to participate.

Inclusion criteria

Adults who meet the following criteria will be eligible to participate: diagnosis of CF; aged ≥ 18 years; under the care of the SCGH CF team; respiratory exacerbation requiring treatment with IVAB in the preceding 12 months; currently in a period of clinical stability, defined as no signs of an exacerbation for the previous 2 weeks; and able to understand written and spoken English.

Exclusion criteria

Criteria to exclude individuals from participating will be: previous lung transplantation or current listing for lung transplantation; inability to undertake a prescribed treatment regimen, or inability to use or not being in possession of a smartphone.

Randomisation and allocation concealment

Participants will be randomly allocated to the intervention or control group using the University of Sydney National Health Medical Research Council Clinical Trials Centre randomisation service. Recruitment will be stratified according to gender, location (metropolitan versus rural and remote) and whether the participant is prescribed Ivacaftor or Lumicaftor/Ivacaftor, as these medications can reduce the rate of exacerbations and improve lung function, weight and HRQoL.^{18 19}

Study measurements

Participants will have assessments performed at baseline (prior to randomisation), and at 6 and 12 months, unless otherwise stated. Assessments will be performed when the participant is in a period of clinical stability, defined as no signs of an exacerbation for the previous 2 weeks. If the participant has signs of an exacerbation at their scheduled follow up time, the assessments will be completed at the subsequent visit when they are clinically stable. The principal investigator will perform all assessments during follow up visits at SCGH. For telehealth assessments, spirometry and weight will be assessed by a health professional local to the participant, with the remainder of the assessments completed by the principal investigator.

Primary Outcome

Number of courses of IVAB and IVAB days

Information regarding the use of IVAB will be obtained from the medical records of the participants by a pharmacist blinded to the study groups and verified against SCGH pharmacy dispensing records. This will be collected at the final assessment only.

Secondary Outcomes

Lung function (FEV₁)

Lung function will be measured using a Medgraphics USB spirometer (MGC Diagnostics, Minnesota, USA), or Easyone spirometer (ndd Medical Technologies, Massachusetts, USA). Each participant will perform up to eight maximal forced expiratory manoeuvres to ensure the two best attempts meet published standards.²⁰

HRQoL – Cystic Fibrosis Questionnaire – Revised (CFQ-R)

This self-complete questionnaire comprises 50 questions answered on a 4 point Likert scale, and covers 12 domains comprising physical, vitality, emotion, treatment burden, health, social, body, role, weight, respiratory, digestion and eating.²¹ The CFQ-R requires the user to recall information from the previous 2 weeks.

Feelings of anxiety and depression – Hospital Anxiety and Depression Scale (HADS)

The HADS²² contains 14 self-report questions, evenly divided into anxiety and depression scales (7 each). A 4-point Likert scale (0 to 3) is used to answer questions based on a recall period of 7 days.

Nutritional status - Body Mass Index (BMI)

Participants will have their height and weight measured, and BMI will be calculated.

Measures of HCU including time to first exacerbation requiring IVABs, oral and inhaled antibiotic use and cost, hospital admissions and cost, and number of CF clinic visits At the final assessment only, the participant's medical record will be reviewed to extract information pertaining to HCU. This includes: time (in days) from randomisation to the first exacerbation requiring IVAB; number of hospital admissions and days; number of courses and days of oral and inhaled antibiotics; number of CF clinic visits; estimated cost of IVAB and other antibiotic treatment; and, estimated cost of hospital admissions. Costs related to HCU will be estimated using information provided from the SCGH Pharmacy and Finance departments. All potential participants receive their care at the SCGH CF Centre or via shared care with an external respiratory physician (and not from a general practitioner). Prescribed antibiotic use will be well documented in the medical record, and participants will not be required to keep a diary of antibiotic use. Non-clinical costs relating to participants' travel, parking, time off work and telephone calls will not be recorded.

Medication adherence – Treatment Adherence Questionnaire – CF (TAQ-CF)

Adherence to routine CF treatments will be assessed using the TAQ-CF.²³ This measure has 12 treatment items and asks the user to report on the frequency of each treatment on a 7 point Likert scale (ranging from "not at all" to "3 or more times per day"), and the duration of treatment on a 6 point Likert scale ("0" to "25+" minutes). The TAQ-CF also asks the participant to report on the barriers to individual treatments.

Absenteeism/Presenteeism – World Health Organisation's Heath and Work Performance Questionnaire (HPQ)

Absenteeism is defined as being absent from work or study due to illness whereas presenteeism is defined as the loss of productivity associated with attending work or study when unwell.²⁴ This questionnaire uses the absenteeism and presenteeism questions of the HPQ.²⁵

System usability – System Usability Scale (SUS)

The SUS²⁶ is a validated assessment tool comprising 10 questions regarding the usability of technological systems, and is answered on a 5 point Likert scale. Other measures of system usability will include the time of day the smartphone application is used and how many times the questions were not completed.

Intervention

A smartphone application has been developed for use by participants to report symptoms on their own smartphones. The application contains 14 questions (Table 1) that are required to be answered 'yes' or 'no'. Twelve questions relate to respiratory symptoms; 10 questions are derived from the Fuchs exacerbation scale,⁷ with the addition of two symptoms (wheeze and chest tightness) commonly identified by adults with CF as indicators of an exacerbation²⁷ and considered important by the research team. The application also asks the participant to answer two questions relating to their feelings of anxiety and depression. The data will then be transmitted securely via password protected email to members of the research team. This smartphone application has demonstrated high system usability in people with CF, as well as good observer agreement between CF clinicians interpreting the data.²⁸

Experimental group

Participants will be asked to use the smartphone application once per week on a set day and time of their choosing, for a period of 12 months. A smartphone calendar alert will remind the participant to answer the symptom questions. Participants can also use the application earlier than the next weekly reporting time if they feel their symptoms have worsened. If the participant misses one of their weekly reporting times, they will be prompted to answer the application questions via text message. If the participant responds 'yes' to any of the symptom questions, the principal investigator will alert the CF nurse practitioner, who will then phone the participant to discuss their symptoms and whether treatment (i.e. oral or inhaled antibiotics) is required, in consultation with a CF physician blinded to the participant's study group. The CF nurse practitioner will also determine if a formal review in the

CF clinic is required. If the participant's respiratory disease has been stable for 3 months, they will attend a routine CF clinic appointment.

Control group

Participants in this group will continue to receive usual care, involving routine CF clinic appointments (approximately 3 monthly). Participants can contact the CF Nurse Practitioner via telephone sooner if there is a change in their clinical status.

Statistical analysis

Statistical analyses will be undertaken with Stata ® (StataCorp. 2015. Stata Statistical Software: Release 14. College Station, TX: StataCorp LP) in accordance with the intention-to-treat principle. The distribution of data will be assessed for normality. A p value of < 0.05 will be considered statistically significant. Negative binomial regression, and independent t-tests or Mann Whitney U tests will be used to determine differences between the two groups. Further analyses will include general linear models to adjust for possible covariates.

Number of courses of IVAB and number of IV antibiotic days

Summary statistics and frequencies will be provided for continuous and categorical variables. Descriptive plots will also be produced for all variables. Independent t-tests will be used to determine whether the number of courses of IVAB, and number of IVAB days, differs between the two groups. Further analyses will include general linear models to adjust for possible covariates including age, gender, BMI, Pseudomonas aeruginosa, pancreatic insufficiency and CF-related diabetes. Appropriate transformations will be conducted if the data is not normally distributed.

Other measurements

For all variables, a between-group analysis of these outcome measures will be performed using an independent t-test (parametric data) or Mann Whitney U test (non-parametric data). Within group

changes in outcome measures collected at each visit will be assessed using one-way repeated measures analysis of variance (parametric data) or Friedman's test (non-parametric data).

Sample size calculations

This is a study comparing participants who will use a smartphone application and those who will not. To investigate the difference in the number of IVAB courses between the two groups, assuming a within group standard deviation of 1.21 (based on IVAB usage at the SCGH CF centre during the two years prior to the study), a sample size of 23 in each group will be required to detect a difference of one exacerbation requiring IVAB per participant, with 80% power and a significance level of p < 0.05. To account for any loss to follow up or drop out, we will recruit an additional seven individuals per group for a total sample size of 60. No additional individuals are required to detect the same difference under non-parametric settings.

Patient and public involvement statement

The intervention used in this study was designed with input from participants in a pilot study conducted by the research team.²⁸ There was no further involvement in the development of this RCT by patients or the public.

DATA COLLECTION AND MANAGEMENT

Participant data will not contain any information that allows the person's identity to be known. All data stored electronically at SCGH will be kept on a password-protected computer or locked in a filing cabinet in the Department of Respiratory Medicine. Any printed information including signed consent forms will be kept in locked filing cabinets. Data will be kept for a minimum of 15 years. After this time, data collection forms will be shredded and electronic files will be deleted.

Data transmitted from the smartphone application will be transmitted securely via password-protected email to members of the research team at SCGH, and not kept in 'cloud' storage at any stage. After transmission from the participant's phone or tablet, the data will be automatically deleted, preventing any potential breach of privacy if the device is lost or stolen. The application will also request the user to enter a unique, 4-digit passcode before being able to access the application questions.

ETHICS

Approval was obtained from the Sir Charles Gairdner Group Human Research Ethics Committee for WA Health (2015 - 030), and Curtin University Human Research Ethics Committee (HR212/2015). This trial has been registered with the Australian and New Zealand Clinical Trials Registry (ACTRN12615000599572). Results of this study will be presented at international conferences and published in peer-reviewed journals in accordance with the CONSORT (Consolidated Standards of Reporting Trials) statement (Figure 1).²⁹

DISCUSSION

This will be the first RCT investigating the impact of a smartphone application used by people with CF to report symptoms suggestive of a respiratory exacerbation directly to the CF team. By reducing the delay often seen in the reporting of symptoms, and therefore facilitating the earlier provision of treatment, the severity of exacerbations and the need for IVAB and hospital admissions may be reduced.

Contributor ship statement

J Wood, S Jenkins, D Putrino, S Mulrennan, S Morey, N Cecins and K Hill assisted in the design of the study protocol. J Wood drafted the manuscript, which was also contributed to by S Jenkins, D Putrino, S Mulrennan, S Morey, N Cecins and K Hill. All authors read and approved the final version of the manuscript.

Competing interests

Mr. Wood reports grants from Sir Charles Gairdner Group Research Advisory Council, during the conduct of the study; grants from Technipro Pulmomed, non-financial support from Novartis, non-financial support from Pharmaxis, outside the submitted work. A/Prof Jenkins reports grants from Sir Charles Gairdner Group Research Advisory Council, during the conduct of the study. A/Prof Putrino has nothing to disclose. Clin A/Prof Mulrennan reports grants from Sir Charles Gairdner Group Research Advisory Council, during the study; personal fees from Vertex, non-financial support from Novartis, outside the submitted work. Mrs. Morey has nothing to disclose. Mrs. Cecins has nothing to disclose. A/Prof Hill has nothing to disclose.

Funding

This study has received funding support from the Sir Charles Gairdner Group Research Advisory Council Grant (2015-16/040). This work has also received in-kind funding from the Sir Charles Gairdner Hospital Physiotherapy Department and Department of Respiratory Medicine, and Curtin University.

Data sharing statement

There are currently no additional data available relating to this protocol.

Acknowledgements

N/A

Figure 1 legend

'Study Design'

References

 Ward N, White D, Rowe H, et al. Physical activity levels of patients with cystic fibrosis hospitalised with an acute respiratory exacerbation. Respiratory medicine 2013;107(7):1014-20. doi: 10.1016/j.rmed.2013.03.002

2. Britto MT, Kotagal UR, Hornung RW, et al. Impact of recent pulmonary exacerbations on quality of life in patients with cystic fibrosis. Chest 2002;121(1):64-72.

3. Sanders DB, Bittner RC, Rosenfeld M, et al. Pulmonary exacerbations are associated with subsequent FEV1 decline in both adults and children with cystic fibrosis. Pediatric pulmonology 2011;46(4):393-400. doi: 10.1002/ppul.21374

4. Robson M, Abbott J, Webb K, et al. A cost description of an adult cystic fibrosis unit and cost analyses of different categories of patients. Thorax 1992;47(9):684-9.

5. de Boer K, Vandemheen KL, Tullis E, et al. Exacerbation frequency and clinical outcomes in adult patients with cystic fibrosis. Thorax 2011;66(8):680-5. doi: 10.1136/thx.2011.161117

6. Riekert KA, Bartlett SJ, Boyle MP, et al. The association between depression, lung function, and health-related quality of life among adults with cystic fibrosis. Chest 2007;132(1):231-7. doi: 10.1378/chest.06-2474 [published Online First: 2007/07/13]

7. Fuchs HJ, Borowitz DS, Christiansen DH, et al. Effect of aerosolized recombinant human DNase on exacerbations of respiratory symptoms and on pulmonary function in patients with cystic fibrosis.
The Pulmozyme Study Group. The New England journal of medicine 1994;331(10):637-42. doi: 10.1056/NEJM199409083311003

 Sanders DB, Bittner RC, Rosenfeld M, et al. Failure to recover to baseline pulmonary function after cystic fibrosis pulmonary exacerbation. American journal of respiratory and critical care medicine 2010;182(5):627-32. doi: 10.1164/rccm.200909-1421OC

9. Zemanick ET, Harris JK, Conway S, et al. Measuring and improving respiratory outcomes in cystic fibrosis lung disease: opportunities and challenges to therapy. Journal of cystic fibrosis : official journal of the European Cystic Fibrosis Society 2010;9(1):1-16. doi: 10.1016/j.jcf.2009.09.003

10. Grzincich G, Gagliardini R, Bossi A, et al. Evaluation of a home telemonitoring service for adult patients with cystic fibrosis: a pilot study. J Telemed Telecare 2010;16(7):359-62. doi:
10.1258/jtt.2010.091006 [published Online First: 2010/07/21]

11. Jarad NA, Sund ZM. Telemonitoring in chronic obstructive airway disease and adult patients with cystic fibrosis. J Telemed Telecare 2011;17(3):127-32. doi: 10.1258/jtt.2010.100309 [published Online First: 2010/12/18]

12. Lechtzin N, Mayer-Hamblett N, West NE, et al. Home Monitoring of Patients with Cystic Fibrosis to Identify and Treat Acute Pulmonary Exacerbations. eICE Study Results. American journal of respiratory and critical care medicine 2017;196(9):1144-51. doi: 10.1164/rccm.201610-2172OC

 Bella S, Murgia F, Tozzi AE, et al. Five years of Telemedicine in Cystic Fibrosis Disease. Clin Ter 2009;160(6):457-60. [published Online First: 2010/03/04]

14. Cox NS, Alison JA, Button BM, et al. Assessing exercise capacity using telehealth: a feasibility study in adults with cystic fibrosis. Respiratory care 2013;58(2):286-90. doi: 10.4187/respcare.01922

15. Wilkinson OM, Duncan-Skingle F, Pryor JA, et al. A feasibility study of home telemedicine for patients with cystic fibrosis awaiting transplantation. J Telemed Telecare 2008;14(4):182-5. doi: 10.1258/jtt.2008.070107 [published Online First: 2008/06/07]

16. Wood J, Mulrennan S, Hill K, et al. Telehealth clinics increase access to care for adults with cystic fibrosis living in rural and remote Western Australia. J Telemed Telecare 2016 doi: 10.1177/1357633X16660646

17. Cox NS, Alison JA, Rasekaba T, et al. Telehealth in cystic fibrosis: a systematic review. J
Telemed Telecare 2012;18(2):72-8. doi: 10.1258/jtt.2011.110705 [published Online First:
2011/12/27]

18. Ramsey BW, Davies J, McElvaney NG, et al. A CFTR potentiator in patients with cystic fibrosis and the G551D mutation. The New England journal of medicine 2011;365(18):1663-72. doi: 10.1056/NEJMoa1105185

19. Wainwright CE, Elborn JS, Ramsey BW. Lumacaftor-Ivacaftor in Patients with Cystic Fibrosis
Homozygous for Phe508del CFTR. The New England journal of medicine 2015;373(18):1783-4. doi:
10.1056/NEJMc1510466

20. Miller MR, Hankinson J, Brusasco V, et al. Standardisation of spirometry. The European respiratory journal : official journal of the European Society for Clinical Respiratory Physiology 2005;26(2):319-38. doi: 10.1183/09031936.05.00034805

21. Quittner AL, Buu A, Messer MA, et al. Development and validation of The Cystic Fibrosis Questionnaire in the United States: a health-related quality-of-life measure for cystic fibrosis. Chest 2005;128(4):2347-54. doi: 10.1378/chest.128.4.2347

22. Zigmond AS, Snaith RP. The hospital anxiety and depression scale. Acta psychiatrica Scandinavica 1983;67(6):361-70.

23. Quittner AL, Modi AC, Lemanek KL, et al. Evidence-based assessment of adherence to medical treatments in pediatric psychology. Journal of pediatric psychology 2008;33(9):916-36; discussion
37-8. doi: 10.1093/jpepsy/jsm064

24. Wang PS, Beck A, Berglund P, et al. Chronic medical conditions and work performance in the health and work performance questionnaire calibration surveys. Journal of occupational and environmental medicine / American College of Occupational and Environmental Medicine 2003;45(12):1303-11. doi: 10.1097/01.jom.0000100200.90573.df

25. Kessler RC, Barber C, Beck A, et al. The World Health Organization Health and Work Performance Questionnaire (HPQ). Journal of occupational and environmental medicine / American College of Occupational and Environmental Medicine 2003;45(2):156-74.

26. Brooke J. SUS: A "quick and dirty" usability scale. In: P. W. Jordan BT, B. A. Weerdmeester, &A. L. McClelland, ed. Usability Evaluation in Industry. London: Taylor and Francis. 1996:189-94.

27. Goss CH, Edwards TC, Ramsey BW, et al. Patient-reported respiratory symptoms in cystic fibrosis. Journal of cystic fibrosis : official journal of the European Cystic Fibrosis Society 2009;8(4):245-52. doi: 10.1016/j.jcf.2009.04.003

28. Wood J, Jenkins S, Putrino D, et al. High usability of a smartphone application for reporting symptoms in adults with cystic fibrosis. J Telemed Telecare 2017:1357633X17723366. doi: 10.1177/1357633X17723366

29. Schulz KF, Altman DG, Moher D, et al. CONSORT 2010 statement: updated guidelines for reporting parallel group randomised trials. BMJ 2010;340:c332. doi: 10.1136/bmj.c332

Smartphone application questions (participants answer yes or no).

In the past week, have you had:

Worsening sputum volume or colour?

New or increased blood in your sputum?

Increased cough?

New or increased chest pain?

New or increased wheeze?

New or increased chest tightness?

Increased shortness of breath or difficulty breathing?

Increased fatigue or lethargy?

Fever?

Loss of appetite or weight?

Sinus pain or tenderness?

In the past week do you feel that your health has worsened?

In the past week, have you felt:

Low in mood?

Worried?

Original article

A smartphone application for reporting symptoms in adults with cystic fibrosis improves the detection of exacerbations: results of a randomised controlled trial

Jamie Wood^{1,2,3}, Sue Jenkins^{1,2,3}, David Putrino^{4,5}, Siobhain Mulrennan^{3,6}, Sue Morey^{3,6}, Nola Cecins², Natasha Bear⁷, Kylie Hill^{1,3}

¹School of Physiotherapy and Exercise Science, Curtin University, Perth, Western Australia

²Physiotherapy Department, Sir Charles Gairdner Hospital, Perth, Western Australia

³Institute for Respiratory Health, Sir Charles Gairdner Hospital, Perth, Western Australia

⁴Abilities Research Center, Department of Rehabilitation and Human Performance, Icahn School of Medicine at Mount Sinai, New York, New York

⁵Department of Rehabilitation Medicine, Icahn School of Medicine at Mount Sinai, New York, New York

⁶Department of Respiratory Medicine, Sir Charles Gairdner Hospital, Perth, Western Australia

⁷Department of Clinical Research and Education, Child and Adolescent Health Services, Perth, Western Australia

Corresponding author:

Associate Professor Kylie Hill, School of Physiotherapy and Exercise Science, Faculty of Health Sciences, Curtin University, Building 408, Brand Drive, Bentley WA 6102.

Email: k.hill@curtin.edu.au

Keywords: Cystic fibrosis; Exacerbation; Smartphone; Telehealth; Telemedicine

Word count: 2992

Abstract

Background: Respiratory exacerbations impair lung function and health-related quality of life in people with CF, with delayed identification of exacerbations often resulting in worse outcomes. We developed a smartphone application (app) for adults with CF to report symptoms to the CF team, and investigated its impact on antibiotic use and other outcomes.

Methods: Participants were randomised to intervention (use of the app weekly or sooner if symptoms had worsened) or control (usual care). The app comprised questions relating to symptoms suggestive of an exacerbation. If worsening symptoms were reported, the participant was contacted by the nurse practitioner. The primary outcome measure was the number of courses and days of intravenous antibiotics.

Results: Sixty participants (31 male, aged [mean \pm SD] 31 \pm 9 years, FEV₁ 60 \pm 18% predicted) were recruited, with 29 (48%) allocated to the intervention group. Over the 12-month follow-up, there was no clear effect of the app on the number of courses of intravenous antibiotics (incidence rate ratio [IRR] 1; 95% confidence interval [CI] 0.6 to 1.7), however number of courses of oral antibiotics increased (IRR 1.5; 95% CI 1.0 to 2.2). The median [IQR] time to detection of exacerbation requiring oral or intravenous antibiotics was shorter in the intervention group compared with the control group (70 [123] vs.141 [140] days; p=0.02). No between-group differences were observed in other outcomes.

Conclusion: The use of an app reduced time to detect respiratory exacerbations that required antibiotics, however did not demonstrate a clear effect on the number of courses of intravenous antibiotics.

Introduction

In cystic fibrosis (CF), a respiratory exacerbation has been defined as having a change in four or more disease-specific criteria comprising cough, sputum, haemoptysis, dyspnoea, fatigue, sinus pain and discharge, fever, reduced forced expiratory volume in 1 second (FEV₁), anorexia or weight loss, and changes in physical examination of the chest or chest radiography (1). Exacerbations can have a severe and long-lasting impact on people with CF, by accelerating the rate of decline in lung function (2), increasing the time spent in hospital (3), reducing health-related quality of life (HRQoL) (4, 5) and physical activity (6), increasing feelings of depression (7), and ultimately increasing the risk of mortality (8).

The early detection of exacerbations is an integral component of CF care. Delayed initiation of treatment (i.e. antibiotics) has been associated with failure of lung function to recover to pre-exacerbation levels (9). Furthermore, up to 25% of people with CF who have an exacerbation fail to have their lung function recover following treatment (9). Therefore, strategies that are targeted at the early identification of exacerbations may be a vital step in optimising long-term outcomes in people with CF.

In people with CF who develop an exacerbation, delayed identification of exacerbations is often due to delayed presentation to a CF clinic. The reasons for this delay are multifactorial, and include commitments such as work, study and family, as well as the barriers to attending a CF clinic such as travel and financial burden, a lack of education and understanding around the need to report symptoms sooner, and feelings of anxiety and depression associated with attending a CF clinic (10). The use of telehealth is becoming an increasingly popular way of reducing the barriers associated with accessing CF care.

The impact of telehealth technology in CF has been evaluated in interventions aimed at identifying exacerbations (11-15), providing outpatient clinics (16), and providing exercise assessment and training (17, 18). Although adults with CF are likely to be comfortable with the use of mobile technologies, the effectiveness of telehealth at detecting exacerbations (11-15) has previously been compromised by suboptimal adherence (i.e. 10% to 67%). This is likely due to the high reporting burden experienced by participants, such as daily reporting of symptoms and/or regular spirometry. Minimising this burden in order to optimise adherence is an important consideration when designing telehealth technology.

We have previously developed a simple, intuitive, and novel smartphone application (app) (19) that allows the user to report changes in their symptoms directly to the CF team. The aim was to explore the effects of this app on overcoming the delay in presentation to a CF clinic by facilitating the earlier detection of exacerbations, which in turn may provide an opportunity to initiate treatment sooner. To optimise adherence, the app was designed to exclusively utilise smartphone technology that was mobile and already owned by the majority of young adults in society. We hypothesised that by improving symptom reporting and reducing the time to detect exacerbations, the risk of more severe exacerbations requiring treatment with intravenous (IV) antibiotics may be reduced.

Methods

This study was a prospective, single centre randomised controlled trial (RCT) which was undertaken according to a published protocol (19).

Participants

People with a diagnosis of CF were eligible if they were; (i) aged \geq 18 years, (ii) attended the adult CF centre at Sir Charles Gairdner Hospital in Perth, Western Australia, (iii) had at least one exacerbation requiring IV antibiotics in the preceding 12 months, (iv) exacerbation free for a minimum of 2 weeks prior to recruitment, (v) in possession of a smartphone, (vi) able to understand

written and spoken English. People who had previous lung transplantation or were currently listed for lung transplantation were excluded.

Participants were recruited during routine outpatient clinic visits during the period from January 2016 to February 2017. The participant information and consent form was provided by the Principal Investigator, and potential participants were contacted at least 48 hours later to discuss their willingness to participate. This study was approved by The Sir Charles Gairdner Group Human Research Ethics Committee for Western Australia Health (2015-030) and Curtin University Human Research Ethics Committee (HR212/2015). It was registered prospectively with the Australian New Zealand Clinical Trials Registry (ACTRN12615000599572).

Study design and intervention

Following informed consent, all participants completed measures prior to randomisation. Thereafter, participants were allocated to the intervention or control group using an independent computergenerated phone randomisation service (Figure 1). The randomisation sequence was stratified according to factors known to influence exacerbation rates, such as sex, location (metropolitan vs. rural/remote) and whether or not the participant had been prescribed either ivacaftor or lumacaftor/ivacaftor.

Participants in the intervention group downloaded the app on their smartphone. The app comprised 12 questions regarding changes in respiratory symptoms that were adapted from Fuchs exacerbation scale (1) and 2 questions regarding emotional well-being. Each question required a 'yes' or 'no' response. The app was previously tested in a usability study and demonstrated high usability amongst people with CF and near-perfect agreement between CF clinicians interpreting the data (19).
Participants were asked to answer the 14 questions via the app once a week or sooner if they felt their symptoms had worsened. If a participant failed to report their symptoms during a week, a text message was sent to remind them to complete the questions the following week. The symptom data was securely transmitted via password-protected email to the research team. If the participant reported that any symptoms had worsened, the CF nurse practitioner contacted them by phone to discuss their symptoms and whether they required a formal review in the CF clinic, or whether treatment (i.e. oral or inhaled antibiotics) was to be initiated immediately. Antibiotics were prescribed by, or in consultation with the CF physicians. Although clinical judgment rather than standardised criteria were used to define an exacerbation, the physicians prescribing the antibiotics were blinded to group allocation. Participants were asked to attend the CF clinic every 3 months if they had not presented sooner.

Participants in the control group continued with usual care, which involved attending the CF clinic appointments every 3 months. They were also able to contact the CF nurse practitioner via phone at any time to report a change in their clinical status.

Measurements

The primary outcome measure was the number of courses and days of IV antibiotics over the 12month intervention period. These data were extracted from the medical records for each participant on study completion. Antibiotic use was verified by a pharmacist who was blinded to group allocation. Secondary healthcare utilisation (HCU) outcomes extracted for each participant on study completion comprised the time (days) to detection of exacerbation requiring IV antibiotics, time to detection of exacerbation requiring oral or IV antibiotics, number of courses and days of oral and inhaled antibiotics, number of hospital admissions and days, and number of CF clinic visits. Adherence to the weekly use of the app and the symptom data were collated upon study completion. Other secondary outcomes, were collected prior to, and at 6 and 12 months following randomisation. These comprised spirometry (Medgraphics USB spirometer [MGC Diagnostics, Minnesota, USA] or Easyone spirometer [ndd Medical Technologies, Massachusets, USA]), nutritional status (body mass index [BMI]), HRQoL (Cystic Fibrosis Questionnaire – Revised [CFQ-R]), feelings of anxiety and depression (Hospital Anxiety and Depression Scale [HADS]), medication adherence (Treatment Adherence Questionnaire-CF [TAQ-CF]), and absenteeism and presenteeism (Health and Work Performance Questionnaire). System usability of the app was measured at 6 and 12 months following randomisation with the System Usability Scale (20), scored out of 100.

Statistical analyses

Statistical analyses were undertaken with Stata (StataCorp, Stata Statistical Software Release: V.14) in accordance with the intention-to-treat principle. Categorical data is presented as frequencies and proportions. Continuous data and count is presented as medians and ranges due to skewed distributions. General linear mixed models were used to examine continuous outcomes with suitable distributions. Health care utilisation data presented as count data with skewed distributions. Negative binomial regression was used to examine difference between groups for HCU, with this model chosen over Poisson as the data was over dispersed. Models were adjusted for adjusted for corresponding baseline variables, sex, FEV₁, and courses of IV antibiotics in the preceding 12 months. Kaplan-Meier and Cox regression were used to compare time to detection of exacerbation.

Sample size calculations

To detect a between-group difference ($\alpha = 0.05$, $1 - \beta = 0.8$) in the number of IV antibiotic courses of one per participant, assuming a standard deviation of 1.21 (based on IV antibiotic usage at the SCGH CF centre during 2013 and 2014), a sample size of 23 in each group was required. To account for any loss to follow up or drop out, we aimed to recruit an additional seven participants to each group for a total sample size of 60.

Results

Participant characteristics

Of the 180 patients attending the CF centre, 70 (39%) met the inclusion criteria and were invited to participate in the study. Sixty participants were enrolled in the study (Table 1), with 29 (48%) allocated to the intervention group. One (3%) participant in the control group did not attend the 6 or 12 month study visits, and one participant in the intervention group received lung transplantation prior to the 12 month assessment.

Number of courses and days of IV antibiotics

Following adjustment for the courses of IV antibiotics in the preceding 12 months, sex, and forced expiratory volume in 1 second, over the 12 month intervention period there was no important effect of the intervention on the number of courses and days of IV antibiotics per participant (Table 2).

Secondary HCU outcomes

Over the 12 month intervention period, there was no clear effect of the intervention on the time to detection of exacerbation requiring IV antibiotics, being a median (interquartile range) of 186 (298) vs. 273 (184) days in the control group (p = 0.20; Figure 2). The time to detection of exacerbation requiring oral or IV antibiotics was shorter in the intervention group compared to the control group with a median (interquartile range) of 70 (123) vs. 141 (140) days, respectively (p = 0.02; Figure 2). There was evidence to support an increase in the prescription of oral antibiotics in the intervention group, but not in the prescription of inhaled antibiotics (Table 2). There was no clear effect on the number of hospital admissions and admission days and number of CF clinic visits. Participants in the intervention group communicated via phone with the nurse practitioner median (IQR) 7 (8) times during the study, compared to 3 (4) times (p = 0.01) for participants in the control group.

Adherence to the weekly use of the app was a mean (range) of 77 (25 to 100)%, with 1,136 of a possible 1,483 reporting occasions completed. Subgroup analysis demonstrated an increase in oral antibiotic prescription, and reduction in IV antibiotic prescription, compared with the 12 months preceding the study in participants in the intervention group with adherence $\geq 80\%$ (n = 15) (Figure 3).

Of the completed reporting occasions, participants reported an increase in at least one symptom 399 (35%) times, (median [range] 10 [1 to 44]) per participant). Oral or IV antibiotics were subsequently prescribed on 113 (28%) of these occasions. The most commonly reported symptoms were increased cough, increased fatigue or lethargy, and worsening sputum volume or colour (Table 3).

Other secondary outcomes

The mean (SD) system usability score for the app at 6 and 12 months was 87 (13) and 89 (13), respectively. There was no clear evidence of effect on spirometry (Table S1), nutritional status (Table S1), HRQoL (Table S2), feelings of anxiety and depression (Table S3), medication adherence, or absenteeism and presenteeism.

Discussion

For adults with CF, we developed a novel app for reporting changes in symptoms suggestive of an exacerbation and investigated its effect on the number of exacerbations requiring IV antibiotics and other health outcomes. Adherence to the weekly use of the app was much greater than reported in previous telehealth studies in people with CF, and system usability was maintained above 85% at both 6 and 12 months. The 95% CI for between-group differences in the number of exacerbations requiring IV antibiotics suggest there was no important effect of the intervention. However, exacerbations requiring oral or IV antibiotics were detected earlier, and there was an increased prescription of oral antibiotics in the intervention group. There was no important effect on spirometry, nutritional status, HRQoL or feelings of anxiety and depression, medication adherence, or absenteeism and presenteeism.

The level of adherence (77%) to the use of our app was somewhat better than earlier work that has investigated home monitoring interventions, in which adherence has ranged from 10% to 67% (11-15). Any intervention that is inconvenient and time consuming will likely have poorer adherence, given it is in addition to the already large treatment burden that people with CF experience (21). Adherence in our study did vary though, with 14 (48%) participants using the app less than 80% of the required weeks. It is also difficult to determine whether adherence to the app reflects real world medication adherence, given the TAQ-CF likely contains self-report bias and complete data relating to pharmacy medication collection were not able to be obtained. Sub-group analyses suggest that those participants in the intervention group who were more adherent (\geq 80%) to the weekly use of the app experienced an increase in oral antibiotic use and a subsequent reduction in IV antibiotic use. Further investigation is needed to determine the characteristics of participants likely to have better adherence to apps and other monitoring interventions so that prospective studies can determine the effectiveness in this sub-group.

The high usability of our app is further supported by the low attrition from the intervention group (3%). This is considerably less than earlier studies of telehealth interventions in a CF population, in which attrition ranged from 24% to 63% (11-14). The level of adherence and low attrition in our study is likely attributed to the use of smartphones, a small and portable technology already incorporated into the lifestyle of the participants, and the once weekly reporting requirement which took only two minutes to complete. The usability scores at both 6 and 12 months reflected those reported in the previous usability study for this app (19). High usability was maintained through to 12 months, indicating that the app is a potential longer term solution within routine clinical care.

The earlier detection of exacerbations and increased provision of oral antibiotics in the intervention group did not translate into results compatible with an important reduction in the number of courses of IV antibiotics. The greater number of courses of IV antibiotics prescribed in the intervention group was consistent with the number of courses seen in the preceding 12 months, and was adjusted for in the analysis. This does raise the question as to whether the early provision of oral antibiotic treatment alone can translate into a reduction in the need for IV antibiotics, and highlights the limited options available for clinicians when treating exacerbations. Likewise, as noted for the difference in IV antibiotics, our data are compatible with no important differences in secondary outcomes such as spirometry, nutritional status, HRQoL, feelings of anxiety and depression, absenteeism and presenteeism and treatment adherence. Despite this finding, it is encouraging that the app provided a means of earlier detection of exacerbations, and if implemented over a longer surveillance period (e.g. 3 years), this may provide clearer answers regarding the potential longer term benefits such as slowing the rate of decline in lung function.".

We were surprised to see that when compared with the 12 months preceding enrolment, there was an important reduction in the number of courses of IV antibiotics in both groups during the 12 month intervention period. While this may simply reflect a spurious finding, it is likely that participating in a study of this kind led to increased contact with the CF centre and potentially offered greater opportunity for people to collect prescriptions from the hospital pharmacy. Participants in both groups attended a higher number of CF clinic visits than are currently recommended (i.e. 4 per year) (22); therefore involvement in the study seems to have improved access to care for all participants. The single centre design of this study is a limitation in the ability to generalise the results to other CF centres, and a larger multicenter study is now warranted.

The most common symptoms reported in our study were cough, sputum, fatigue, shortness of breath and chest tightness, mirroring those previously reported as being triggers for seeking treatment for an exacerbation by children and adults with CF (23). Most of the symptoms included in the app questions were adapted from Fuchs exacerbation scale (1), but of note the symptom 'chest tightness' was included as it has been previously reported by people with CF as being commonly associated with exacerbations (23). As there is no universal agreement on how to diagnose an exacerbation in this population, (24), we made the pragmatic decision to mirror clinical practice and relied on symptom reporting alone as the first line of detection.

We chose not to include monitoring with smartphone spirometry because of the increased burden on the participant, which is likely to have compromised adherence. Future studies planning to add additional technology to supplement data obtained from the app must carefully consider the benefit versus the risk to adherence. The use of other readily available smartphone technologies, such as measurements of daily steps may be a useful addition as physical activity may decline during exacerbations, and this data can be collected with minimal additional burden.

It is also possible that once weekly reporting of symptoms was insufficient and that increased frequency may be helpful. That is, with more frequent reporting of symptoms, exacerbations may be detected earlier than in the present study. Given the high level of adherence observed with weekly reporting, there is scope to increase this to two or three times per week; however the impact of increasing the frequency of reporting on adherence will need careful evaluation, especially given that the results of this study have not demonstrated that using an app is more effective than simply calling or emailing the CF team.

Conclusion

This is the first app used by adults with CF to report changes in their symptoms directly to the CF team. The high level of adherence is attributed to the low reporting burden and use of technology already owned by participants. Provision of the app facilitated the earlier detection of respiratory exacerbations and treatment in the form of oral antibiotics. While our data did not demonstrate an

important effect on the number of courses of IV antibiotics for the participants over a 12 month period, this study supports the continued development and investigation of similar apps and technologies used to identify exacerbations, with the aim of improving long term health outcomes.

Acknowledgements

We would like to acknowledge the participants in this study. This study received funding support from the Sir Charles Gairdner Osborne Park Healthcare Group Research Advisory Committee, and the SCGH Physiotherapy Department and Department of Respiratory Medicine. Antibiotic data was verified by Natalia Popowicz, Senior Pharmacist at Sir Charles Gairdner Hospital.

Authorship

JW, SJ, DP, SAM, SM, NC, SJ contributed to the conception and design of the study; JW, DP, SM contributed to the acquisition of data; JW, SJ, DP, SAM, SM, NC, NB, SJ contributed to the analysis and interpretation of data; JW, SJ, DP, SAM, SM, NC, NB, SJ drafted the article and gave final approval of the version to be submitted.

Declaration of conflicting interests

Mr. Wood reports grants from Sir Charles Gairdner Osborne Park Healthcare Group during the conduct of the study; A/Prof Jenkins reports grants from Sir Charles Gairdner Osborne Park Healthcare Group during the conduct of the study. A/Prof Putrino has nothing to disclose. Clin A/Prof Mulrennan reports grants from Sir Charles Gairdner Osborne Park Healthcare Group, during the conduct of the study; Principal Investigator in Vertex clinical trials, non-financial support from Novartis, outside the submitted work; non-financial support from Novartis, outside the submitted work; non-financial support from Novartis, outside the submitted to disclose. Mrs. Cecins has nothing to disclose. Mrs. Bear has nothing to disclose.



Figure 1. CONSORT flow diagram.

	Intervention	Control
Participants $n(\%)$	29 (48)	31 (52)
	27 (10)	51 (52)
Female, n (%)	17 (59)	12 (39)
Age, yr	31 (10)	31 (8)
FEV ₁ , % predicted	58 (18)	61 (17)
-		
FVC, % predicted	71 (16)	78 (14)
$BMI kg/m^2$	21 (5)	23 (3)
Divit, Kg/iii	21 (3)	23 (3)
CFQ-R respiratory domain	67 (17)	71 (16)
	<u></u>	1-
IV antibiotics courses in past 12 months, n	60	45
Living in a rural / remote region, n (%)	4 (14)	6 (19)
Pseudomonas aeruginosa, n (%)	29 (100)	30 (97)
CF-related diabetes, n (%)	5 (17)	13 (42)
Pancreatic insufficiency, n (%)	23 (79)	29 (93)
······································	9 (29)	0.(20)
ivacattor or lumacattor/ivacattor, n (%)	8 (28)	9 (29)

Table 1. Participant characteristics at baseline (n = 60)

Data are presented as mean (SD) unless otherwise stated. $FEV_1 =$ forced expiratory volume in 1 second, FVC = forced vital capacity, BMI = body mass index, CFQ-R = Cystic Fibrosis Questionnaire - Revised, IV = intravenous.

	Interve	ention $(n = 29)$	Control (n = 31)			
	Nº	per participant	Nº	per participant	IRR	95% CI
IV antibiotic courses	42	1 (0 to 4)	30	1 (0 to 6)	1	0.6 to 1.7
IV antibiotic days	609	14 (0 to 64)	468	14 (0 to 105)	1.1	0.5 to 2.6
Oral antibiotic courses	71	2 (0 to 8)	46	1 (0 to 5)	1.5	1.0 to 2.2
Inhaled antibiotic courses	24	0 (0 to 4)	18	0 (0 to 4)	1	0.7 to 1.4
Hospital admissions	27	1 (0 to 4)	21	0 (0 to 5)	1	0.5 to 2.0
Admission days	244	4 (0 to 44)	214	0 (0 to 57)	0.9	0.5 to 2.0
CF clinic visits	249	9 (1 to 21)	216	6 (1 to 22)	1.1	0.8 to 1.4

Table 2. Participant healthcare utilisation data

Data are presented as number (N°) and median (range) per participant. Analyses performed using Negative Binomial regression, adjusted for sex, forced expiratory volume in 1 second, and courses of IV antibiotics in the preceding 12 months, measured prior to randomisation.

IRR = incidence rate ratio. CI = confidence interval. IV = intravenous.



Figure 2. Intravenous and oral antibiotic use in the 12 months pre and during the study for a) participants in the intervention group who were $\ge 80\%$ (n = 15); and b) < 80% adherent (n=14) to the weekly use of the app.

Data are presented as the total number of courses. Analyses performed using negative binomial regression. IV = intravenous.

a) An increase in oral antibiotic prescription (p = 0.02) and reduction in IV antibiotic prescription (p = 0.03) was observed during the intervention;

b) No change in oral or IV antibiotic prescription during the intervention.

Table 3. Smartphone application questions and response data for participants in the intervention group(n = 29). Questions required a 'yes/no' response

In the past week, have you had:	Nº	per participant
Worsening sputum volume or change in colour?	144	3 (0 to 22)
New or increased blood in your sputum?	50	0 (0 to 10)
Increased cough?	184	5 (1 to 28)
New or increased chest pain?	53	1 (0 to 20)
New or increased wheeze?	101	2 (0 to 25)
New or increased chest tightness?	124	2 (0 to 20)
Increased shortness of breath or difficulty breathing?	124	3 (0 to 19)
Increased fatigue or lethargy?	171	4 (0 to 28)
A fever?	39	0 (0 to 18)
Loss of appetite or weight?	52	1 (0 to 7)
Sinus pain or tenderness?	75	1 (0 to 19)
In the past week do you feel that your health has worsened?	143	4 (0 to 22)
In the past week, have you felt:		
Low in mood?	78	1 (0 to 19)
Worried?	79	1 (0 to 24)

Data are presented as the total number (N°) of times each question had a 'yes' response, and the median (range) per participant.



Figure 3. Time to detection of exacerbation requiring a) intravenous (IV) antibiotics; and b) oral or IV antibiotics plotted against time (days).

Data are presented as the proportion (%) of participants. Analyses performed using Kaplan-Meier, Cox regression. IV = intravenous.

a) No difference between the intervention and control groups (median ([IQR] of 186 [298] vs. 273 [184] days, respectively), p = 0.20; b) Shorter time to detection of exacerbation requiring oral or IV antibiotics in the intervention group compared to the control group (median [IQR] of 70 [123] vs. 141 [140] days, respectively), p = 0.02.

References

1. Fuchs HJ, Borowitz DS, Christiansen DH, Morris EM, Nash ML, Ramsey BW, et al. Effect of aerosolized recombinant human DNase on exacerbations of respiratory symptoms and on pulmonary function in patients with cystic fibrosis. The Pulmozyme Study Group. The New England journal of medicine. 1994;331(10):637-42.

2. Sanders DB, Bittner RC, Rosenfeld M, Redding GJ, Goss CH. Pulmonary exacerbations are associated with subsequent FEV1 decline in both adults and children with cystic fibrosis. Pediatric pulmonology. 2011;46(4):393-400.

3. Ouyang L, Grosse SD, Amendah DD, Schechter MS. Healthcare expenditures for privately insured people with cystic fibrosis. Pediatric pulmonology. 2009;44(10):989-96.

4. Britto MT, Kotagal UR, Hornung RW, Atherton HD, Tsevat J, Wilmott RW. Impact of recent pulmonary exacerbations on quality of life in patients with cystic fibrosis. Chest. 2002;121(1):64-72.

5. Flume PA, Suthoff ED, Kosinski M, Marigowda G, Quittner AL. Measuring recovery in health-related quality of life during and after pulmonary exacerbations in patients with cystic fibrosis. J Cyst Fibros. 2018.

6. Ward N, White D, Rowe H, Stiller K, Sullivan T. Physical activity levels of patients with cystic fibrosis hospitalised with an acute respiratory exacerbation. Respiratory medicine. 2013;107(7):1014-20.

7. Riekert KA, Bartlett SJ, Boyle MP, Krishnan JA, Rand CS. The association between depression, lung function, and health-related quality of life among adults with cystic fibrosis. Chest. 2007;132(1):231-7.

de Boer K, Vandemheen KL, Tullis E, Doucette S, Fergusson D, Freitag A, et al.
 Exacerbation frequency and clinical outcomes in adult patients with cystic fibrosis. Thorax.
 2011;66(8):680-5.

236

9. Sanders DB, Bittner RC, Rosenfeld M, Hoffman LR, Redding GJ, Goss CH. Failure to recover to baseline pulmonary function after cystic fibrosis pulmonary exacerbation. American journal of respiratory and critical care medicine. 2010;182(5):627-32.

10. Zemanick ET, Harris JK, Conway S, Konstan MW, Marshall B, Quittner AL, et al. Measuring and improving respiratory outcomes in cystic fibrosis lung disease: opportunities and challenges to therapy. J Cyst Fibros. 2010;9(1):1-16.

11. Bella S, Murgia F, Tozzi AE, Cotognini C, Lucidi V. Five years of Telemedicine in Cystic Fibrosis Disease. Clin Ter. 2009;160(6):457-60.

12. Jarad NA, Sund ZM. Telemonitoring in chronic obstructive airway disease and adult patients with cystic fibrosis. J Telemed Telecare. 2011;17(3):127-32.

Murgia F, Bianciardi F, Solvoll T, Tagliente I, Bella F, Carestia A, et al. Telemedicine
 Home Program in Patients with Cystic Fibrosis: Results after 10 Years. Clin Ter.
 2015;166(6):e384-8.

14. Sarfaraz S, Sund Z, Jarad N. Real-time, once-daily monitoring of symptoms and FEV in cystic fibrosis patients--a feasibility study using a novel device. Clin Respir J. 2010;4(2):74-82.

 Lechtzin N, Mayer-Hamblett N, West NE, Allgood S, Wilhelm E, Khan U, et al. Home Monitoring of Patients with Cystic Fibrosis to Identify and Treat Acute Pulmonary Exacerbations. eICE Study Results. American journal of respiratory and critical care medicine. 2017;196(9):1144-51.

16. Wood J, Mulrennan S, Hill K, Cecins N, Morey S, Jenkins S. Telehealth clinics increase access to care for adults with cystic fibrosis living in rural and remote Western Australia. J Telemed Telecare. 2016.

17. Cox NS, Alison JA, Button BM, Wilson JW, Holland AE. Assessing exercise capacity using telehealth: a feasibility study in adults with cystic fibrosis. Respiratory care.

2013;58(2):286-90.

237

 Tomlinson OW, Shelley J, Trott J, Bowhay B, Chauhan R, Sheldon CD. The feasibility of online video calling to engage patients with cystic fibrosis in exercise training. J Telemed Telecare. 2019:1357633X19828630.

19. Wood J, Jenkins S, Putrino D, Mulrennan S, Morey S, Cecins N, et al. High usability of a smartphone application for reporting symptoms in adults with cystic fibrosis. J Telemed Telecare. 2018;24(8):547-52.

Brooke J. SUS: A "quick and dirty" usability scale. In: P. W. Jordan BT, B. A.
 Weerdmeester, & A. L. McClelland, editor. Usability Evaluation in Industry. London: Taylor and Francis.; 1996. p. 189-94.

21. Sawicki GS, Sellers DE, Robinson WM. High treatment burden in adults with cystic fibrosis: challenges to disease self-management. J Cyst Fibros. 2009;8(2):91-6.

22. Bell SC. Cystic Fibrosis Standards of Care, Australia 2008 [Available from: http://www.thoracic.org.au/imagesDB/wysiwyg/CF_standardsofcare_Australia_2008.pdf.

23. Goss CH, Edwards TC, Ramsey BW, Aitken ML, Patrick DL. Patient-reported respiratory symptoms in cystic fibrosis. J Cyst Fibros. 2009;8(4):245-52.

24. Bhatt JM. Treatment of pulmonary exacerbations in cystic fibrosis. European
respiratory review : an official journal of the European Respiratory Society. 2013;22(129):20516.

Appendix 10

MY CF TREATMENTS tr an Treatment Adherence

Most adults find it hard to do all of their treatments each day for good CF care. Please tell us which treatments you have done over the *last week*. You're not alone if you've been missing some medications and treatments. It is hard to fit it all in everyday. Please answer these questions honestly. Please check the box that is closest to your answer.

